Exploring Phenomena of Aggressive Intrusive Thoughts and Aggressive Scripts: The Influence of Associated Beliefs and Features

By Stephanie Jane Fernandez

BPsySc, Deakin University, 2017

BPsySc (Hons), Deakin University, 2018

A thesis submitted in partial fulfilment of the requirements for the degree of Doctor of Psychology (Clinical and Forensic Psychology)

Centre for Forensic Behavioural Science
School of Health Sciences
Swinburne University of Technology
Melbourne, Australia
2023

Supervisors

Associate Professor Maja Nedeljkovic
Professor Michael Daffern
Associate Professor Richard Moulding

© 2023 Stephanie Jane Fernandez

All rights Reserved

Under the Copyright Act 1968, this thesis must be used only under the normal conditions of scholarly fair dealing. In particular, no results or conclusions should be extracted from it, nor should it be copied or closely paraphrased in whole or in part without the written consent of the author. Proper written acknowledgement should be made for any assistance obtained from this thesis

ABSTRACT

Thoughts about harming another person are commonly experienced by the general population. Two constructs that pertain to aggressive thought experiences are aggressive intrusive thoughts (AITs) and aggressive scripts. AITs are a common feature of Obsessive Compulsive Disorder (OCD), where thoughts about harming another person are generally not thought to be associated with aggressive acts. Contrastingly, aggressive scripts are defined as thoughts or people experience about harming another person, and are commonly reported by people with a history of violent offending. Current understanding of aggressive scripts stems from academic literature within forensic domains, which hold a primary focus on how aggressive cognitive processes associate with levels of risk and aggressive behaviours. The convergence of research regarding AITs and aggressive scripts is limited despite the similarities in the level of measurement and definition of these constructs. It is important to note however that the consequences associated with these two types of thoughts are purportedly different, with one associated with aggressive acts (i.e., aggressive scripts) and the other with compulsive behaviours often conducted to prevent harmful events from occurring. It is therefore important to further understand the features of AITs in OCD, including their frequent, intrusive, ego-dystonic and distressing nature, and use these features to compare with how aggressive scripts are experienced by people in criminal justice settings who have a history of violence. Against this background, the aim of this thesis was to explore the phenomenology of AITs and aggressive scripts across non-clinical and forensic samples.

A critical review was conducted to explore the similarities and differences between AITs and aggressive scripts, using well established features of intrusive thoughts as a basis for comparison between the two phenomena. The critical review identified that both AITs and aggressive scripts may be experienced as frequent and recurrent, and that thought control strategies may be employed to manage the thought experience. Aggressive scripts were

suggested to differ from AITs with regards to being deliberately rehearsed, provoked by anger inducing situations, and influenced by a history of aggressive behaviour. The emotional precursors and consequences of aggressive script rehearsal were difficult to elucidate as research in this area was scarce, whereas, the emotional experience associated with AITs included feelings of distress and discomfort. Overall, the critical review identified that both AITs and aggressive scripts may be reported as frequent, recurrent, and associated with the use of thought control strategies. Differences between AITs and aggressive scripts were identified with regards to their emotional experience and the influence of past experiences of aggressive behaviour. It remains unclear whether aggressive scripts are experienced as unwanted and intrusive, and if they are associated with distress by those who reported them. Further, the critical review highlighted some challenges with regards to the conceptualisation and measurement of these phenomena.

Empirical study one examined the relationships between self-report measures of AITs, aggressive scripts and their relationship to the beliefs that individuals have about themselves, others, and aggression within a community sample. Specifically, the study examined similarities and difference between the two constructs in terms of specific correlates (i.e., ego-dystonicity, beliefs, aggression). Results demonstrated that anger rumination and violence supportive beliefs were associated with aggressive script rehearsal, and consistent with prior research, aggressive script rehearsal, anger rumination and violence supportive beliefs predicted a history of aggressive behaviour. Further, AITs were found to associate with obsessive beliefs, and only AITs were related with ego-dystonicity. Findings also demonstrated that both AITs and aggressive script rehearsal were associated with the use of thought control strategies. Overall, the findings from this empirical study highlight the importance of maladaptive beliefs in the context of AITs and aggressive script rehearsal, and

demonstrated that features pertaining to beliefs about violence, a life history of aggressive behaviour, and ego-dystonicity may differentiate aggressive scripts from AITs.

Empirical study two explored the experience of aggressive scripts and AITs in a sample of men recruited from a forensic mental health service. Utilising mixed methods, this study explored the characteristics and subjective experience associated with aggressive scripts and AITs. Thematic analysis of qualitative interviews indicated that the experience and impacts of aggressive thinking in individuals with a history of aggressive behaviour are complex, including both positive and negative experiences associated with the rehearsal of aggressive thoughts. The participant interviews highlighted the difficulty in attempting to differentiate between AITs and aggressive scripts using current measures of these constructs. Findings also indicated that aggressive thinking in individuals with a history of violence or anger problems may serve an emotional regulatory function. Overall, these results demonstrate the complexity of aggressive thinking and highlight the many features that may be associated with thoughts related to harming others.

The results from this thesis identify several features pertinent to AITs and aggressive scripts, which may be used as avenues for differentiation. These features include one's history of aggressive behaviour, the endorsement of violence supportive beliefs, and the interpretation of aggressive thoughts as either ego-dystonic or ego-syntonic. Findings also identified some issues with extant measurement instruments used to assess AITs and aggressive scripts, and recommendations for future research and suggestions for how to improve these instruments are discussed. This thesis has clinical implications for both AITs and aggressive scripts, where the identification of features that differentiate these phenomena may improve the early detection of these thoughts and assist with violence risk assessment and treatment for those who report them.

ACKNOWLEDGEMENTS

This thesis would not have been possible without the support from my academic team, friends, and family. I could not have undertaken this journey without you all, so thank you.

To my primary supervisor, Associate Professor Maja Nedeljkovic, I am so grateful and thankful for the endless guidance, reassurance, and support you have given me over this doctoral degree. Thank you for being such a great supervisor.

To my secondary supervisor, Professor Michael Daffern, thank you for your support, for always being available to answer my questions, and for encouraging me throughout this journey. Your supervision has been immensely helpful.

To my associate supervisor, Associate Professor Richard Moulding, thank you for your support, and for the guidance you gave me towards pursuing this doctoral degree. Thank you.

Thank you to the DPsych girls – Abi, Ella and Alissa. I am grateful for you support throughout this, and I am so glad to have shared this journey with you all. To Angelika, thank you for being such a great friend to me, and for the support you have given me throughout this long journey. I have been extremely lucky to have been surrounded by like-minded, intelligent, and kind friends and I will cherish the memories we have shared.

To my loved ones, thank you for the love, support, and encouragement you have given me the past four years. Thank you to my Mum who has supported me throughout all my years of studies, and has encouraged me to always follow my dreams. I am indebted to you. Thank you to my love J – your kindness, patience, and encouragement has been invaluable and I will be forever thankful for your support. Thank you to all who read the many psychology essays, listened to my thesis theories, and who motivated me to keep going.

You all have made completing this doctorate possible, so thank you!

DECLARATION

I, Stephanie Fernandez, declare that this thesis, titled, "Exploring Phenomena of Aggressive

Intrusive Thoughts and Aggressive Scripts: The Influence of Associated Beliefs and

Features":

1. Contains no material that has been accepted for the award to me of any other degree

or diploma, except where due reference is made about this in the text of the

examinable outcome

2. To the best of my knowledge, contains no material previously published or written by

another person except where due reference is made in the text of the examinable

outcome

3. Where the work is based on joint research or publications, the thesis discloses the

relative contributions of the respective workers or authors

I warrant that I have obtained, where necessary, permission for the copyright owners to use

any third party copyright material reproduced in the thesis (such as artwork, images,

unpublished documents), or to use any of my own published work (such as journal articles) in

which they copyright is held by another party (such as publisher, co-author). This thesis

includes three original papers published or under-review in peer-reviewed journals.

Signature of candidate: **fernandez**

Stephanie Jane Fernandez

Doctor of Psychology (Clinical and Forensic) Candidate

Centre for Forensic Behavioural Science

Faculty of Health, Arts and Design

Swinburne University of Technology

March 2023

V

TABLE OF CONTENTS

ABSTRACT	i
ACKNOWLEDGEMENTS	iv
DECLARATION	V
TABLE OF CONTENTS	vi
LIST OF TABLES	xiv
LIST OF ABBREVIATIONS	XV
LIST OF THESIS PAPERS	xvii
PART I – GENERAL BACKGROUND AND INTRODUCTION	1
CHAPTER 1 – OVERVIEW OF THESIS	2
1.1 Introduction	2
1.2 Thesis Overview	3
1.3 Thesis Research Aims	5
1.4 Research Impact and Contribution	6
CHAPTER 2 - PHENOMENOLOGY OF AGGRESSIVE INTRUSIVE THOUGHTS	8
2.1 Intrusive Thoughts	8
2.1.1 Epidemiology	10
2.1.2 Aggressive Intrusive Thoughts	12
2.1.3 Measuring Aggressive Intrusive Thoughts	17
2.2. A Cognitive Model of Intrusive Thoughts: Maladaptive Beliefs	20
2.2.1 Salkovskis' Cognitive Behavioural Model of OCD	20
2.2.2 Rachman's Cognitive Model of OCD	22
2.2.3 A Summary of Salkovskis and Rachman's Cognitive Models of OCD	
2.2.3 Maladaptive Beliefs in OCD	25

2.2.4 Self-related Beliefs and OCD	32
2.3 Treatment for Intrusive Thoughts and OCD	35
2.4 Summary	36
CHAPTER 3 - PHENOMENOLOGY OF AGGRESSIVE SCRIPTS	38
3.1 Aggressive Scripts	38
3.1.1 Epidemiology of Aggressive Scripts	41
3.1.2 Measuring Aggressive scripts	42
3.2 Cognitive Theories of Aggressive Scripts	43
3.2.1 Script Theory	44
3.2.2 The General Aggression Model	45
3.2.3 The Multiple Systems Model	46
3.2.4 Normative Beliefs and Attitudes Supportive of Aggression	48
3.2.5 Additional Cognitive Beliefs Implicated in Aggressive Script Rehearsal	50
3.3 Aggressive Scripts and other Conceptually Relevant Constructs	51
3.3.1 Fantasy	51
3.3.2 Anger Rumination	53
3.3.3 Aggressive Intrusive Thoughts versus Aggressive Scripts	54
3.4 Treatment for Aggressive Script Rehearsal	57
3.5 Summary	58
CHAPTER 4 - A CRITICAL COMPARISON OF AGGRESSIVE INTRUSIVE	
THOUGHTS IN OBSESSIVE COMPULSIVE DISORDER AND AGGRESSIVE	
SCRIPTS IN OFFENDER POPULATIONS	60
4.1 Preamble for Critical Review Paper	60
4.2 Author Indication Form for Empirical Research Paper One	61
4.3 Concentual Overview	67

4.3.1 Aggressive Intrusive Thoughts	67
4.3.2 Aggressive Scripts	69
4.4 Features of Intrusive Thoughts: Comparisons with Aggressive Scripts	71
4.4.1 Unwanted, Intrusive and Spontaneous	71
4.4.2 Frequent, recurrent and not easily dismissed	76
4.3.3 Distressing versus pleasant subjective experiences	78
4.4.4 Control and neutralising strategies	84
4.4.5 Self-Themes: ego-dystonicity & the feared self	90
4.5 Summary	95
PART II –EMPIRICAL RESEARCH: OVERVIEW, METHODOLOGY AND FA	INDINGS
	9 7
CHAPTER 5 – EMPIRICAL RESEARCH INCLUDED IN THIS THESIS	
CHAFTER 5 – EMFIRICAL RESEARCH INCLUDED IN THIS THESIS	70
5.1 Empirical Study One	98
5.2 Empirical Study Two	99
CHAPTER 6 – METHODOLOGY	100
6.1 Overview	100
6.2 Research Design	100
6.3 Participant Selection	101
6.4 Ethical Procedures	102
6.5 Procedure	103
6.5.1 Recruitment	103
6.5.2 Data Collection Process	106
6.6 Materials/Measures	107
6.6.1 Participant Demographics	107

6.6.2 Socially Desirable Responding	107
6.6.3 Aggressive Script Rehearsal	107
6.6.4 Aggressive Intrusive Thoughts	108
6.6.4 Anger Rumination.	110
6.6.5 History of Aggressive Behaviour	111
6.6.6 Violence Supportive Beliefs	111
6.6.7 Ego-dystonicity	112
6.6.8 Thought Control	113
6.6.9 Obsessive Beliefs	113
6.6.10 Self Ambivalence	114
6.6.11 Feared Self	114
6.6.12 Obsessive Compulsive Symptoms	114
6.6.13 Depression, Anxiety and Stress	115
6.6.14 Construct Phenomenology	115
6.7 Participant Sample	116
6.8 Data Preparation and Analysis Methods	117
6.8.1 Data Preparation and Analyses for Empirical Study One: Exploring Predictors of Aggressi	ive
Intrusive Thoughts and Aggressive Scripts: Similarities and Differences in Phenomenology	117
6.8.2 Analyses for Empirical Study Two: Exploring the Experiences of Aggressive Intrusive Th	oughts
and Aggressive Scripts in an OCD and Forensic Sample - A Case Study	118
CHAPTER 7 – EMPIRICAL RESEARCH PAPER ONE	122
7.1 Preamble to Empirical Research Paper One	122
7.2 Author Indication Form for Empirical Research Paper One	123
Abstract	125
The Appraisal Process of AITs and Aggressive Scripts	128
Aggressive Intrusive Thoughts	128
Neutralisation and Thought Control Strategies	130

Current Study	131
Method	131
Participants	131
Measures	132
Procedure	136
Results	136
Data Inspection	136
Correlational Analyses	137
Regression Analyses	140
Discussion	142
CHAPTER 8 – EMPIRICAL RESEARCH STUDY TWO	148
8.1 Preamble to Empirical Research Paper Two	148
8.2 Recruitment Context for Empirical Study Two	148
8.3 Authorship Indication Form for Empirical Research Paper Two	150
Abstract	152
Cognitive Explanations of AITs and Aggressive Scripts: Consideration of Self-themes	154
Differential Features of AITs versus aggressive scripts	157
Current Study	159
Method	159
Research design	159
Participants	160
Materials	161
Procedure	162
Qualitative and Quantitative Analysis	162
Results	163
Quantitative findings	163

Qualitative findings	165
Discussion	170
Experiences Associated with Aggressive Thinking	171
Negative Impacts	171
Thought Management	171
Precipitants of Thoughts	172
Negative and Positive Experiences	
Strengths, Limitations and Future Research Directions	176
Conclusion	176
PART III – GENERAL DISCUSSION AND CONCLUSION	178
CHAPTER 9 – INTEGRATED DISCUSSION	179
9.1 The Features of AITs and Aggressive Scripts	180
9.1.1 Similarities in Phenomenology of AITs and Aggressive Scripts	
9.1.2 Differences in Phenomenology of AITs and Aggressive Scripts	184
9.2 Limitations	189
9.2.1 Sample	189
9.2.2 Measurement	190
9.3 Implications of Research	191
9.3.1 Future Research Directions	191
9.3.2 Assessment Implications	192
9.3.1 Clinical and Treatment Implications	194
9.4 Conclusion and Final Comments	196
APPENDICES	222
Appendix A	222
Appendix B	224

Appendix C	226
Appendix D	227
Appendix E	228
Appendix F	229
Appendix G	234
Appendix H	236
Appendix I	237
Appendix J	238
Appendix K	241
Appendix L	242
Appendix M	243
Appendix N	247
Appendix O	250
Appendix P	251
Appendix Q	252
Appendix R	253
Appendix S	255
Appendix T	256
Appendix U	258
Appendix V	259
Appendix W	260
Appendix X	262

Appendix Y	264
Appendix Z	267

LIST OF TABLES

Table 1. Differentiating Characteristics of Aggressive Intrusive Thoughts and Aggressive
Scripts95
Table 2. Pearson's Correlations for all variables
Table 3. Beta coefficients (t statistics) for regression analyses predicting AITs, Aggressive
scripts (SIV), obsessive-compulsive symptoms (OCI), and life history of aggression (LHA).
Table 4. The six phases of thematic analysis used in the current study derived from Braun and
Clarke (2006)
Table 5. Demographics and Self-Report Measure Scores for Each Participant
Table 6. List of features and their presence in the experience of AITs or Aggressive Scripts
Table 7. Example questions for the assessment of differential features of AITs and aggressive
scripts

LIST OF ABBREVIATIONS

AITs Aggressive Intrusive Thoughts

ARS Anger Rumination Scale

AUD Australian Dollar

CBT Cognitive Behavioural Therapy

CFMHS Community Forensic Mental Health Service

CINAHL Cumulative Index to Nursing and Allied Health Literature

DASS Depression Anxiety Stress Scale

EDQ Ego-dystonicity Questionnaire

EM Expectation-Maximisation

ERP Exposure Response Therapy

FSQ Fear of Self Questionnaire

GAM General Aggression Model

III Interpretation of Intrusions Inventory

IITIS International Intrusive Thoughts Interview Schedule

LHA Life History of Aggression

MCAA Measure of Criminal Attitudes and Associates

MCAR Missing Completely at Random

MEDLINE Medical Literature Analysis and Retrieval System Online

OBQ Obsessive Beliefs Questionnaire

OC Obsessive-Compulsive

OCCWG Obsessive Compulsive Cognitions Working Group

OCD Obsessive Compulsive Disorder

OCI-R Obsessive Compulsive Inventory - Revised

PTSD Post-Traumatic Stress Disorder

ROII Revised Obsessional Intrusions Inventory

SAM Self Ambivalence Measure

SDS Social Desirability Scale

SIV Schedule of Imagined Violence

SPSS Statistical Package for Social Sciences

STOP Systematic Treatment of Obsessive Compulsive Phenomena

TCQ Thought Control Questionnaire

QUIT Questionnaire of Unpleasant Intrusive Thoughts

LIST OF THESIS PAPERS

- Fernandez, S. J., Daffern, M., Moulding, R. & Nedeljkovic, M. (2022). Exploring Predictors of Aggressive Intrusive Thoughts and Aggressive Scripts: Similarities and Differences in Phenomenology. *Aggressive Behavior*.1-3.

 http://doi.org/10.1002/ab.22061
- Fernandez, S. J., Daffern, M., Moulding, R. & Nedeljkovic, M. A Critical Comparison of Aggressive Intrusive Thoughts in Obsessive Compulsive Disorder and Aggressive Scripts in Offender Populations. *Manuscript submitted for publication (Aggressive and Violent Behavior)*. (Chapter Four)
- Fernandez, S. J., Daffern, M., Moulding, R. & Nedeljkovic, M. Exploring the Experiences of Aggressive Script Rehearsal in a Sample of Adult Males Recruited from a Forensic Mental Health Service. *Manuscript submitted for publication (Journal of Interpersonal Violence)*. (Chapter Eight)

PART I – GENERAL BACKGROUND AND INTRODUCTION

CHAPTER 1 – OVERVIEW OF THESIS

1.1 Introduction

Thoughts of harming another person are cognitions commonly reported by the general population as well as people with a history of violent offending (Daff et al., 2015; Rowa & Purdon, 2003). These aggressive thoughts are associated with different outcomes depending on the population these thoughts are investigated in and the measurement instruments used to assess them. Two types of constructs that pertain to aggressive thinking include aggressive intrusive thoughts (AITs) and aggressive scripts. AITs are a common feature of Obsessive Compulsive Disorder (OCD) where thoughts of inflicting harm or injury onto others are experienced as highly distressing and worrying (Rachman, 1997). It is the general consensus that AITs within people presenting with OCD are not associated with overt acts of violence or aggression, rather individuals may go to extraordinary lengths to prevent harm occurring to others after experiencing these thoughts (Veale et al., 2009). Aggressive scripts on the other hand are thoughts about harming another person and are commonly reported within populations drawn from criminal justice and forensic mental health service settings. There is extensive empirical research highlighting the association between aggressive script rehearsal and acts of violence (Grisso et al., 2000).

Current understanding of how AITs are experienced suggests that there is an interaction between different features of intrusive thoughts including level of intrusiveness, ego-dystonicity, and maladaptive beliefs about the self and others, which may influence how aggressive intrusions are appraised (Clark, 2005). Aggressive scripts on the other hand are said to be interpreted through the lens of different normative beliefs about aggression (e.g., Believing that someone who makes you angry deserves to be hit; Mills et al., 2002) and are related to one's past experience with aggressive or violent behaviours. These two types of aggressive thoughts have distinct behavioural outcomes, one with non-existent aggressive

behaviours (i.e., OCD) and the other with a direct relationship with violence (i.e., forensic population). What is not clear in the research is how and why these thoughts, which share content similarity, have diverse behavioural outcomes. There is limited research exploring whether differential features common to AITs (e.g., level of intrusiveness, ego-dystonicity, distress, beliefs) are applicable to the experience of aggressive scripts. Similarities and differences in the phenomenology of these thought experiences does not appear to have been given due consideration. Research exploring these two types of aggressive thoughts has been conducted independently, and within two separate fields of research.

The present thesis argues that AITs and aggressive scripts have some similarities but some associated features differ, which may explain the differences in behavioural outcomes. Exploring the comparable and differentiating features of AITs and aggressive scripts may inform the assessment of violence risk and intervention.

1.2 Thesis Overview

This thesis explores the phenomenology of AITs and aggressive scripts, and investigates the similarities and differences between these thought phenomena. Across seven chapters, the thesis presents some of the literature that informed the current research (chapters two and three), a critical review of the phenomenology of AITs and aggressive scripts (chapter four), the methods and rationale for two empirical studies (chapters five and six), two empirical studies (chapters seven and eight), and an integrative discussion (chapter nine).

Chapter one (current chapter) presents an outline of the thesis topic and the research aims of the thesis.

Chapter two presents a theoretical overview and definition of AITs including a definition of the construct and how it is understood from general intrusive thought conceptualisations. A summary of empirical research exploring cognitive behavioural

explanations of intrusive thoughts is provided. According to these explanations, AITs are associated with obsessive-compulsive symptoms due to the misappraisal of the thought phenomena. The chapter also highlights common measurements approaches to intrusive thoughts.

Chapter three presents a theoretical overview of aggressive scripts. Empirical research from the field of forensic psychology is explored and summarised, and the role that aggressive scripts have on aggressive behaviour is detailed. According to Script Theory (Huesmann, 1988), aggressive scripts are developed through the observation of, or engagement in, aggressive behaviour, and the more one engages with aggressive scripts in mind, the more likely they are to behave aggressively. Additionally, common measurement approaches to the study of aggressive scripts are described.

Chapter four presents a critical review that explores whether characteristics of AITs and aggressive scripts are comparable. Further, the critical review aims to address whether specific features common to intrusive thoughts (e.g., level of intrusiveness, distress, egodystonicity) are applicable to current conceptulisations of aggressive script experiences.

Chapters five and six present the methodological approaches for the empirical studies, including a background of the rationale and decision-making processes used for recruitment and data collection.

Chapter seven presents the exploration of AITs and aggressive script phenomena in a non-clinical sample – empirical study one. The purpose of this study was to explore the differential predictors of AITs and aggressive scripts using self-report measures. This chapter includes an overview of AITs and aggressive scripts, followed by the aims and methodology of the study. The results from the regression analyses and the psychometric properties of the scales used are presented and the findings of the study are discussed.

Chapter eight presents the investigation of aggressive thoughts in individuals with a history of aggressive behaviour or problems with aggression and anger – empirical study two. The purpose of this study was to explore the features of aggressive thoughts and examine the subjective experiences associated with aggressive thinking. This chapter includes both qualitative and quantitative results from the study, and presents the themes identified from qualitative analyses. The findings of the study are discussed and the implications of the results are explored.

Finally, chapter nine provides an integrated discussion of the results from the two empirical studies in the context of current literature on AITs and aggressive scripts. The research aims, limitations, and strengths of the research are discussed.

Please note: As the current thesis follows a thesis by publication format, there will be some unavoidable repetition across chapters, particularly in the introductory and methodology chapters and the discussion where results from manuscripts are considered.

1.3 Thesis Research Aims

This thesis aims to explore the phenomenology of AITs and aggressive scripts.

Specifically, this thesis aims to determine the similarities and differences in the phenomenology of AITs and aggressive scripts. This thesis will investigate the role that differential beliefs about the self, others, and aggression have on the experiences of AITs and aggressive scripts.

Three research questions were formulated:

- 1. What is the phenomenology of AITs and aggressive scripts?
- 2. What are the similarities and differences in the phenomenology of AITs and aggressive scripts?
- 3. What are the subjective experiences associated with aggressive thoughts?

1.4 Research Impact and Contribution

This project is unique in that it investigates phenomena that show similarities in content (i.e., thoughts about harming another person) but purported differences in behavioural outcomes. These phenomena have not been examined concurrently. The purpose of this project is to determine the differentiating features of AITs, as compared with aggressive scripts, which may inform assessment and clinical practice. Specifically, this study may inform violence risk assessment by determining which features of aggressive thought content are associated with acting aggressively.

To date, the experience of aggressive behaviours in groups of individuals with OCD is non-existent (Booth et al., 2014; Veale et al., 2009). By definition, intrusive thoughts are experienced as unpleasant, distressing, and represent an ego-dystonic aspect of one's self (i.e., not in line with one's sense of self; Moulding, Aardema, et al., 2014a; Purdon et al., 2007). The fears and worries associated with intrusive thoughts, which are commonly reported by individuals with OCD, influence repetitive behaviours in the hope to avert the perceived risks associated with the intrusive thought (Veale et al., 2009). It is therefore the general consensus amongst clinicians with expertise in OCD diagnosis and treatment that little to no risk is associated with intrusive thoughts of an aggressive nature (Aardema & O'Connor, 2007; Veale et al., 2009). However, there has not been prior empirical scrutiny of this assumption. Moreover, the current project seeks to identify the factors associated with the absence of aggressive behaviours in individuals who experience AITs. More specifically, the current project seeks to examine whether aggressive scripts are associated with intrusiveness, discomfort, and various emotional experiences that are commonly associated with AITs.

Limited research has explored the features (e.g., intrusiveness, distress, egodystonicity) of AITs that may be similarly experienced by people rehearsing aggressive scripts. This project will contribute to knowledge on how aggressive thoughts are experienced within a non-clinical and forensic population, and may also inform assessment and treatment avenues for OCD and forensic populations.

CHAPTER 2 - PHENOMENOLOGY OF AGGRESSIVE INTRUSIVE THOUGHTS

This chapter explores and synthesizes research pertaining to intrusive thoughts, specifically aggressive intrusive thoughts (AITs), in OCD. This chapter begins by exploring the prevalence of intrusive thoughts more generally, and describes the experience of AITs and their impacts on those who experience them. This chapter summarises nearly 40 years of intrusive thought theoretical explanations, including contributions from Salkovskis (1985) with the Cognitive Behavioural Model of OCD and Rachman (1997) with his development of the Cognitive Model of intrusive thoughts and obsessions. Measurement of intrusive thoughts is also explored with attention given to issues in differentiating between whether a thought is deemed intrusive or not. This chapter also explores the cognitive appraisals and beliefs that are known to contribute to obsessive-compulsive symptoms, and their relationship with different compulsive behaviours are explored. Attention is also given to a relatively new area of focus in intrusive thoughts, self-themes, and their contribution to obsessive-compulsive symptoms. This chapter ends with an examination of treatments for intrusive thoughts and OCD more generally.

2.1 Intrusive Thoughts

Most individuals have experienced an unpleasant or disturbing thought entering consciousness spontaneously, sometimes without provocation (Clark & Purdon, 1995) and empirical evidence confirms that approximately 94% of healthy individuals will report having experienced a recent intrusive thought (Radomsky et al., 2014). Intrusions have been empirically investigated for several decades, with pioneering research by Rachman and de Silva (1978) defining *intrusions* as unwanted thoughts, images or impulses that are similar in form and content to clinical obsessions. Unwanted thoughts are conceptualised as thoughts or doubts individuals may have about something, whereas images may be conceptualised as

photographs that may appear suddenly in one's mind. Further, impulses are considered the experience of a strong and urgent need to do or say something (Pascual-Vera et al., 2019; Rachman & de Silva, 1978).

An intrusive thought is defined as the experience of persistent and repetitive thoughts about something, that are interpreted as unacceptable and unwanted, and are associated with discomfort (Rachman, 1981). Intrusive thoughts can reflect both negative or positive content themes. For example, experiencing a thought about harming another person is typically regarded as a negative content theme by individuals with OCD, whereas positive intrusive thoughts can be those relating to inspiration and creativity that increase one's motivations or productivity (i.e., thinking about the future, or pleasant memories; Clark & Purdon, 1995; Salkovskis, 1989). Reynolds and Salkovskis (1992) explored the phenomenology of positive and negative intrusive thoughts and identified that negative intrusions are associated with disturbances in mood and distress when thought content is considered unacceptable, whereas positive intrusions are found to occur more frequently, however are not associated with mood deterioration (Reynolds & Salkovskis, 1992). Negative intrusive thoughts can comprise various themes, and the most commonly reported content themes include repeated doubts, sexual thoughts, aggression, and violence (Grisso et al., 2000; Moulding, Aardema, et al., 2014b; Parkinson & Rachman, 1981; Purdon & Clark, 1993). By definition, negative intrusive thoughts are those experienced as ego-dystonic, where the content and experience of the thought is inconsistent with ones' belief system and they are interpreted as alien to the self (Salkovskis, 1985).

It is postulated that intrusive thoughts can be placed on a continuum ranging from normal intrusive thoughts experienced by the general population, to clinical obsessions like those empirically investigated in OCD (OCD; Rachman, 1981; Rachman & de Silva, 1978). More recent theoretical research has challenged this notion of a continuum, questioning the

universality of intrusive thoughts in healthy individuals (Berry & Laskey, 2012; O'Connor, 2002). Berry and Laskey (2012) argue that a strict focus of viewing intrusive thoughts and clinical obsessions on a continuum fails to encapsulate broader factors that explain the differing experiences. In light of this, continuum theory, which purports that clinical obsessions are characterised as a stronger, more severe version of normal intrusive thoughts, has been replaced with a broadening focus on differential factors (i.e., distress, maladaptive appraisal processes, content themes) that may explain differences between what we understand as clinical obsessions versus intrusive thoughts (Berry & Laskey, 2012). Given this, the main defining features that segregate intrusive thoughts from clinical obsessions are not only the frequency with which they are reported (Berry & Laskey, 2012), but also the extent of distress associated, the appraisal process, and the content themes of the intrusion (Belloch et al., 2004; Clark et al., 2000). Given the commonality of intrusive thoughts it can be assumed that intrusive or unwanted thoughts experienced by the general population are able to be disregarded, with little attention or meaning given to their experience (Salkovskis, 1985). However, in situations where intrusive thoughts are appraised as a negative experienced and felt as ego-dystonic, the interaction between an individual's belief system and the appraisal given to the thought results in significant deteriorations in mood and behaviour (Rachman & de Silva, 1978; Salkovskis, 1985). It is this process of interconnections between maladaptive beliefs, appraisals, and behaviours that is central to cognitive behavioural models of intrusive thoughts, which explain the progression of normal intrusions to those commonly reported in cases of OCD (Salkovskis, 1985).

2.1.1 Epidemiology

Several non-clinical and clinical studies have investigated the prevalence and incidence of intrusive thoughts. Early research on intrusive thoughts has confirmed that unwanted intrusive thoughts, images and impulses are experienced commonly by most

participants assessed (Purdon & Clark, 1993; Rachman & de Silva, 1978). More recent empirical research by Radomsky et al. (2014) confirms the commonality of intrusive thoughts, where an international study of non-clinical students from 15 cities, across 12 countries, and over six continents (*N* = 777) found that approximately 94% of participants reported the experience of an unwanted intrusive thought in the previous three months. Studies comparing the incidence of intrusive thoughts between student samples and clinical participants with OCD have identified that both groups experience at least one type of intrusion, where those with OCD experience intrusive thoughts more frequently and with greater distress (Bouvard et al., 2017; Garcia-Soriano et al., 2011). These comparison studies have also identified that non-clinical and clinical samples may differ in the content themes of the intrusive thoughts, however the frequency and level of distress associated with the intrusions appear to be clearer indicators of differences between samples.

While intrusive thoughts can encompass varied content (i.e., doubt, aggression, and sexual themes), several studies have explored which themes are experienced most frequently by community samples (Parkinson & Rachman, 1981; Radomsky et al., 2014; Rowa & Purdon, 2003). A seminal non-clinical study conducted by Parkinson and Rachman (1981) found that 95% of participants (N = 60) reported the experience of an intrusive thought, with the most frequently reported content theme related to death, followed by significant harm or injury to family or friends. Within comparison studies of non-clinical and clinical samples of OCD participants, Bouvard et al. (2017) identified that the most prevalent intrusions for participants with OCD (n = 28) related to contamination and doubt, and the most common intrusion for non-clinical participants (n = 28) was doubt intrusions. Although both samples demonstrate similarity in thought content, Bouvard et al. (2017) found that participants with OCD experienced intrusive thoughts more frequently than the non-clinical participants. It is important to note however, that there is inconsistency across studies that report the

prevalence of specific content themes between clinical and non-clinical samples, where some studies have found that samples do no differ in thought content (Belloch et al., 2007).

Further research is required to investigate whether differential experiences in thought content occur between clinical and non-clinical samples of participants.

2.1.2 Aggressive Intrusive Thoughts

An under studied area of intrusive thoughts are intrusive thoughts with aggressive content themes. AITs are an OCD symptom dimension that comprise themes relating to inflicting physical harm or injury onto another person, which may occur in the form of thoughts, images or urges and have often been reported as symptoms in various mental disorders like depression, schizophrenia, post-partum OCD, sexual-OCD, and OCD more generally (Abramowitz et al., 2003; Grisso et al., 2000; Vella-Zarb, 2017). Thoughts relating to violence¹ and aggression² have been commonly researched in social and clinical psychology, with various studies reporting the commonality of such types of thoughts (Grisso et al., 2000; Moulding, Aardema, et al., 2014b). However, the exploration of aggressive thoughts that are experienced intrusively and within clinical psychology domains, like OCD samples (Moulding, Aardema, et al., 2014b), is limited.

The prevalence of AITs is difficult to ascertain from reviews of the literature as differences in the measurement of AITs and definitions across research domains influences the reliability of incidence rates. Rowa and Purdon (2003) found that in an undergraduate student sample (N = 64) the most reported content theme of intrusive thoughts related to harm, aggression, or sexual impulses towards others, with 75% of students confirming this

¹ Violence is defined as an extreme form of aggression with the intention to cause significant harm and cause deliberate serious physical injury (Anderson & Bushman, 2002; Berkowitz, 1993).

² Aggression is defined as any behaviour that is carried out with the intention to cause immediate harm toward another person, and that the target person is motivated to avoid this behaviour (Anderson & Bushman, 2002).

experience. However, it is important to note that relevant literature focusing on aggressive or violent intrusions are conflicted in reporting the commonality of such content themes. For example Radomsky et al. (2014) found that aggressive, sexual or "unacceptable" content intrusions were the least commonly reported types of thoughts, where the most frequent included thoughts about doubt (e.g., "did I lock the front door before I left the house?"). Differences in the prevalence of AITs reported as most upsetting between the Rowa and Purdon (2003) and Radomsky et al. (2014) studies may be due to differences in the way intrusive thoughts were measured. Radomsky et al. (2014) utilised highly trained researchers experienced in the administration of the International Intrusive Thoughts Interview Schedule, and interviews with participants were both quantitative and qualitative in nature. Rowa and Purdon (2003) used a self-report measure of intrusive thoughts (e.g., Interpretation of Intrusions Inventory [III]), which required participants to identify their own intrusive thought experiences without specific guidance or further questioning by researchers. It may be that the use of self-report measures of intrusive thoughts, such as those used by Rowa and Purdon (2003), made it easier for respondents to report their repugnant thoughts, as it may have avoided perceived shame and judgement in disclosing these thoughts within an interview context. This highlights the importance of being able to accurately identify these thoughts and differentiate them from other similar phenomena in order for the correct treatment to be offered to those who report these thoughts. Further, recent studies of AITs and intrusive thoughts more generally have identified the risk to which these thoughts pose on suicidality (Brakoulias et al., 2017), where it was found that violent obsessions were significantly associated with suicidality, beyond the association of depressive symptoms (Ching et al., 2017). Considering these implications of AITs, and the mixed findings surrounding the prevalence of intrusive thoughts with aggressive content themes, further investigations are

warranted to elucidate this phenomenon and the prevalence to which such thoughts are experienced.

In order to characterise and differentiate intrusive thought experiences based on their content and form, Lee and Kwon (2003) identified two subtypes of obsessions (autogenous and reactive obsessions). Autogenous obsessions were described as comprising thoughts of a sexual, aggressive, and immoral nature, or impulses and urges that are perceived as inconsistent with one's morals or intentions (i.e., ego-dystonic) – they are also known as unacceptable and repugnant thoughts. These autogenous obsessions are considered to be internally generated, spontaneously appear in one's mind without the presence of an identifiable trigger and are perceived as threatening to the individual. Reactive obsession comprised other OCD symptom dimensions including thoughts relating to contamination, symmetry, mistakes and accidents, and are said to be externally triggered. Lee and Kwon (2003) further distinguished differences between autogenous and reactive thoughts and it is emphasised that autogenous thoughts (like AITs) appear to 'intrude' into one's mind, and are perceived as irrational and ego-dystonic, and are not always related to logical overt control or neutralisation strategies. Reactive obsessions were described as being perceived as more realistic, and influence logical thought control strategies that may reduce the perceived negative consequences associated (Lee & Kwon, 2003). Lee and Kwon's (2003) separation of obsessions into two categories provides a basis for classifying and understanding the different heterogenous obsessions in OCD.

Further, unacceptable/taboo thoughts originally were defined as "pure obsessions" that included obsessive thinking with content themes of an aggressive, sexual and/or religious nature. "Pure obsessions" were distinct from typical representations of intrusive thoughts in OCD, where the absence of compulsions were a defining feature emphasised by Baer (1994). However more recent studies have confirmed that unacceptable/taboo thoughts are in fact

separate to "pure obsessions", as the presence of such intrusions have been reported alongside specific compulsive behaviours (Leonard & Riemann, 2012; Williams et al., 2011)

2.1.2.1 Compulsive Behaviours and AITs

Although compulsive behaviours and neutralisation strategies may be unique to an individual, their use often has a common purpose – to reduce the distress or perceived negative consequences associated with the intrusive thought. In *autogenous* obsessions, like AITs, people usually fear the consequences of having such thoughts intrude their mind and may resort to overt and covert strategies to neutralise or control the intrusions, and to alleviate the negative emotions associated with the intrusion. Research examining the compulsive and neutralising behaviours used to manage AITs is limited. Research in the area of compulsive and neutralising behaviours in OCD has generally focused on the symptom dimension of repugnant obsessions which contains thoughts related to sexual and blashphemous themes and not solely AITs (Lee & Kwon, 2003; Purdon & Clark, 1994a).

Within the context of AITs, individuals experiencing a persistent thought or image of stabbing their friend or loved one, may in this instance, engage in avoidance behaviours, such as limiting their proximity to sharp objects like knives. Additionally, there are instances where covert compulsions are not obvious to individuals who experience AITs, and such experiences have been classed as "pure O" intrusions/obsession (Rachman, 1971). Covert compulsions may include engaging in mental rituals, or neutralising strategies (counting in mid, distracting self) in the hope to reduce the prevalence or intensity of the AIT experience (Belloch et al., 2004). "Pure O" obsessions have been recognised as a symptom dimension of OCD which encompass aggressive, sexual, and religious intrusions but no compulsive behaviours (Baer, 1994). Historically, mental compulsions and reassurance seeking were overlooked as compulsive symptoms in OCD, however more recent advances in OCD

research have recognised that responses to "pure O" obsession are more covert rather than absent (Williams et al., 2011).

Imperative to understanding compulsive behaviours and treatment of OCD is the long-term efficacy of using control or neutralisation strategies for the persistent experience of AITs. As highlighted by Ahern et al. (2015), the use of neutralisation strategies to control unwanted intrusive thoughts more generally significantly reduce the experience of distress. However, Ahern et al. (2015) noted that the return of the unwanted intrusive thought at a later stage increased distress and the need to neutralise the thought. This addresses issues with the longevity of neutralisation strategies over time and raises questions around the efficacy of neutralisation strategies as a way of managing intrusive thoughts in the long term. This has also been addressed in other studies of thought control, where the use of certain strategies has been found to be maladaptive not only for the persistence of intrusive thoughts and dysfunctional beliefs, but also detrimental to general health (i.e., impact of thought control on insomnia and general functioning; Stokes et al., 2022). Moulding, Aardema, et al. (2014a) highlight that cognitive treatments for repugnant obsessions may include cognitiverestructuring which aims to normalise the experience of intrusive thoughts, as well as assisting clients to refrain from using compulsive behaviours. Further, Moulding, Aardema, et al. (2014a) address the importance of aligning treatment goals with helping the individual to learn that their intrusive thought experiences are not meaningful and are not associated with negative consequences. Psychoeducation around the use of neutralisation and compulsive behaviour to manage intrusive thoughts is imperative for understanding OCD phenomena (Moulding, Aardema, et al., 2014a). The research exploring control strategies highlighted that the use of covert or overt methods are motivated towards eliminating the thought or distress, and preventing the associated consequences (Moulding, Coles, et al., 2014). While thought control strategies may provide relief from the negative emotional

experience associated with the intrusive thought, the utility of using such strategies in the long-term remains questionable (Ahern et al., 2015).

In summary, AITs are considered a common symptom of OCD, often classified under the umbrella of repugnant (Moulding, Aardema, et al., 2014a) or autogenous obsessions (Lee & Kwon, 2003). AITs are interpreted as highly immoral and ego-dystonic to the individual and are not associated with aggression or violence (Veale et al., 2009). The experience of AITs is highly distressing to the individual, and it may often motivate the use of overt or covert compulsive strategies to manage the discomfort and perceived consequences associated with the thoughts. The experience of AITs in OCD are not considered a risk for violence as they are reported as highlighy distressing, inconsistent with one's sense of self and conduct, and interpreted as immoral. These features of AITs may prove important in differentiating similar aggressive thought phenomena, and may help to understand what features may act as contributors to violence risk.

2.1.3 Measuring Aggressive Intrusive Thoughts

Intrusive thoughts with aggressive content themes can be measured in a variety of ways, where several questionnaires of intrusive thoughts include sections dedicated to identifying aggressive or violent content (Radomsky et al., 2014). Historically, intrusive thoughts have been measured using self-report instruments like the Revised Obsessional Intrusions Inventory (ROII) developed by Purdon and Clark (1993), which has been implemented in non-clinical communities. The first part of the ROII involves asking individuals to rate their level of endorsement, on a 7-point Likert scale (0 = I have never had this thought"; 6 = I have this thought frequently during the day), towards 52 statements representing intrusive thought contents: aggression, sex, dirt, and contamination. The second part of the ROII involves the participant selecting which of their intrusive thoughts, from part one of the questionnaire, was most upsetting to them, which is evaluated on a 5-point Likert

scale (1 = not at all; 5 = extremely). The ROII also considers the individuals' response to the intrusive thought, by exploring cognitive and thought control processes using 10 appraisal dimensions and 10 thought control strategies (Purdon & Clark, 1994a). The use of self-report measures in the assessment of intrusive thoughts can be effective in measuring these phenomena in non-clinical samples, however, without clear defintions of what constitutes an intrusive thought and how it differs from other similar constructs, these instruments may be limited. Therefore the use of semi-structured interviews alongside self-report measures may provide a safeguard for these conceptual issues. Using semi-structured interview methods, a more recent measure of intrusive thoughts was developed and the International Intrusive Thoughts Interview Schedule (IITIS; Clark et al., 2014; Moulding, Aardema, et al., 2014b; Radomsky et al., 2014) has been empirically tested. The IITIS measures intrusive thoughts similarly to the ROII, by asking participants to rate their endorsement, frequency, distress, and control strategies associated with an intrusive thoughts experience. The use of measures within a semi-structured interview context allows the researcher to clarify conceptual issues respondents may have in identifying what an intrusive thought consists of. This ensures similar constructs such as worrysome thoughts or thoughts that are not intrusive are not explored by mistake (Radomsky et al., 2014).

More recently, a newly developed measure of intrusive thoughts, the Questionnaire of Unpleasant Intrusive Thoughts (QUIT), has been used with non-clinical individuals. The QUIT was derived on the basis of the aforementioned intrusive thought questionnaires (Purdon & Clark, 1993; Radomsky et al., 2014). The QUIT is split into 5 different sections, each delineated to a specific intrusive theme or concern: unpleasant content, physical appearance, diet and physical exercise, health-related, and relationship-related. The unpleasant content section includes items that assess aggressive intrusions (i.e., "harming a person close to me"), and individuals rate the frequency and discomfort associated (Pascual-

Vera et al., 2019). The QUIT has been validated in a non-clinical sample (Pascual-Vera et al., 2019), and findings from this sample have further confirmed the universality of intrusive thoughts with more than half of participants (N = 1,473) reporting having experienced an intrusive thought. It is important to note however, Pascual-Vera et al. (2019) did not investigate the extent to which AITs were experienced, rather intrusive thoughts were collapsed across different disorders (i.e., OCD-related intrusions, Body Dysmorphic Disorder-related, Illness Anxiety/Hypochondriasis, and Eating Disorders).

2.1.3.1 Limitations of AIT Measures and Assessment Considerations

Although validated measures of intrusive thoughts exist which can identify AITs, there appears to be some issues around the measurement of intrusive thoughts with regards to how respondents identify them. Both the ROII and QUIT provide participants with descriptions of what constitutes intrusive thoughts, as this process has been identified as an important element in ensuring that intrusive thoughts are specifically measured and not construed as other similar phenomena (e.g., ruminative thoughts; Clark & Purdon, 1995). However, although these measures stipulate that they measure intrusive thoughts, it has become clear that no single question can be used to identify whether a thought is intrusive or not. Whether a thought is considered intrusive is dependent on a number of factors commonly experienced with intrusive thought phenomena, such as the frequency of the thought, the level of associated distress, and whether the thought is experienced as ego-dystonic (Clark & Purdon, 1995). Consideration of these issues with measurement of intrusive thoughts is covered in chapter four, which provides a critical review of the literature on the measurement and phenomenological background of AITs and aggressive scripts.

In addition to the measurement issues related to the assessment of AITs, there are issues surrounding clinicians' understanding and experience exploring harm related intrusions with clients. There is limited research exploring clinicians' knowledge and ability

to recognise and differentiate AITs from other thought phenomena (e.g., thoughts of harm in psychosis, or aggressive thoughts in individuals with a history of violence). This may be due to the difficulty in one measurement tool containing all features required to differentiate AITs from other similar thought phenomena, as well as the limited research that exists in differentiating AITs from other similar thought constructs, such as aggressive scripts in forensic psychology fields. This highlights a need for further education and consideration of the need to provide clinicians with a better understanding of the prevalence and outcomes for those who report such thoughts (Booth et al., 2014; Friedman et al., 2008). By increasing clinicians' understandings of the current models of OCD and the factors associated with the development and response to AITs, it may provide crucial grounds for clarifying and assessing intrusive thought phenomena against other similar constructs.

2.2. A Cognitive Model of Intrusive Thoughts: Maladaptive Beliefs

2.2.1 Salkovskis' Cognitive Behavioural Model of OCD

In a seminal paper by Salkovskis (1985), the conceptual elements of the cognitive behavioural model of intrusive thoughts are explored, linking various relevant cognitive processes and behaviour. According to this model, intrusive thoughts may be triggered by a wide array of external stimuli, and the difference between healthy individuals who experience intrusive thoughts and those people with a diagnosis of OCD lies within the differential avoidance strategies used to prevent further triggers or inducements for the intrusion (Salkovskis, 1985). This model postulates that obsessional thoughts occur when the content of the thought is experienced as ego-dystonic, reflecting inconsistencies with one's belief system (Salkovskis, 1985). The reaction evoked in an individual will depend on how salient the content of the intrusion is to them, and how the intrusion is filtered through beliefs that are often maladaptive or dysfunctional (Salkovskis, 1985). It is these maladaptive beliefs, that

when activated, influence the individual to appraise their intrusive thoughts in ways that consider the degree of responsibility they should have of the safety of themselves or others, as well as considering all the potential consequences associated with the intrusive thought content.

The appraisal process of intrusive thoughts has been extensively investigated both in non-clinical and OCD studies (Freeston et al., 1991). Emphasised in the literature is the importance of interactions between an individual's belief system and appraisal process, as it leads one to develop negative evaluations about the meaning of their intrusion, and maladaptive inferences about their own self-concept (e.g., "this is a bad thing to be thinking about – therefore I am a bad person"; Salkovskis, 1985). Cognitive behavioural models of intrusive thoughts, such as Salkovskis' model, propose that the transition from normal intrusions to clinical obsessions derives from cognitions individuals engage with to evaluate and assign meaning to the contents of the intrusions (Salkovskis, 1989). Individuals may then engage in different behaviours performed to neutralise or reduce the levels of discomfort associated with the intrusive thought (Purdon & Clark, 1993). This process of obsessions and compulsions are key characteristics of OCD, where the experience of frequent and recurrent obsessions, that are experienced intrusively and associated with distress, are followed with compulsions that are resorted to in the hope to minimise the perceived consequences and distress associated (Purdon & Clark, 1993). The differences between healthy individuals and those with OCD lie in the way the intrusive thoughts are processed, where several studies have confirmed that differential beliefs are key to understanding how maladaptive appraisals may contribute to the development of OCD (Moulding, Aardema, et al., 2014b; Radomsky et al., 2014; Rowa & Purdon, 2003). Various studies have confirmed the relationships that exist between the faulty appraisal process of intrusive thoughts and their frequency (Brakoulias et al., 2014; Izadi et al., 2012; Rowa & Purdon, 2003). This appraisal process is implicated by

specific beliefs pertaining to the *uncontrollability of the thought*, the belief that *the thought* may come true (Belloch et al., 2004; Purdon & Clark, 1994a), responsibility one assigns to themselves for having the thought (Salkovskis, 1985, 1989), and the belief that the thought infers something about one's personality (Nikodijevic et al., 2015). These faulty appraisals are a prominent feature in OCD where these different beliefs are said to predispose individuals at risk of OCD by contributing towards maladaptive interpretations and compulsive behaviours that function to minimise the discomfort associated (Obsessive Compulsive Cognitions Working Group [OCCWG], 2003).

2.2.2 Rachman's Cognitive Model of OCD

Rachman (1971) proposed that the general population experience intrusive thoughts not unlike those reported by clinical samples. In a comparison study of OCD and non-clinical samples, Rachman and de Silva (1978) found that intrusive thought content was similar between groups, and that differences were revealed in the OCD sample where intrusive thoughts were reported with greater frequency, discomfort, and difficulty in the ability to control the thoughts. In order to understand the mechanisms involved in the process of 'normal' intrusive thoughts developing into clinical intrusions, Rachman (1997) developed a cognitive model for understanding intrusive thoughts, emphasising the role faulty appraisals, that are directed towards the self, have on the perpetuation of intrusive thoughts and compulsive behaviours over time. In support of this, Rachman (1997) observed that the main content of intrusive thoughts relate to abhorrent themes connected to one's moral systems, including aggression, sexual deviance, and blasphemy, which foster a sense of personal significance, impacting the appraisal of the thought. Further, these types of thoughts are often experienced as "sinful, disgusting, alarming or threatening", and may often lead one to interpret these thoughts as revealing hidden components of one's personality, desires, or intentions (Rachman, 1997; p. 795). It is through this interpretation process that individuals

experience distress and fear connected to the perceived consequences of the thoughts (e.g., "I could harm someone"). Once the intrusive thought has been interpreted as containing personal significance, resistance to the thought, in the form of suppression, avoidance, or compulsive behaviours may be used to ameliorate the perceived risks and distress associated (Rachman, 1997).

Rachman (1998) also addressed the different factors believed to influence individuals in misinterpreting their intrusive thoughts. Internal and external provocations have been highlighted to render one vulnerable to experiencing intrusive thoughts, including exposure to stress, bodily sensations that trigger intrusive thoughts, and increases in distress following periods of compulsive actions or unsuccessful thought control strategies (Rachman, 1998). It is through the interpretation of these internal and external cues that intrusive thoughts become salient to the individual and personally significant. Additionally, it is proposed that individuals who hold pre-existing beliefs that certain thoughts are dangerous or important are more likely to become impacted by intrusive thoughts that contain aggressive, sexual or blasphemous contents (Rachman, 1998). Further, Rachman (1998) argued that individuals who possess a 'tender conscience³' are vulnerable to obsessional experiences as they are more likely to interpret intrusive thoughts as unacceptable or inappropriate by attaching strong moral values or religious standards. The concept of Thought Action Fusion has also been addressed by Shafran et al. (1999) in influencing misinterpretations of the significance of intrusive thoughts. Specifically, thought action fusion beliefs include believing that the mere presence of an intrusive thought increases the likelihood of a negative event becoming true, as well as believing that having a morally repugnant or blasphemous thought is

³ A tender conscience has not been explicitly operationalised by Rachman (1998), however Rachman and Hodgson (1980) posit that individuals who hold elevated moral standards and are preoccupied by their personal conduct are likely to possess a tender conscience.

equivalent to committing the morally repugnant behaviour (Shafran et al., 1999). Rachman (1997) also proposed that depression and anxiety proneness can be considered as vulnerability factors to intrusive thoughts by influencing pessimistic outlooks or hypervigilance to external stressors and triggers, respectively.

Rachman's (1997) cognitive model of intrusive thoughts acknowledges the factors contributing to intrusive thought experiences, particularly those concerning unacceptable content themes like aggressive thoughts. Rachman's theory emphasises how intrusive thoughts that are misinterpreted as personally significant become problematic, highlighting the role that maladaptive self-referential beliefs may play in the perpetuation of intrusive thoughts in OCD.

2.2.3 A Summary of Salkovskis and Rachman's Cognitive Models of OCD

The models propsed by Salkovskis and Rachman both highlight the importance of faulty beliefs and their problematic influence over intrusive thought experiences. Salkovskis' theory emphasises the process by which individuals interpret an intrusive thought event, including how one may attach a significant amount of responsibility to themselves for having an intrusive thought experience deemed ego-dystonic. Salkovskis argues that it is through these beliefs, such as percieved responsibility, that a normal thought phenomena may develop into obsessions. Similarly, Rachman's cognitive model highlights the role maladaptive beliefs have on intrusive thoughts, but rather focuses on the moral systems he believes are common to all individuals. These include moral beliefs centred around sex, aggression, violence, and religion, whereby Rachman propses that these moral systems reflect common themes in intrusive thought experiences. While Rachman does address the influence certain beliefs have on the appraisal of intrusive thought experiences, the focus of these beliefs are centred on the personal significance (e.g., I may be a dangerous person) individuals place on themselves after an intrusive thought experience.. Both Salkovskis and Rachman emphasise

how these faulty appraisals may lead individuals to use compulsive, neutralising, or avoidance behaviours to manage their thoughts. Although these models provide a basis for understanding the maintenance of intrusive thought phenomena, these models provide little enquiry as to the origins of these beliefs. Further, the ego-dystonic nature of the intrusive thoughts is considered a factor of importance in these models, however, there is limited conceptualisation of the role ego-dystonicity may play in the trajectory of these thoughts overtime. Neverthless, Salkovskis and Rachman's cognitive models advance understandings of the type of faulty appraisals that influence intrusive thought experiences, highlighting the importance of cognitive mechanisms in intrusive thought development and trajectory.

2.2.3 Maladaptive Beliefs in OCD

The influence of maladaptive beliefs on intrusive thought experiences has been emphasised in cognitive models of OCD developed by Rachman and Salkovskis. In order to understand the specific type of beliefs associated with intrusive thoughts in OCD, research has been dedicated to exploring these belief systems. The Obsessive Compulsive Cognitions Working Group (OCCWG, 1997), a collaborative team of leading researchers in the field of OCD, originally identified six maladaptive cognitions thought to influence the transition from normal intrusive thoughts to clinical obsessions. These included: a) *overestimation of threat*, an exaggerated belief regarding the severity of harm/danger; b) *inflated responsibility*, the belief that one is responsible for the negative outcomes and harm towards oneself and others (e.g., illness, accidents); c) *over-importance of thoughts*, the belief that the mere presence of a thought deems it significant; d) *the importance of controlling thoughts*, the belief that one is responsible for controlling thoughts entering into mind in order to avoid the associated consequences; e) *intolerance of uncertainty*, beliefs about needing to be certain, and that one should be doubtful in the face of uncertainty/ambiguous situations; and f) *perfectionism*, the belief that one should not make mistakes, and that solutions to problems should be perfect.

These different beliefs may be measured using the Obsessional Beliefs Questionnaire (OBQ), and several factor analytic studies have been conducted to determine the most relevant factor solution that encompass the core beliefs in the development and maintenance of obsessive compulsive symptoms (Myers et al., 2008; OCCWG, 2005). To date, the most salient cognitions measured by a short form version of the OBQ (i.e., OBQ-20) confirm four maladaptive beliefs relating to: a) overestimation of threat; b) inflated responsibility; c) the over-importance/control of thoughts, and d) perfectionism/uncertainty (Moulding et al., 2011). These four belief domains are proposed to influence the misinterpretation of an intrusive thought, and in the context of clinical samples of OCD, these beliefs are said to play a contributing role in the transition of normal intrusive thoughts to clinical obsessions (Parkinson & Rachman, 1981; Salkovskis, 1989).

These belief domains have also been explored in non-clinical samples. An international study of intrusive thoughts by Radomsky et al. (2014) found that several of these beliefs, as measured by the OBQ-20, were relevant to the experience of intrusive thoughts. It was found that participants derived meaning of their intrusive thoughts by endorsing specific beliefs such as overestimating the perceived threat and importance of the thought, being intolerable to the anxiety and uncertainty related to the thought, and believing that one was required to control the thought as they deemed themselves responsible for having the thought in the first place (Radomsky et al., 2014). It was also found that perfectionistic beliefs, thought action fusion beliefs, and feeling that the thought was egodystonic (i.e., inconsistent with one's self-concept or intentions) were endorsed to a lesser frequency (Radomsky et al., 2014). Given that Radomsky et al. (2014) study included participants from several international sites, the results from this study indicate that the experience of maladaptive beliefs can occur cross-culturally, where the differences between samples of OCD and non-clinical individuals will lie in the frequency and level of rigidity to

which these beliefs are endorsed (Izadi et al., 2012; Radomsky et al., 2014). For example, Izadi et al. (2012) investigated the difference between appraisals of intrusive thoughts experienced by a sample patients with OCD (n = 59) and a healthy control sample (n = 54) in Iran. Using an earlier version of the OBQ (OCCWG, 2005), it was found that both groups utilised the following beliefs in the interpretations of their intrusive thoughts: threat, responsibility, over-importance of thought/control of thought, and perfectionism/uncertainty and that differences between groups lied in the frequency to which the beliefs were endorsed. The OCD group were found to report higher scores on all belief domains (i.e., OBQ total score), and were more likely to reflect perfectionism/uncertainty beliefs than the healthy control group (Izadi et al., 2012). This is consistent with previous findings which confirm that clinical presentations of obsessive beliefs are more pronounced than that seen in healthy group comparisons (Abramowitz et al., 2009). Findings from both Radomsky et al. (2014) and Izadi et al. (2012) are consistent with previous research in intrusive thoughts (Morillo et al., 2007; Purdon & Clark, 1994b) which postulate that not only are intrusive thoughts experienced by non-clinical populations, but the beliefs endorsed by non-clinical samples reflect similarities to those seen in clinical samples.

2.2.3.1 Maladaptive Beliefs and Specific Content Themes

Investigating the type of maladaptive beliefs that are implicated by specific content themes of intrusive thoughts have been explored recently (Brakoulias et al., 2013; Morillo et al., 2007). Through the investigation of appraisal processes that are involved with specific content themes, like AITs, it has been identified that individuals commonly report such thoughts as "most upsetting" relative to other content themes such as contamination fears (Rowa & Purdon, 2003). With a sample of undergraduate students from various disciplines (N = 64), Rowa and Purdon (2003) found that the most frequent thought reported were those relating to harm or sexual themes. It was found that students who experienced their intrusive

thought as most distressing and upsetting endorsed beliefs of *responsibility* and the need to *control thoughts*. Similar findings to Rowa and Purdon (2003) were identified in a study conducted by Belloch et al. (2004) where it was found that students endorsed appraisals of *importance of controlling thoughts*, as well as beliefs of *unpleasantness* and *unacceptability* when they were prompted to consider the meaning of their most upsetting intrusion.

However, Belloch et al. (2004) found that the least frequent intrusive thought were those relating to harm or aggressive behaviours, which is inconsistent with Rowa and Purdon (2003). Thus, these studies provide evidence for the endorsement of OC related beliefs (e.g., importance of controlling thoughts) among non-clinical populations when thinking about upsetting intrusive thoughts. Consistent with early investigations into the appraisal process of intrusive thoughts (Rachman, 1997; Salkovskis, 1989), the significance applied and meaning derived from the content of an intrusion appears to influence how the thought is experienced by non-clinical samples.

Both the frequency and meaning derived to the intrusive thought influences the subjective experience associated with the intrusion. Individuals commonly experiencing intrusive thoughts report high levels of discomfort or distress (Salkovskis, 1989). Within OCD research, AITs are recognised under the symptom dimension of unacceptable/taboo thoughts (Brakoulias et al., 2013). Unacceptable/taboo thoughts which encompass a combination of intrusive content themes (e.g., aggressive, sexual and religious) have been researched within samples of OCD patients, and results from these studies confirm the presence of specific maladaptive beliefs that are implicated by such types of intrusive thought (Brakoulias et al., 2013; Moulding, Aardema, et al., 2014b; Rachman, 1997). Unacceptable/taboo thoughts were investigated by Brakoulias et al. (2013) with a group of participants with OCD (N = 154). The study aimed to investigate the characteristics of unacceptable/taboo thoughts, using a clinician administered semi-structured interview and the

Yale-Brown Obsessive-Compulsive Scale to measure obsessive-compulsive symptoms. Participants were also asked to report their belief styles using the OBQ, to confirm the specific endorsement of beliefs pertaining to the unacceptable/taboo symptom domain. Results from the study found that intrusive thoughts relating to sexual, aggressive and religious themes were implicated by specific beliefs relating to the importance/control of thoughts (i.e., the belief that it is possible and one should exert control of one's thoughts; Brakoulias et al., 2013). The study also confirmed that the increased experience of sexual, aggressive and religious intrusive thoughts was associated with higher levels of distress, increased time spent with the thought in mind, greater levels of hostility, and being male. These findings not only confirm that unacceptable/taboo thoughts involve themes such as physically hurting/injuring someone or profound blasphemy, but that individuals experiencing them are likely to believe that the mere presence of the thought signifies importance, and that the thought should be controlled. Brakoulias et al. (2013) findings, despite being limited to samples of individuals with OCD, are consistent with previous research of similar nature where the experience of intrusive thoughts deemed as most upsetting are associated with the belief of importance of thought and the control of thoughts (Belloch et al., 2004)

Why some thoughts are more salient to individuals than other thoughts has become a focus of research within the OCD domain, with an emphasis placed on the contribution that maladaptive beliefs have on the interpretation and experience of different thought content (Brakoulias et al., 2014). More recent studies have identified how self-related beliefs, which involve misinterpretations of intrusive thoughts by believing they represent hidden or unwanted aspects of one's persona, may be applicable to the experience of intrusive thoughts with abhorrent or unacceptable content themes (Doron et al., 2008; Nikodijevic et al., 2015). These types of self-related beliefs warrant exploration to address the potential contribution

they have to intrusive thought interpretations and furthermore their influence on obsessive compulsive symptoms.

2.2.3.2 Ego-Dystonicity and Aggressive Intrusive Thoughts

It is widely accepted that intrusive thoughts are differentiated from normal cognitions through one major characteristic: ego-dystonicity. Ego-dystonicity is defined as experiencing a thought that has "little or no context within one's sense of self or personality" (Purdon et al., 2007, p. 94). Purdon et al. (2007) emphasises that for a thought to be considered ego-dystonic, the thought is perceived as contradicting one's morals, attitudes, beliefs, intentions, and past-behaviours, and it is associated with significant distress and trepidation.

With regards to AITs, empirical findings have identified how the experience of thoughts about harming another person, particularly loved ones or friends, are often rated as being inconsistent with one's morals and personality, and this has been identified for both individuals diagnosed with OCD and non-clinical samples (Belloch et al., 2012; Purdon et al., 2007). According to the cognitive behavioural model of OCD, ego-dystonic thoughts become problematic to an individual when they assign meaning to their thoughts or questions their moral culpability (Rachman, 1997). It is suggested that because ego-dystonic thoughts by definition are those that are experienced as contrary to one's self-view, the thought occurrence itself is more likely to be assigned a greater significance and interpreted as a threat (Clark, 2004; Garcia-Soriano et al., 2011; Purdon et al., 2007). As explored earlier, dysfunctional beliefs systems are implicated in this process of recognising and appraising intrusive thoughts, especially those that are deemed ego-dystonic. Through this overinterpretation as a result of these dysfunctional beliefs, individuals are likely to resort to obsessive compulsive behaviours to ameliorate the distress and discomfort associated with the thought (Rachman, 1997). However, as noted by Purdon et al. (2007), whether a thought is considered ego-dystonic is dependent on the context in which the thought occurs. For

example, Purdon et al. (2007) found that ego-dystonic thoughts were not always associated with obsessive-compulsive symptoms, and that this relationship did not follow a linear trend. It was postulated that individuals may experience thoughts in a transient or contextual fashion, or ego-dystonic thoughts may be accommodated for over time, perhaps becoming ego-syntonic, and thus alleviating the urge to engage in obsessive compulsive (OC) behaviours. It is also highlighted by Purdon et al. (2007) that intrusive thoughts can initially be appraised as ego-dystonic, and after this thought is experienced frequently and appraised over time, this same thought may be perceived as ego-syntonic and thus accommodated into one's self-view.

Audet et al. (2016) explored the relationship between ego-dystonicity and unwanted intrusive thoughts, specifically examining whether a lack of evidence for the reality of the intrusion influences the degree of ego-dystonicity experienced. Using a student sample (N =248) participants were asked to identify their most unpleasant intrusive thought, and the background surrounding the experience of the intrusive thought was examined. Following this, three clinicians rated participants responses as to whether they were considered egodystonic or ego-syntonic. Results demonstrated that ego-dystonic thoughts were almost always classified as intrusive thoughts that occurred without direct evidence – that is there was no evidence that would suggest the intrusive thought was based on reality (Audet et al., 2016). Findings also revealed that ego-dystonic thoughts were associated with obsessional beliefs concerning threat, responsibility, uncertainty, and importance and control of thoughts; which is consistent with extensive prior research (Rowa & Purdon, 2003). These findings highlight the notion that intrusive thoughts, especially those classified as ego-dystonic, may often occur without evidence of reality. This further addresses the implications of appraisal processes involved in intrusive thought deduction and points to several treatment targets which are explored in later chapters of this thesis. Further empirical exploration of how egodystonic thoughts, particularly those focused on aggressive content, influence the experience and appraisal of intrusive thoughts, and what role ego-dystonicity has overtime is warranted. Exploring the process through which ego-dystonic thoughts become ego-syntonic may prove beneficial to aggressive thoughts more generally.

2.2.4 Self-related Beliefs and OCD

Increased attention has been dedicated to investigating the impact that self-related beliefs have on the appraisal process of intrusive thoughts (Aardema et al., 2013; Nikodijevic et al., 2015). It has been postulated that individuals are more likely to infer contradictions in their own persona or believe that they are immoral when they experience intrusions that are significantly upsetting or contain unacceptable themes (Doron et al., 2008; Rowa et al., 2005). The misinterpretation of intrusive thoughts is suggested to influence the development and maintenance of obsessions, thus warranting investigation into their effects (Doron et al., 2008). Self-beliefs such as the fear of self (Aardema et al., 2013) or self-ambivalence (Bhar & Kyrios, 2007) have been considered important factors in understanding intrusive thought development and maintenance.

2.2.4.1 The Fear of Self

The feared self has been found to relate strongly with OCD presentations, where beliefs concerning "what the self might be or might become" has been found to relate with the content and appraisal process of intrusive thoughts (Aardema et al., 2013; Aardema & O'Connor, 2007; Ferrier & Brewin, 2005). This concept has been researched alongside specific content themes of intrusive thoughts (i.e., aggressive, blasphemous, sexual), where Aardema and O'Connor (2007) argue that these content themes are strongly aligned with perceptions of the "self-as-could-be" rather than the "self-as-is". The relevance that has been highlighted between unwanted/repugnant intrusions and the feared self has been further

confirmed with samples of individuals with OCD (Aardema et al., 2017). Using the Fear of Self Questionnaire (Aardema et al., 2013), Aardema et al. (2017) found that after controlling for obsessional beliefs, negative mood and inferential confusion, beliefs pertaining to the feared self were the only unique predictor of obsessions in a group with OCD. Further, Llorens-Aguilar et al. (2021) explored the experience of actual and feared self perceptions and their association with intrusive thoughts and obsessions in a sample of patients with OCD (N = 58). Participants were asked to identify their most upsetting obsession and intrusion experienced in the past three months. They were then asked to describe their actual self and their feared self, and whether they thought these descriptions were related to their obsession/intrusive thought experience. Results suggested that participants used descriptors such as insecure/volatile, doubtful, anxious, fearful and obsessive, empathetic, good, honest, and joyful to describe their actual self; and descriptors such as materialistic, selfish, cruel, aggressive, bad, dishonest/liar, fearful or cowardly to describe their feared self (Llorens-Aguilar et al., 2021). Findings also suggested that participants believed that their obsessions revealed something about their actual self, and that the experience of their obsession brought them closer to their feared self (e.g., violent, aggressive, manipulative, out of control). These findings support previous characterisations of the feared self (Aardema & O'Connor, 2007; Melli et al., 2016) where the meaning derived to intrusions reflect beliefs pertaining to states of immorality, dangerousness, and insanity towards ones' self-concept (Ferrier & Brewin, 2005). How the feared self relates to intrusions with aggressive themes specifically is not yet clearly understood but given that previous research on unacceptable thought domains has identified significant relationships with the feared self (Aardema et al., 2017) further research may produce similar findings.

2.2.4.2 Self-Ambivalence

Early investigations by Guidano and Liotti (1983) proposed that intrusions become frequent and distressing when they intersect with one's beliefs about their own morality and social approval. Specifically, Guidano and Liotti (1983) described the influence that self-ambivalence has on obsessional thoughts and behaviours, where one's ambivalence towards what their self consists of or overall sense of worth and morality, is said to influence interpretations of intrusions. Self-ambivalence is said to encompass dichotomous type thinking, which entails the self being viewed as either "good" or "bad", with no middle ground (Bhar & Kyrios, 2007). The concept of self-ambivalence shares similar appraisal processing to the feared-self construct, where individuals with high self-ambivalence are more likely to endorse an overall sense of mistrust towards their self-concept (Bhar & Kyrios, 2007; Guidano & Liotti, 1983).

More recently, Bhar and Kyrios (2007) investigated the relationship between self-ambivalence with OCD-specific beliefs and obsessive-compulsive behaviours. In a mixed sample of individuals with OCD, anxiety disorders, and a non-clinical community group, findings from this study indicated that self-ambivalence significantly related to the obsessional beliefs of importance of thoughts, responsibility, and perfectionism; and predicted obsessive-compulsive symptoms, after controlling for self-esteem, depression and anxiety (Bhar & Kyrios, 2007). It was also found that individuals with OCD were more self-ambivalent than the non-clinical community group, however no differences were found between individuals with OCD and anxiety disorders. These findings suggest that individuals who are unsure of their own self-worth and morality, and hold rigid beliefs about what their self-concept consists of (i.e., "good" or "bad") are more likely to experience obsessional beliefs and behaviours. It was postulated that an individual with high self-ambivalence will

experience obsessional beliefs and resort to obsessive-compulsive behaviours in order to maintain control over their thoughts and self-validation (Bhar & Kyrios, 2007).

It is of interest to the current thesis to investigate the relationship between selfambivalence and AITs, given that previous explorations of self-ambivalence have not limited the type of intrusions experienced by respondents.

2.3 Treatment for Intrusive Thoughts and OCD

While the scope of the thesis focuses on investigating the distinct phenomena between AITs and aggressive scripts, a treatment section has been included to provide context for the clinical assessment and implications of AITs. While not directly related to the aims of the thesis, this treatment section will provide a broader context for the implications of AITs explored in the integrated discussion (chapter 9) of the thesis.

Guidelines for the treatment of intrusive thoughts specifically, are scarce given that these thoughts commonly present with other symptoms of OCD, and thus are treated under the umbrella of OCD treatment guidelines. The gold standard treatment approach for OCD and clinically significant intrusive thoughts and obsessions involves the application of cognitive and behavioural intervention strategies (Clark, 2004). The use of cognitive-behavioural therapy (CBT) is founded on the premise that obsessions, intrusive thoughts, and other obsessive-compulsive symptoms are maintained by the presence of maladaptive cognitive appraisals. It is therefore applicable in the experience of unwanted AITs that modifying cognitive misappraisals and maladaptive beliefs become treatment targets for the management of these symptoms. As elucidated earlier in this chapter, the experience of violent and AITs increased suicide risk, over and above that of depressive symptoms (Ching et al., 2017), highlighting the importance of understanding not only the phenomena of AITs but also their assessment and treatment methods.

As outlined by Clark (2004) CBT approaches in OCD aim to reduce symptom distress through attempts to modify maladaptive appraisals and beliefs including excessive doubt and pre-occupation, as well as considering the use of automatic and enduring neutralisation strategies that maintain intrusive thoughts over time. Early behavioural therapy perspectives considered Exposure and Response Prevention techniques to treat OCD, which include exposing individuals to a feared situation (e.g., holding a sharp object like a knife; in the case of AITs) and preventing them from carrying out their compulsive or neutralisation strategy (Clark, 2004). This continues to form a major component of current CBT techniques for OCD, where the effectiveness of exposure and response prevention treatment approaches have been empirically investigated and support for their use is established (Whittal et al., 2005).

Explored by Clark (2004) are the therapeutic components of CBT for OCD which focuses on processes and applications for modifying faulty appraisals and beliefs related to intrusions. Clark (2004) provides descriptions of processes that can be completed with OCD clients including but not limited to psychoeducation around the cognitive appraisal model and its relationship with OC symptoms, identifying and differentiating faulty appraisals from intrusive thoughts, learning adaptive ways of appraising intrusive thoughts, and behavioural experiments focusing on restructuring maladaptive beliefs attached to intrusive thoughts. Clark (2004) highlights the importance of educating clients on the processes involved in OCD phenomena, especially the perpetuation of OC symptoms through some compulsive and neutralisation strategies.

2.4 Summary

This chapter summarises the key phenomenology and epidemiology of intrusive thoughts, including those specific to AITs. AITs are characterised by frequent, intrusive, and ego-dystonic thoughts that centre on harm or injury occurring to others or loved ones. AITs

are a common feature of OCD, but have also been reported in the general population and amongst people with other mental disorders (Grisso et al., 2000; Moulding, Aardema, et al., 2014a; Rowa & Purdon, 2003). This chapter also explored the cognitive models of OCD which stipulate that AITs become problematic when they are misinterpreted through dysfunctional beliefs (Rachman, 1997). These dysfunctional beliefs influence one to become concerned with the content of the intrusive thought, but also the meaning of the thought and its personal significance (Aardema et al., 2013; Nikodijevic et al., 2015).

Given that AITs within OCD are experienced as ego-dystonic, it is suggested that these thoughts contradict one's self-view and intentions, which in turn increases the likelihood that they interpreted as threatening and important (Clark, 2004; Purdon et al., 2007). It is acknowledged that the experience of AITs in OCD is not associated with aggressive behaviour (Veale et al., 2009), and whether features such as ego-dystonicity and dysfunctional beliefs preclude this behaviour warrants investigation. This chapter also addressed treatment considerations for intrusive thoughts and OCD more generally, highlighting the importance of CBT and ERP as components of treatment modalities.

CHAPTER 3 - PHENOMENOLOGY OF AGGRESSIVE SCRIPTS

Chapter three defines and explores the phenomenon of aggressive scripts and synthesizes research related to aggressive script rehearsal. The frequency of occurrence of aggressive scripts in various populations is highlighted, as well as issues related to measurement of aggressive scripts given its overlap with other related constructs such as aggressive fantasies and aggressive rumination. This chapter summaries cognitive theories of aggression including the pioneering research from Huesmann and Eron (1984) exploring factors related to aggressive behaviour, and more recent developments by Denson (2013) which considers the impacts of the related and opotentially overlapping phenomena of angry rumination and its relationships with anger and aggression. This chapter considers aggressive scripts in the context of other relevant constructs including aggressive fantasies and anger rumination, and aims to educate the differences between these constructs based on current empirical findings.

3.1 Aggressive Scripts

Aggressive scripts are defined as thoughts or daydreams about physically harming another person (Grisso et al., 2000). Research around aggressive scripts is most commonly conducted within forensic fields where the interaction between aggressive thoughts and aggressive behaviour has been researched, particularly from a social psychological perspective (Grisso et al., 2000; Huesmann, 1988; Moeller et al., 2017). To date, the intersection of AITs and aggressive scripts has not been clearly investigated and the subjective experience of either phenomena has not been directly compared.

Prior research has identified various cognitive processes relevant to the experience of aggressive scripts (Huesmann, 1988; Huesmann & Eron, 1984; Huesmann & Guerra, 1997). Huesmann (1988) defines scripts as cognitions that are stored in one's memory and are activated by external environmental cues and which guide behaviour. Conceptually, aggressive scripts are knowledge structures⁴ that play a role in the information-processing system and guidance of aggressive thinking and behaviour (Gilbert et al., 2013; Huesmann & Eron, 1984). Following a similar process to the acquisition of knowledge, aggressive scripts are maintained in an individual's cognitive repertoire by encoding, rehearsal, and retrieval processes (Huesmann & Eron, 1984). In the encoding process, a representation of an external stimuli (e.g., how aggressive behaviour is performed) is formed into one's memory, with differential cues within the environment attached to this representation (Huesmann & Eron, 1984). Once this representation is formed, which is now phenomenologically considered an aggressive script, the rehearsal of this script will determine its retention in memory. This rehearsal process can take many forms inclusive of simple recall, fantasising, or play acting (Huesmann & Eron, 1984). Retrieval of an aggressive script involves one's ability to access the script in memory, and it is during this process that the differential cues which were attached to the script during the encoding phase, play a useful role in facilitating access to the script (Gilbert et al., 2017; Huesmann & Eron, 1984). It is postulated that the more an individual behaves aggressively, or observes aggressive acts, the more exposure they have to the encoding process of an aggressive script (Huesmann & Eron, 1984).

-

⁴ Knowledge structures are related to schemas, which are encoded in memory and contain knowledge on a concept, its characteristics, and relationship with other related concepts. Scripts represent a schema which contain information on the expected events and behaviours associated with a particular situation (Huesmann, 1988).

Several studies by Huesmann (1988; 1998) and colleagues (Huesmann & Eron, 1984; Huesmann & Guerra, 1997) have identified factors that determine the likelihood of individuals engaging with their scripts, thus predisposing them to aggressive behaviours. It is proposed that the more an individual rehearses the script in mind, the stronger the connections of the script in memory, which facilitates its retrieval in similar or provoked environments (Huesmann & Eron, 1984). In a 3-year prospective study, Huesmann and Eron (1984) investigated the role that cognitive rehearsal of aggressive scripts have on aggressive behaviour. Sampling a primary school children cohort (N = 800) and replicated with samples of over 200 to 300 children in Finland, Israel, and Poland, and Australia it was found that children who engaged in aggressive-type fantasies were more likely to act aggressively, as measured by peer-nominated aggression. Within this study, the concept of fantasies was measured as a child's daydream, nightdream, and imaginary play experience using Rosenfeld et al. (1982) scale of Children's Fantasy Inventory. This experience was what researchers used to determine a child's engagement with an aggressive script, where they proposed that fantasising about aggression in the aforementioned ways was considered a cognitive rehearsal process. This rehearsal process is a core component of the aggressive script model that proposes the more an individual engages or rehearses an aggressive act or fantasises it in mind, the more likely they are to think and act aggressively (Huesmann, 1988; Huesmann & Eron, 1984). A particular area worthy of consideration is the subjective experience of an individual during this rehearsal process of the aggressive script. Hosie et al. (2021) showed that in a sample of incarcerated males with a history of violence (N = 94), a range of emotions were associated with aggressive script rehearsal but that feeling positive towards their aggressive script rehearsal was associated with a greater inclination to aggression. To date, there is limited research that has examined how individuals feel during the rehearsal of aggressive scripts. Rather, extensive research has identified the role fantasies play on mood

states and subjective experiences (Carabellese et al., 2011; Halderman et al., 1985; Klinger, 1990). Exploration of the emotional states associated with aggressive script rehearsal will improve our understanding of this phenomena.

3.1.1 Epidemiology of Aggressive Scripts

The prevalence of aggressive scripts has been explored within several studies, however differences in the measurement of aggressive script experiences has made it difficult to ascertain accurate indication of the extent of the phenomenon. Grisso et al. (2000) found that in a sample of hospitalised acute inpatients (N = 1,136) approximately 30% (n = 339) reported previously experiencing daydreams or thoughts about physically hurting or injuring another person, and that these aggressive scripts occurred more frequently than once a week for about half of patients who reported such thoughts. Similar frequency rates of aggressive scripts were also found by Daff et al. (2015) in a community sample of males with a history of offending (N = 71), with 55% of respondents reporting they had rehearsed aggressive scripts at least several times a year and within the last two months. Both Grisso et al. (2000) and Daff et al. (2015) used the Schedule of Imagined Violence (SIV) to measure the frequency of aggressive script rehearsal.

The experience of aggressive scripts in non-clinical samples of individuals with no known histories of violence has also been reported. Auvinen-Lintunen et al. (2015) found that in a sample of university students (N = 617), approximately 67% of respondents endorsed 'homicidal thoughts', with males reporting these thoughts more frequently than females. These findings are consistent with earlier explorations of homicidal thoughts by Kenrick and Sheets (1993) which also identified that 68% of undergraduate psychology students (N = 312) reported having experienced at least one homicidal fantasy, with males endorsing these fantasies more frequently than females. In both Auvinen-Lintunen et al. (2015) and Kenrick and Sheets (1993) respondents were given descriptions of what homicidal thoughts/fantasies

consist of to assist participants in identifying these thoughts in mind. These findings, which show commonality of aggressive script rehearsal in non-clinical samples, have been identified in further studies including Crabb (2000) and Nagtegaal et al. (2006), which have revealed similar prevalence rates to the studies mentioned prior. Reflecting on the studies which have demonstrated prevalence rates of aggressive scripts in both forensic and non-clinical samples, it appears that the experience of thoughts about harming another person may be considered a common phenomenon which requires further investigation.

3.1.2 Measuring Aggressive scripts

It is suggested that aggressive scripts are activated in situations involving provocation, influencing the individual to think aggressively, and in some cases act aggressively too (Riskind et al., 2007). To date, aggressive scripts are commonly measured using the Schedule of Imagined Violence (SIV; Grisso et al., 2000). Similar to measures of intrusive thoughts, the SIV measures the frequency, recency and content of an aggressive thought, as well as additional details relating to aggressive behaviour (e.g., nature of harm, proximity to target in thought, and the extent to which the thought will escalate or diminish). However, the SIV does not assess the subjective experience associated with the aggressive script, nor does it consider the level of distress or discomfort that may be related. The subjective experience and emotional reaction to thought experiences are a core determinant of intrusive thought phenomenology. Understanding the subjective experience and associated emotional rections of aggressive script rehearsal may aid in the differentiation of this construct from other similar phenomena. The extensive research that has reviewed aggressive scripts using the SIV have documented how such thinking patterns influence aggressive states and violent behaviours within forensic samples (Grisso et al., 2000; Moeller et al., 2017). The research on aggressive scripts is limited however, due to scant investigation of the subjective experience and emotional response to such thoughts. To date, DeLapp et al. (2018) and Hosie et al. (2021) appear to be the only recent studies investigating the emotional experiences associated with aggressive script rehearsal. DeLapp et al. (2018) found that aggressive thoughts were experienced as intrusive and distressing in a student (n = 103) and incarcerated sample (n = 78), and Hosie et al. (2021) identified that a range of emotions including anger, hate, sadness, and confusion were associated with aggressive script rehearsal in a sample of incarcerated males (N = 94). Investigating the extent to which aggressive scripts are experienced as intrusive, the level of emotionality associated, and additional features inclusive of maladaptive beliefs would provide knowledge of how different cognitive factors influence aggressive thinking. Further, understanding the subjective and emotional experiences associated with aggressive scripts may help to clarify whether they are similar or different to AITs.

3.2 Cognitive Theories of Aggressive Scripts

Social cognitive theories have extensively emphasised the role that different cognitive processes have on both the appraisal of aggressive scripts (Bushman & Anderson, 2002; Huesmann & Guerra, 1997). The General Aggression Model (GAM; Bushman & Anderson, 2002) is a social-cognitive model which postulates that aggressive behaviour is largely influenced by a combination of knowledge structures involved in a cognitive information processing sequence which influence the development of aggressive behaviours (Huesmann & Eron, 1984). Another theory that guides understanding of aggressive thinking and its relationships with aggressive behaviour is Script Theory (Huesmann, 1988, 1998) which proposes that behaviour is guided by mental templates (scripts). GAM incorporates multiple specific theories, including Script Theory. Similarly, another model that seeks to explain aggression and its development is the Multiple Systems Model devised by Denson (2013) which posits that angry rumination, conceptualised as "perseverative thinking about a personally meaningful anger-inducing event", plays a role in activating different

physiological, cognitive, affective, and behavioural processes which all interact in maintaining or increasing aggression (Denson, 2013, p. 103). Denson (2013) also suggests that angry rumination may include feelings of anger, or thoughts about revenge – this appears to be the similarity shared with aggressive scripts which concern thoughts of harming others, often in cases of revenge (Grisso et al., 2000). The following section will explore these three theories of aggressive thinking and behaviour relevant to aggressive scripts: the General Aggression Model, Script Theory, and the Multiple Systems Model and address their relationship with aggressive script rehearsal using empirical research.

3.2.1 Script Theory

Script theory (Huesmann, 1988, 1998) proposes that aggressive behaviour is guided by cognitions known as 'scripts'. As a cognitive information-processing model, this theory purports that cognitive scripts, which are learnt through observation or displays of aggressive behaviour, are stored in a person's memory and are used to guide aggressive behaviour (Huesmann, 1988, 1998). A script is said to include both procedural knowledge (i.e., how aggressive behaviour is performed) and declarative knowledge (i.e., factual information regarding aggressive events), and the script is proposed to suggest what the likely outcome of events will be, as well as how one should behave in accordance with the events at play (Huesmann, 1998). Once created, a script is said to develop from a 'controlled' to 'automatic' mental process which becomes resistant to change, particularly when they are frequently rehearsed or enacted (Huesmann, 1998). When a script becomes automatic and is part of an individual's cognitive repertoire, the individual is more likely to accommodate and normalise aggressive behaviour, commonly known as 'normative beliefs' (see below; Huesmann, 1998). These beliefs influence scripts by providing guides for evaluating behaviour, such as deciding whether certain scripts or behaviour are appropriate (Huesmann, 1998; Huesmann & Guerra, 1997). Scripts are proposed to be activated through various situations, including

one's mood state (anger, hostility), arousal, or aggression-related schemas (i.e., perceiving others as holding hostile intentions), through one's interpretation of social cues (e.g., viewing a weapon may activate the retrieval of scripts associated with using weapons; Huesmann, 1998), as well as one's preparedness and self-efficacy in responding non-aggressively. When scripts are frequently rehearsed in mind, they become more easily accessible, and their connection with aggressive behaviour is strengthened.

3.2.2 The General Aggression Model

The GAM postulates that aggression is motivated by aggression-related structures that are stored in memory (e.g., aggressive scripts, normative beliefs about violence) and when activated, contribute to aggressive behaviours (Bushman & Anderson, 2002). This model also acknowledges the influence of social learning over time, and how an individual's experience of aggression, either through observation or performance, contributes to their understanding and future applications of aggression (Bushman & Anderson, 2002). According to this model, an individual's propensity to act aggressively is determined by situational factors (e.g., provocation or anger-inducing event), one's preparedness (e.g., pro-attitudes towards violence), environmental influences (e.g., family practices), and biological determinants (e.g., executive functioning; Anderson et al., 2007). Aggressive scripts are relevant to this model in that they may be activated by anger-inducing provocations, where the thought of harming another person is then maintained by differential normative beliefs about violence (Bushman & Anderson, 2002). This in turn influences an individual's affect by increasing arousal and angry feelings (Bushman & Anderson, 2002). According to Huesmann and Eron (1989), aggressive scripts are stored in one's memory and acts as guides for behaviour. It is postulated that individuals who have a propensity to act aggressively are likely to contain more aggressive scripts in mind and rehearse them more frequently (Huesmann, 1998). Within the GAM it appears that aggressive scripts play a central role in maintaining

aggressive behaviour in one's repertoire by not only guiding aggressive behaviour in angerinduced events, but also by providing individuals with the likely outcomes associated with these behaviours which are based on past experiences of aggression (Gilbert & Daffern, 2017).

3.2.3 The Multiple Systems Model

The Multiple Systems Model emphasises the role angry rumination has on aggressive behaviour, and the implications this type of rumination has on executive control abilities and emotional regulation (Denson, 2013). Specifically, the model explores how angry rumination implicates different systems (i.e., cognitive, neurobiological, affective, executive control, and behavioural) which in turn influences aggression. The Multiples Systems Model differs from the GAM in that its focus is on how and why people ruminate on anger provoking events, rather than merely focusing on why people become aggressive (Denson, 2013). Although aggressive scripts can be induced through the experience of anger-provoking events (e.g., threats to one's ego), it is important to note that aggressive scripts may appear spontaneously and are not restricted to scripts of acting in revenge or retaliation (Patel, 2015). According to Denson (2013), angry rumination can encompass different forms: provocation-focused or self-focused rumination. Denson (2013) highlights how one's engagement with provocationfocused rumination is more likely to increase anger and aggressive behaviour, compared to self-focused rumination, which is concerned with how different events can have implications on oneself, which contrastingly appears to impact one's affect rather than influencing aggression. The conceptual ambiguity that exists between aggressive scripts and anger rumination highlights the importance of considering definitions of these constructs (Hosie, Simpson, et al., 2022). Anger rumination is concerned with perseverative thinking of angerinducing events, or situations of provocation, whereas aggressive scripts represent the aggressive action or plan for aggressive behaviour (Hosie, Simpson, et al., 2022).

To explore the effects of provocation-focused anger, Bushman (2002) divided 600 undergraduate students into three randomly assigned groups: provocation focused rumination, distraction, and control. Participants were insulted and then asked to hit a punching bag. Participants in the provocation-focused rumination condition were asked to think of the insulting person whilst hitting the punching bag, and those in the distraction condition were asked to think about become physically fit whilst hitting the punching bag. Participants in the control group were asked to sit for two-minutes and were not instructed to think about anything specifically. Results from this study found that participants in the provocation-rumination condition reported higher self-reported anger, than participants in the distraction and control conditions (Bushman, 2002). Following from this study's findings, Pedersen et al. (2011) also investigated whether differences in self-reported anger exist depending on the type of anger-rumination process. Pedersen et al. (2011) found that provocation-focused rumination increases self-reported angry affect, however, self-focused rumination saw increases in self-critical negative affect (i.e., feeling disappointed with oneself; feelings of regret). Findings from Pedersen et al. (2011) study suggests that the type of rumination process individuals engage in have an impact on aggression and one's affect.

Where aggressive scripts sit within the Multiple Systems Model is still under investigation, and whether aggressive scripts are synonymous to angry rumination requires further exploration and scrutiny. Current understandings suggest that aggressive scripts and anger rumination share similarities with regards to inducing feelings of anger, and perpetuating thoughts around situations of provocation. However, whether differences exist with regards to the influence anger rumination and aggressive scripts have on the experience of aggressive behaviour requires further exploration.

3.2.4 Normative Beliefs and Attitudes Supportive of Aggression

As highlighted in social cognitive models of aggression and aggressive scripts, normative beliefs and attitudes supportive of aggression appear to play a significant role in the experience of aggressive script rehearsal and subsequent aggressive behaviour. These normative beliefs concern the acceptability of aggressive behaviours towards others in certain situations, and have been found to predict acts of aggression and violence in several forensic research studies (Archer & Haigh, 1997; Mills et al., 2002; Sukhodolsky & Ruchkin, 2004). Two recent studies have identified that specific beliefs and attitudes that endorse aggression are related to aggressive script rehearsal (Gilbert et al., 2013; Podubinski et al., 2017). Gilbert et al. (2013) examined whether normative beliefs and attitudes that support aggression postdicted aggressive behaviour. In a community forensic sample (N = 87), with a history of violent offences, respondents were examined with regards to their rehearsal of aggressive scripts, their normative beliefs which endorsed aggression, early maladaptive schemas, and level of trait anger. These factors were examined alongside respondents' life history of aggression. Findings revealed that 61% of respondents reported rehearsal of aggressive scripts between several times a year to several times a day, and the frequency of one's script rehearsal was related to greater past experiences of aggression. With regards to aggressive attitudes, respondents who endorsed beliefs that condoned aggressive behaviour reported greater involvement in past aggressive behaviours. A significant positive relationship was also identified between aggressive script rehearsal and attitudes towards aggression. Similarly, Podubinski et al. (2017) examined the predictors of aggression by investigating associations with aggressive script rehearsal, attitudes towards aggression, and trait anger. In a sample of 200 non-forensic psychiatric inpatients, it was revealed that aggressive script rehearsal and attitudes towards violence were positive predictors of aggressive behaviour. These findings confirm the influence that normative beliefs about aggression have on

aggressive behaviour in a non-forensic sample, further highlighting the importance that these beliefs, and the rehearsal of aggressive scripts have on aggressive behaviour. Although Podubinski et al. (2017) did not directly examine the relationship between aggressive script rehearsal and attitudes towards aggression, given the positive association identified by Gilbert et al. (2013) between these two factors, it can be expected that such a relationship exists. Both these studies highlight the importance of considering attitudes towards aggression when investigating aggressive script rehearsal as there appears to be a particular connection between these two constructs in influencing aggressive behaviour.

To date, the most widely used measure of normative beliefs is the Measures of Criminal Attitudes and Associates (MCAA) developed by Mills et al. (2002). This self-report measure assesses several domains of Violence (e.g., hitting someone who deserves to bit hit is ok), Entitlement (e.g., I deserve to be treated equally, despite with I have done), Anti-social Intent (e.g., I could lie easily, and be convincing), and Attitudes Towards Associates (e.g., most of my friends have criminal records), as well as the degree of relationship that the individual has with criminal associates. The primary purpose of this measure is to assess dimensions of criminal attitudes to determine their relevance to criminal behaviour (Mills et al., 2002). Currently, this measure is most commonly used in forensic settings with incarcerated samples, where it has been found to be a valid measure of criminal attitudes and behaviour (Mills et al., 2002). Bäckström and Björklund (2008) compared the attitudes of individuals with a criminal history (n = 184) with those of a community sample (n = 556) using the MCAA, and results indicate that both samples report attitudes towards criminality where the differences lie in the extent to which they are endorsed. The sample of individuals with a criminal history demonstrated higher attitudes endorsing criminal behaviour. Gender differences were also apparent, with males in the community sample significantly endorsing more attitudes towards antisocial intent and violence, when compared to females in that

sample. Bäckström and Björklund (2008) findings highlight differences between criminal attitudes in forensic and community samples, however, given the limited breadth of research on the MCAA in comparison studies, further research is warranted.

3.2.5 Additional Cognitive Beliefs Implicated in Aggressive Script Rehearsal

In addition to normative beliefs about aggression which have been found to associate strongly with aggressive script rehearsal and subsequent aggressive action, additional frameworks that explore beliefs within a forensic context have been identified. Although limited in empirical research, some focus has been dedicated to exploring the cognitions that reduce an individuals' propensity to act aggressively or engage in criminal behaviour (Paternoster & Bushway, 2009). Paternoster and Bushway (2009) developed a framework to understand why some individuals may desist from crime and the cognitive factors that may play a role in this process. Desistance is recognised as a process by which individuals develop from an 'offender' to non-offender (Bersani & Doherty, 2018). Paternoster and Bushway (2009) emphasise earlier work by Oyserman and Markus (1990) on the conceptualisation of "possible selves", and the way one's imagination of their positive and negative self may play a role in delinquency. Specifically, Oyserman and Markus (1990) found that youth who reported delinquency were more likely to report their percieved possible future selves as representing negative characteristics such as "depressed", "alone" or a "junkie" (Oyserman & Markus, 1990, p. 121). Although Oyserman and Markus (1990) did not identify the long term influence negative self views have on delinquent behaviour, the study did highlight the future self perceptions of individuals who have engaged in delinquent behaviour.

Extending from Oyserman and Markus (1990) and earlier work by Higgins (1987) discrepancy theory, the concept of the 'feared self' has been identified as an important cognitive factor that appears to have a role in establishing negative beliefs about one's

Bushway (2009) suggest that feared self perceptions include imaginations of the self that the individual does not want to become, and that by holding this negative self in mind, it may induce initial motivations towards desisting from crime. Similarly to what is observed in populations with OCD when feared self-beliefs influence interpretations of AITs, within forensic samples, it is proposed that in order to avoid further offending, individuals must engage with their working identities from a criminal to non-offending one, through the use of beliefs centred on feared selves or feared possible identities (Paternoster & Bushway, 2009). It is postulated that the feared self plays a role in influencing perceptions of what the self might be or become, motivating individuals to avoid these feared selves altogether; thus desisting from crime (Paternoster & Bushway, 2009). This theory proposed by Paternoster and Bushway (2009) has received limited empirical research attention and the role feared self-perceptions in aggression and desistance processes is not entirely clear.

3.3 Aggressive Scripts and other Conceptually Relevant Constructs

3.3.1 Fantasy

Early investigations into aggressive thinking consistently referred to engagements in thoughts of a violent or aggressive nature as fantasies (Crabb, 2000; Kenrick & Sheets, 1993). However more recently, the literature has identified differences between aggressive thinking and aggressive fantasies, which aids in the interpretation and implications of such thoughts (Gilbert et al., 2017). Conceptually, fantasies are understood as mental pictures actively created by an individual which combine multiple components into a script (Rokach, 1990). It has been argued that fantasies are not experienced exclusively due to external events or behaviours like aggressive scripts, where it is purported that fantasies provide substitutes and/or preparations for action (Beres, 1960).

However, one's engagement with their fantasy has been described similarly to the processes of general script rehearsal (Beres, 1960). It is proposed that fantasising can be considered akin to script rehearsal; Huesmann and Eron (2013; 1989) argue that regularly rehearsed aggressive scripts can be maintained in one's cognitive repertoire through the process of fantasising, inevitably increasing one's potential to act aggressively. The process of using fantasies as a means of consolidating or elaborating an aggressive script is emphasised by Huesmann (1998). Fantasies may strengthen the connection of the aggressive script in one's mind which further eases the processes of access and retrieval when the script is required – which is elaborated in the GAM by Bushman and Anderson (2002). Gilbert and Daffern (2017) elaborate that fantasising, as a process of mental rehearsal, may help to change or shape the content of aggressive scripts, which may have implications on subsequent aggressive behaviour by influencing either more creative and/or fixed ways of behaving.

Given the term fantasy and aggressive scripts have been used interchangeably in the literature, it has led to some confusion in the nomenclature of these constructs. It is worth noting that a common conception of fantasies, and often aggressive thoughts more generally, is that their subjective experience is a positive one (i.e., particularly those relating to sexual contents; Veale et al., 2009), however limited research has explored the subjective experience of aggressive scripts more generally, particularly in relation to the subjective emotional experience during the script rehearsal. As summarised by Hosie, Simpson, et al. (2022), fantasies are conceptualised as perseverative thinking, which can involve elaboration or rehearsal of an aggressive scripts. Whereas aggressive scripts represent the action or behaviour being considered in the situation at hand. Given that some studies have referred to aggressive thoughts as fantasies (Grisso et al., 2000) and others as aggressive scripts (Hosie

et al., 2014), for the purposes of this thesis the term 'aggressive scripts' will be used to denote thoughts about harming another person.

3.3.2 Anger Rumination

Anger rumination is defined as repetitive thoughts individuals have about an anger inducing event which maintains a state of anger arousal (Denson, 2013; Sukhodolsky et al., 2001). Findings confirm a relationship between anger rumination and anger (Bushman, 2002), as well as anger rumination and aggressive behaviour (Peled & Moretti, 2009). Anger rumination can be conceptualised as thoughts about anger that are unintentional and recurrent which occur during or after an anger inducing event (Sukhodolsky et al., 2001). Anger rumination can be measured by the Anger Rumination Scale (ARS; Sukhodolsky et al., 2001), which assesses the tendency to think about anger and its related experiences.

Questions on the ARS include thinking about angry mood, thoughts of revenge, recalling anger provoking experiences, and thinking about the antecedents and consequences of anger experiences.

In a cross-sectional study of undergraduate students, Anestis et al. (2009) examined the effect of anger rumination across different forms of aggression: trait physical aggression, trait verbal aggression, hostility and anger. Using the ARS and the Buss-Perry Aggression Questionnaire Subscale, findings confirmed the association between anger rumination and aggression. Specially, anger rumination was found to significantly predict physical aggression, verbal aggression, and hostility even after controlling for covariates including depression, anxiety, impulsive behaviour, and emotional regulation. An unexpected finding was that anger rumination was not significantly related to anger, which is inconsistent with previous research that suggests anger rumination maintains anger arousal and affect (Denson et al., 2012). Anestis et al. (2009) purport that experiencing higher levels of anger may not

always influence anger rumination, and that anger rumination may induce an angry affect but this may not lead to higher levels of anger more generally.

Further, in a sample of incarcerated males (N = 29), Hosie, Simpson, et al. (2022) explored the relationships between anger rumination, aggressive script rehearsal, and aggressive behaviour. Results demonstrated a strong relationship between aggressive script rehearsal and anger rumination, namely the Thoughts of Revenge subscale. Hosie, Simpson, et al. (2022) also identified a moderate relationship between anger rumination and aggressive behaviour. These findings highlight the similarities in phenomenology between aggressive scripts and anger rumination, namely the rehearsal of retaliatory plans of aggression within the context of provocation.

According to the Multiple Systems Model (Denson, 2013), anger rumination appears to differ from aggressive scripts as it does not include ways one will act aggressively, rather it influences or maintains angry affect. Although anger rumination does not include plans to act aggressively, it has been found to maintain anger arousal as well as influence individuals to dwell on past anger experiences, thus increasing ones propensity to act aggressively (Pedersen et al., 2011). Anger rumination has been identified as a dysfunctional emotional regulation process as well as a risk factor for aggressive behaviour (Denson, 2013). Therefore, anger rumination appears to be a relevant construct to the experience of aggression more generally, and therefore will be investigated alongside aggressive scripts in the current thesis.

3.3.3 Aggressive Intrusive Thoughts versus Aggressive Scripts

Of particular interest to this thesis are the features that best differentiate between the experience of AITs and aggressive scripts. The specific features that are best able to differentiate these phenomena have not been reliably investigated, and to date there is no empirical research exploring AITs and aggressive scripts simultaneously. As highlighted

previously, current definitions of AITs, and intrusive thoughts more generally, include the experience of thoughts about harming another person that are frequent, spontaneous, intrusive, disrupt functioning, are associated with distress, and are ego-dystonic to the individual (Moulding, Aardema, et al., 2014a). They are a common symptom of OCD, and due to these aforementioned features, AITs are not associated with acts of aggression, nor is their presence considered a risk factors for aggressive behaviour (Fairbrother et al., 2022; Veale et al., 2009). Definitions of aggressive scripts include any thought or daydream about physically harming another person. Aggressive scripts may also be experienced frequently and spontaneously (Grisso et al., 2000; Patel, 2015; Sheldon & Patel, 2009), and there is research to suggest that aggressive scripts can be experienced as intrusive, distressing, and difficult to control (Patel, 2015). The clear difference between AITs and aggressive scripts appear to be the behavioural outcomes associated, where aggressive scripts have been reliably reported to be associated with a history of aggressive behaviour (Gilbert et al., 2017), and used as a means of predicted one's propensity to act aggressively (Hosie et al., 2021)

Research in aggressive script rehearsal is commonly conducted with samples of individuals who have a history of violent offences, and these studies provide evidence for the prevalence of these thoughts and the relationship these have with violent behaviour (Gilbert et al., 2013; Moeller et al., 2017). AITs on the other hand are a common symptom of OCD, and their key feature is the frequent experience of unwanted thoughts is associated with significant distress, and the contents of the thought are experienced as ego-dystonic to the individual (Moulding, Aardema, et al., 2014a) Similar to studies AITs, aggressive scripts have also been identified in non-clinical samples, however the breadth of research is limited.

The few studies that have explored aggressive scripts in non-clinical samples confirm that such thoughts are a common phenomenon that occur outside forensic samples (Grisso et al., 2000; Nagtegaal et al., 2006). In a development and validation study of the SIV, Grisso et

al. (2000) explored the prevalence of violent thoughts in a comparison study of students (n =519) and individuals hospitalised for a mental illness (n = 1,136). It was found that several individuals in both groups reported the experience of violent thoughts, where the prevalence of such thoughts were found to be higher in the clinical sample (i.e., clinical sample = 30% vs. students = 14.5%). Similarly, Nagtegaal et al. (2006) investigated the frequency to which aggressive scripts are experienced in a non-clinical sample of university students. Using the SIV(Grisso et al., 2000) it was found that approximately 60% (N = 72) of students reported that at some point in their life, they experienced a thought about physically hurting or injuring another person. Both Nagtegaal et al. (2006) and Grisso et al.'s (2000) studies provide evidence for the occurence of aggressive scripts in the general population. When considering the implications of aggressive script rehearsal, it is important to also consider the presence of other factors including violence supportive beliefs and a history of aggressive behaviour, which in combination predict the likelihood of aggressive behaviours (Daff et al., 2015; Gilbert et al., 2013). These studies on aggressive scripts also raise the question as to whether the aggressive thoughts reported in Nagtegaal et al. (2006) and Grisso et al. (2000) are indeed aggressive scripts or could they be classified as AITs. Given the SIV does not enquire whether respondents have experienced their aggressive scripts as intrusive or distressing, it is unclear whether aggressive scripts are indeed different in phenomenology to AITs. While there has been anecdotal assumptions that aggressive scripts are distinct from AITs, these assumptions may have led to the creation of different measurement tools that suggest differences these constructs. Further, there have not been any empirical studies comparing whether AITs and aggressive scripts differ in terms of phenomenology based on the experience of certain features (e.g., intrusiveness, distress).

In order to understand the process through which AITs and aggressive scripts influence differing outcomes for an individual, exploring features of maladaptive belief

systems (Radomsky et al., 2014; Rowa & Purdon, 2003; Sukhodolsky & Ruchkin, 2004) and life-time experiences of aggression (Coccaro et al., 1997; Gilbert et al., 2013) may prove useful in understanding this development. Specifically, utilising features from the established research field of AITs and intrusive thoughts in OCD, exploring whether aggressive script rehearsal are associated with features such as intrusiveness, spontaneity, ego-dystonicity, ego-syntonicity, thought control, and the emotional valence may prove important in differentiating between AITs and aggressive scripts. Currently the distinction between AITs and aggressive scripts is not clear, and it is difficult to ascertain whether the subjective experience of either cognitive processes share similar components, or whether they are in fact separate constructs.

3.4 Treatment for Aggressive Script Rehearsal

While the scope of the thesis was to elucidate the phenomena of AITs and aggressive scripts, a treatment section for aggressive script rehearsal has been included to provide context for some of the factors relevant to aggressive script rehearsal that have become targets of intervention. While this treatment section explores the broad intervention options available for aggressive scripts, this section aims to contextualise some of the implications of aggressive script rehearsal addressed in chapter 9 of the thesis.

Empirical research examining the effectiveness of specific treatment interventions for aggressive scripts is scarce. Extant literature exists for treatment guidelines and interventions for conceptually related constructs such as rumination, fantasies, and intrusive thoughts (Clark, 2004; Hvenegaard et al., 2015), and the application of these treatment principles for aggressive scripts has been recommended (Gilbert & Daffern, 2017). For example, rumination-focused cognitive behavioural therapy aims to modify the process of thinking through a combination of behavioural activation, and analysis of the behavioural function of rumination (Hvenegaard et al., 2015). As explored in a previous chapter of this thesis,

intrusive thought treatment focuses on challenging and modifying faulty appraisals and maladaptive beliefs, and this can specifically be achieved through exposure and response prevention strategies (Clark, 2004). Whether components of these treatment interventions that come from conceptually related constructs to aggressive scripts are applicable to treatment of aggressive script rehearsal requires exploration. Research on the effectiveness of treatment programs designed for violent offending suggests that the impact these programs have on violent recidivism is mild (Papalia et al., 2019). Papalia et al.'s (2019) meta-analytical review suggests that the implementation of programs that offer multimodal treatments have the strongest treatment efficacy on violent recidivism.

In a sample of incarcerated adults males, Morrison (2022) compared the effectiveness of a novel group-based aggressive script rehearsal treatment program, developed specifically for the study, with an emotional regulation treatment program. Findings from the study revealed that although there was no reduction in the frequency of aggressive script rehearsal overtime across either treatment programs (i.e., at 46 days, and at 104 days post-treatment), Morrison (2022) found a significant reduction in self-reported aggressive behaviour related to aggressive script rehearsal, as well as reduced emotion regulation difficulties. Given Morrison (2022) was one of the first studies to address aggressive script rehearsal in a treatment program, further research into the development and implementation of interventions for aggressive scripts is required.

3.5 Summary

This chapter summarised the construct of aggressive scripts including its phenomenology and epidemiology. This chapter also considered several social cognitive models of aggression, namely the GAM, Script Theory, and the Multiple Systems which offer interpretations of script rehearsal and subsequent aggressive behaviour. Distinctions between related constructs of aggressive scripts were also explored, and issues related to

nomenclature were identified. This thesis is particularly interested in aggressive scripts, and although distinctions have been made with related constructs, these constructs will be revisited in further chapters as they are relevant to investigations of aggression more generally. This chapter also explored some overlap between the construct of AITs and aggressive scripts, including how they are both commonly reported by the general population and concern thoughts about harming or injuring another person. This chapter concluded with a section on the current treatment options available for aggressive script rehearsal.

CHAPTER 4 - A CRITICAL COMPARISON OF AGGRESSIVE INTRUSIVE THOUGHTS IN OBSESSIVE COMPULSIVE DISORDER AND AGGRESSIVE SCRIPTS IN OFFENDER POPULATIONS

4.1 Preamble for Critical Review Paper

The previous chapters have presented current understandings of AITs and aggressive script rehearsal from two distinct lines of research. In order to further examine the potential similarities and differences between these two phenomena, the following chapter attempts to integrate literature from AITs and aggressive script rehearsal research and critically review the features of these constructs. This chapter aims to provide a critique of the literature whilst comparing well known features of intrusive thought phenomenology from OCD research with aggressive script rehearsal. Prior research has established features pertinent to intrusive thoughts in OCD, however, the features of aggressive script rehearsal requires further empirical investigations. It was therefore accepted that intrusive thought features would be compared and critical reviewed with aggressive script rehearsal.

The critical review addresses the first thesis aim, to explore and establish the phenomenology of AITs and aggressive scripts. The critical review is one of its first in concurrently examining AITs and aggressive script rehearsal, and comparing differential features of these phenomena. The following chapter has been submitted to a journal and is awaiting review. It is therefore presented in its publication format.

4.2 Author Indication Form for Empirical Research Paper One



Swinburne Research

Authorship Indication Form

For PhD by Publication candidates

	ded in your thesis.
DECLARATION	
We hereby declare our contribution to the publication	on of the 'paper' entitled:
A critical comparison of Aggressive Intrusive Thou Aggressive Scripts in Offender Populations	ughts in Obsessive Compulsive Disorder and
First Author	٨
Name_ Stephanie Fernandez	Signature: Signature:
Percentage of contribution: 85_%	Date: 20 / 02 / 2023
Brief description of contribution to the 'paper' and y	rour central responsibilities/role on project:
Reviewing and synthesizing literature, preparation	of manuscript
Second Author	- 200
	Signature:
	V
Percentage of contribution: 5%	Date: 21 / 02 / 23
Brief description of your contribution to the 'paper':	
Supervision on literature search and focus, review	wed and edited manuscript drafts
Third Author	
Name: Richard Moulding	Signature:
Percentage of contribution:5%	Date: 20/ 02 / 2023
Brief description of your contribution to the 'paper':	
Supervision on literature search and focus, review	
Fourth Author	
Name: Maja Nedeljkovic	Signature: Ollkay
Percentage of contribution: 5 %	Date: 20 / 02 /2023
Brief description of your contribution to the 'paper':	

In the case of more than four authors please attach another sheet with the names, signatures and contribution of the authors.

Authorship Indication Form

Principal Coordinating Supervisor:

Name: ___ Maja Nedeljkovic

Date: 20 / 02 / 2023

A Critical Comparison of Aggressive Intrusive Thoughts in Obsessive Compulsive Disorder and Aggressive Scripts in Offender Populations

Stephanie Fernandez^{1,2}, Maja Nedeljkovic¹, Michael Daffern^{1,2}, & Richard Moulding³.

¹Department of Psychological Sciences, Swinburne University of Technology, Melbourne,

Australia

²Centre for Forensic Behavioural Science, Swinburne University of Technology, Melbourne,

Australia

³The Cairnmillar Institute, Melbourne, Australia

Corresponding author: Stephanie Fernandez, Centre for Forensic Behavioural Science, 582 Heidelberg, Alphington, Victoria, Australia, 3078; Email: sfernandez@swin.edu.au

Contact number: 0421 325 795

Acknowledgements

Declarations of interest: none

Declaration: This manuscript has not been published elsewhere and has not been submitted simultaneously for publication elsewhere.

Abstract

Thoughts about harming or injuring another person are a common phenomenon and can be understood through two different constructs: aggressive intrusive thoughts or aggressive scripts. The outcomes associated with these thoughts differ significantly depending on the population group they are investigated in. Within Obsessive Compulsive Disorder, aggressive intrusive thoughts are not associated with acts of aggression, rather they are experienced as significantly distressing to the individual and may influence a range of obsessive-compulsive symptoms to reduce the perceived consequences of the intrusive thought. In offender populations, aggressive scripts are associated with aggressive behaviours and inform examinations of risk assessment and intervention. Whether similarities exist between these phenomena remains unclear as these two constructs have not been compared, conceptually or empirically. The current review explores whether these two phenomena share similarities with regards to well established features from intrusive thought research, including frequency, intrusiveness, influence of maladaptive appraisals, associated emotional experiences, and thought control strategies. The relationship these features have with aggressive scripts is explore may prove beneficial for not only differentiating between these two phenomena but for informing risk assessment and intervention for those who experience them.

Keywords: aggressive intrusive thoughts, aggressive scripts, obsessive compulsive disorder, aggression, cognition

Thoughts about harming or injuring another person are commonly experienced by the general population (Kenrick & Sheets, 1993; Rowa & Purdon, 2003). In most instances these thoughts do not forecast subsequent aggressive action and in some cases, individuals may go to extraordinary lengths to prevent themselves from acting aggressively and causing harm to others (Veale et al., 2009). However, aggressive thoughts are common in violent offenders and associations between aggressive thoughts and violence has been reliably reported in offender populations (Daff et al., 2015; Gilbert et al., 2013). Furthermore, antisocial cognitions are amongst the most important risk factor for criminal behaviours (Andrews et al., 2011). Currently, while the literature considers these to be separable phenomena, the actual differences have not been subject to systematic empirical review.

Two understudied constructs that have been used in the literature to describe aggressive thought experiences are Aggressive Intrusive Thoughts (AITs) and Aggressive Scripts. These two phenomena share similarities in terms of thought content; however, they may be associated with very different behavioural consequences. AITs are a common in Obsessive Compulsive Disorder (OCD), in which repetitive and unwanted thoughts about causing harm or injury to another person lead to high levels of distress and worry (Rachman, 1997). It is the general consensus that the experience of AITs within the context of OCD are not associated with aggressive behaviour (Veale et al., 2009). Rather individuals who experience these thoughts within the context of OCD resort to compulsive behaviours that aim to reduce negative affect and the perceived consequences associated with the thought about harming a loved one or another person (Rachman, 1997). Although the experience of anger states has been reported in individuals with OCD (Moscovitch et al., 2008; Radomsky et al., 2007), the experience of anger is attributed to the high level of distress associated with the thought, rather than reflecting tendencies toward aggressive behaviour (Whiteside & Abramowitz, 2005). The level of distress, ego-dystonic nature of AITs (i.e., inconsistent with

one's beliefs and value system; Purdon et al., 2007), and compulsive behaviours aimed at reducing perceived negative consequences have been argued to relate to the low risk that exists for an individual with OCD to act on these thoughts (Veale et al., 2009). Aggressive scripts by contrast, are conceptualised as thoughts about acting aggressively toward another person. They are mental templates for aggressive behaviour which are created from situations where aggression or violence is observed or displayed (Huesmann & Eron, 1984). Once formed, aggressive scripts are maintained through rehearsal, which is recognised as ones engagement with their aggressive thoughts (Huesmann & Eron, 1984). Aggressive script rehearsal has been shown to relate to aggressive behaviour (Gilbert et al., 2013).

AITs and aggressive scripts share similarities in terms of the content of the thought; however, there is a general acceptance that the experience of these two types of thoughts result in different behavioural and emotional outcomes for the individuals who experience them (Daff et al., 2015; Veale et al., 2009). There is however limited understanding of the key distinguishing features of these phenomena both in research and in clinical applications, including how these types of thought implicate clinician rated risk assessments in those who experience them (Veale et al., 2009). This may be due to the disparate lines of research which have explored these two constructs. Our understandings of AITs is grounded in OCD research, which has identified key characteristics that distinguish AITs from other thought phenomena. These include the unwanted, intrusive, frequent, distressing, and ego-dystonic (i.e., inconsistent with ones' belief system) nature of AITs (Moulding, Aardema, et al., 2014a). It is argued that the phenomenology of AITs may provide protective qualities against overt acts of aggression (Veale et al., 2009). The features of unwantedness, intrusiveness, ego-dystonicity, and distress have received limited attention in the aggressive scripts' literature. Whether aggressive scripts share these same qualities as AITs is unclear. There is little research exploring script rehearsal amongst violent offenders and it is unclear whether,

in this population, these experiences have the same features as AITs that are experienced by people with OCD (i.e., that they are intrusive, unwanted, ego-dystonic).

To the best of the authors' knowledge, to date, no comparisons have been made between AITs and aggressive scripts with regards to these features and it is unknown whether similarities or differences exist in the phenomenology of these thoughts. Direct comparisons between AITs and aggressive scripts, including the similarities and distinguishing features, may facilitate the identification of potential protective or risk factors for aggressive behaviours. This may also have clinical implications, specifically in the assessment of risk for aggressive behaviour, where distinguishing between AITs or aggressive scripts is vital in the implementation of risk assessments but also in the treatment of the cognitions. For example, in the Historical Clinical and Risk Management-20 v3, a structured clinical judgement instrument used for violence risk assessment, one important risk item relates to aggressive thinking. It is unclear whether all aggressive thoughts would be regarded similarly in this tool (i.e., AITs and aggressive scripts) or if thoughts that are intrusive, unwanted, distressing, egodystonic would not be associated with risk for violence. Future research may be guided by this exploration through the identification of how the experience of aggressive thoughts may influence differences in aggressive behaviours.

The goal of this literature review is to critically evaluate empirical findings regarding the characteristics of AITs and aggressive scripts, while identifying whether these characteristics are comparable across the two phenomena. The review begins with a conceptual overview of AITs and aggressive scripts. The review is then broken down into sections that outline specific features of AITs which are then compared to current understanding and measurement of aggressive scripts. This is due to the more developed nature of OCD literature, and the extensive exploration of intrusive thought characteristics.

Each section explores the existing empirical literature relating to the given feature, from nonclinical, OCD, and forensic research domains.

A targeted systematic review was deemed inappropriate due to the nature of the review and the lack of specific literature in the area. However, to inform this critical review, we used Boolean search methods and keywords to inform the search strategy (i.e., aggressive intrusions, aggressive obsessions, aggressive thoughts, repugnant thoughts, unwanted thoughts, aggressive scripts, violent thoughts, intrusions). In July 2021, the following databases were searched using the key words: CINAHL via EBSCOhost, MEDLINE, and PsychINFO. References from articles found via these searches were also used. Papers were included in the review if they contained the aforementioned keywords. Meta-analyses, dissertations, and reviews were not excluded.

4.3 Conceptual Overview

4.3.1 Aggressive Intrusive Thoughts

AITs are thoughts about harming or injuring another person and are considered personally repugnant, unwanted and are highly distressing (e.g., thoughts about stabbing a family member or partner; Moulding, Aardema, et al., 2014a). They are a common feature of OCD where approximately 45% of individuals with OCD have been found to report thoughts of aggression (N = 292; Pinto et al., 2008a), although rates vary depending on the sample (e.g., ranging from 13% to 45%; Brakoulias et al., 2013; Pinto et al., 2008a; Rowa et al., 2005). The general population have also been found to experience AITs; for example, Bouvard et al. (2017) found no difference between individuals with OCD and non-clinical participants in the prevalence of their aggressive intrusions which is consistent with prior findings that have compared these two population groups (Garcia-Soriano et al., 2011). When considering symptom dimensions that have been identified for OCD, AITs have generally been classified under the symptom dimension of repugnant thoughts, also known as

'unwanted thoughts' (Bloch et al., 2008; Brakoulias et al., 2013). In addition to aggressive thoughts, the repugnant thoughts dimension of OCD also includes sexual and religious obsessions. The key feature of this dimensions is the unacceptable and forbidden themes that the content of the thoughts involve, where these thoughts are experienced as ego-dystonic (i.e., inconsistent with ones beliefs) and contradict one's sense of self (Moulding, Aardema, et al., 2014b). As the presence of AITs is perceived as abhorrent to the individual, it often influences compulsive behaviours or neutralising strategies that help to deal with the anxiety and fear associated with the thought, to prevent the harm associated with the thought, or to reassure oneself that the negative content is not representative of self (Moulding, Aardema, et al., 2014b). For example, a mother or father who experience recurrent and unwanted thoughts about harming their infant child may utilise overt compulsive strategies including constant reassurance seeking, or avoidance of their child (Fairbrother & Woody, 2008). In these instances, seeking reassurance or avoiding the child reduces the distress and anxiety associated with the aggressive intrusion as well as preventing the perceived harm associated with the intrusion (Brakoulias et al., 2013).

Emergent from the literature and for the purposes of this review, AITs are operationalised as thoughts about harming or injuring another person that are experienced as unwanted, intrusive, spontaneous, ego-dystonic (i.e., inconsistent with one's sense of self), and distressing, and are associated with compulsive or neutralising behaviours. AITs are distinguished from thoughts that are classified as rumination or worry as such phenomena are generally considered to be ego-syntonic. As Clark (2005) has highlighted, the appraisal process involved with AITs often concerns the implications these thoughts have for one's personality and responsibility over the actions; rather than the mere concern that these thoughts may become true - as is seen with worrisome thoughts.

4.3.2 Aggressive Scripts

Aggressive scripts are conceptualised as daydreams or thoughts about physically harming or injuring another person (Grisso et al., 2000). They are generally thought to be created through the exposure to aggression; most commonly in early childhood experiences (Huesmann & Eron, 1984). Aggressive scripts include both procedural knowledge (i.e., how aggressive behaviour is performed) and declarative knowledge (i.e., factual information regarding aggressive events), which are used as guides for aggressive behaviour at a later stage (Huesmann, 1998). Once created, an aggressive script is stored in memory and retained through mental rehearsal such as thinking, daydream, and fantasising (Huesmann & Eron, 1989). It is this mental rehearsal process that is commonly understood as someone experiencing thoughts (in the form of an aggressive script) about harming another person. The more these aggressive scripts are rehearsed, the stronger the connection of these scripts with other aggressive-related constructs (e.g., condoning beliefs about violence; Huesmann, 1998); this in turn increases the likelihood of subsequent aggressive action.

Several attempts have been made to differentiate aggressive scripts from aggressive fantasies (Gilbert & Daffern, 2017). Fantasies have been conceptualised as an internally generated thought process where an individual combines multiple symbols into a script, which is then used as a substitute or preparation for behaviour (Rokach, 1990). Both aggressive scripts and aggressive fantasies are said to play a role in the preparation and guidance of aggressive behaviour (Gilbert & Daffern, 2017), and there is empirical evidence to suggest that they both are influenced by the modelling of aggression, past experiences of aggression, and maladaptive beliefs condoning aggression (see review by Gilbert & Daffern, 2017). However, it has been noted that the literature on aggressive scripts and fantasies has been confused by the tendency to use the terms 'script rehearsal' and 'fantasy' in an interchangeable way, as well as the different methods used to measure these constructs

(Gilbert & Daffern, 2017). Given that research has used general definitions for both constructs (e.g., thoughts about harming another person), this has made it especially difficult when reviewing the literature to differentiate between the two constructs. Regardless of this, implicit in the discussion of both aggressive scripts and fantasies is the individual's deliberate engagement with these thoughts. However, limited empirical evidence exists confirming this proposition (Patel, 2015). Given this, the current review has operationalised aggressive scripts as any thoughts about harming or injuring another person that appear to influence aggressive states or aggressive behaviour.

An individual's attitudes towards, and beliefs about aggression have also been found to play a role in how they evaluates their own behaviour and that of others, as well as helping to normalise the experience of aggressive scripts (Huesmann & Guerra, 1997). These normative beliefs about aggression include cognitions such as 'someone who makes me angry deserves to be hit' and 'it's ok to hit someone who insults you' (Mills et al., 2002, p. 249). These types of maladaptive beliefs are commonly measured in offender samples who present with aggressive histories and antisocial behaviour (Gendreau et al., 1996; Mills et al., 2002). Research by Gilbert et al. (2013) and Kelty et al. (2011) demonstrate evidence of the relationship between attitudes towards aggression and script rehearsal, where individuals who endorse beliefs that condone aggressive are more likely to rehearse aggressive scripts. Understanding this relationship is essential in identifying the types of dysfunctional beliefs that may exacerbate or maintain aggressive script rehearsal, and in turn influence aggressive action. Dysfunctional belief patterns may prove an important target point for treatment interventions in offender samples, particularly those with a history of aggressive script rehearsal and aggressive behaviours.

The following sections compare aggressive scripts to well established characteristics of intrusive thoughts. Each section will comprise of a specific feature, or group of features,

pertinent to intrusive thoughts, which are then compared to the current understandings and measurements of aggressive scripts. Identifying the similarities and differences between these phenomena may have implications for the assessment and treatment of these constructs in a range of settings. The research reviewed in these sections derive from non-clinical, clinical OCD, and forensic research domains.

4.4 Features of Intrusive Thoughts: Comparisons with Aggressive Scripts

4.4.1 Unwanted, Intrusive and Spontaneous

Aggressive Intrusive Thoughts

Early conceptualisations of intrusive thoughts by Rachman and colleagues (Rachman, 1978, Rachman & Hodgson, 1978) defined intrusive thoughts as thoughts which are unacceptable and experienced as unwanted. The unacceptable and unwanted quality of the thoughts pertain to the content of the thought (e.g., aggression/violence, sexual themes), as well as to how the thought enters the mind without encouragement (Clark & Purdon, 1995; Rachman, 1981). Both Klinger (1996) and Rachman (1981) speak to the unintended and spontaneous nature of intrusive thoughts which often results in difficulties with thought suppression and negative affect due to the emotional response associated with the content and unwilful appearance of the thoughts in mind.

Definitions of intrusive thoughts emphasise the unwanted and intrusive nature of these thoughts, which become relevant when distinguishing them from normal thought experiences. The concept of intrusiveness has not been specifically defined, nor does a specific measurement exists which delineates if a thought is 'intrusive' or not. Rather several factors have been collated and investigated together to uncover whether thoughts can be deemed as 'intrusive'. As elucidated by Clark and Purdon (1995), early measurements of unwanted intrusive thoughts primarily emphasised the way intrusive thoughts were experienced as 'intrusive' through understanding process characteristics (e.g., spontaneity,

entering the mind unwillingly). For individuals to discern whether their intrusive thoughts are indeed classified as "unwanted intrusive thoughts", descriptions of what intrusive thoughts consist of and the different ways they may be experienced (e.g., as thoughts, images, or impulses) are provided to ensure individuals can identify this phenomenon consciously (Clark & Purdon, 1995; Pascual-Vera et al., 2019). For example, when determining whether an individual is experiencing intrusive thoughts with aggressive contents, a description of what the intrusive thought may look like is provided (e.g., a thought about harming or injuring another person), and the individual is made aware that the thought appears suddenly in mind, and therefore is considered unwanted and upsetting or unpleasant (Pascual-Vera et al., 2019). Rachman (1981) highlights that any thought, image or impulse can be considered an unwanted intrusive experience, as long as it is associated with distress, experienced recurrently, and feels ego-dystonic to the individual (i.e., inconsistent with one's beliefs). Rachman (1981) also suggested that the individual experiences difficulty with sustaining ongoing concentration or activity due to the presence of the intrusive thought, and the thought is difficult to control. This further reiterates the importance of considering several factors together when determining whether a thought is 'intrusive'. This process of determining whether a thought is intrusive assumes that the individual is able to consciously identify a thought and uses the provided definitions of 'intrusive thoughts' in distinguishing them from other normal thought experiences.

The experience of intrusiveness was also highlighted by O'Neill et al. (2009), who explored intrusive thoughts in inmates (n = 79) and students (n = 86). A semi-structured interview for intrusive thoughts was utilised where participants were asked to report "any readily identifiable intrusive thoughts or impulses" they experienced as well as the frequency, spontaneity, dismissability and distress levels associated with the thoughts (O'Neill et al., 2009, p. 149). Findings from this study showed that inmates reported significantly less

intrusive thoughts than the student sample, and inmates with higher levels of psychopathic traits reported significantly less intrusive thoughts overall. Differences in content of the intrusive thoughts were not directly compared across groups, however as anticipated, the inmates' intrusions were exclusively aggression related. Majority of students (64.9%) and inmates (61.1%) reported being able to easily dismiss their intrusive thought, and no significant differences were observed when comparing distress of thoughts across the two samples. However, it is worth noting that for the student sample, as intrusive thoughts were experienced more frequently, distress levels increased – this was not observed in the inmate sample. A strength of the study is that O'Neill et al. (2009) provided participants with a 10minute interactive introduction about the concept of intrusive thoughts, which ensured participants understood what constituted an intrusive thought. The findings re-iterate how differences in intrusive thoughts are observed across samples, particularly the impact that underlying factors (e.g., psychopathic traits) have on the way the intrusive thoughts are experienced. These findings suggest the role that one's subjective account, and experience of several factors (e.g., level of distress, spontaneity) have on what is considered intrusive, rather than intrusiveness measured as a single construct.

The spontaneous or non-volitional aspect of intrusive thoughts has been incorporated into definitions of this phenomena, namely by Klinger (1990) and Rachman (1981) which emphasise that the presence of these thoughts occur without an intended purpose. Even though there are reports of individuals experiencing intrusive thoughts as a result of evoking stimuli, for example a thought of stabbing a loved one after seeing a knife, the connection between the evoking stimuli and the AIT is likely to reflect an illogical and unrealistic situation for the individual (O'Connor et al., 2009). This speaks to the difficulty of understanding the connection between environmental stimuli that may trigger the occurrence of intrusive thoughts, as this process may be highly dependent on the subjective state of the

individual and the symbolism that is assigned to external precipitants of the intrusive thoughts. For example, whether individuals are influenced by external triggers (e.g., a knife or sharp object) is dependent on how salient these triggers are to the person but also how they are interpreted and appraised by the individual (e.g., 'I must be careful around a knife as I could harm someone'). The spontaneous quality of intrusive thoughts emphasises that such thought experiences are not prompted by the person, nor are they actively engaged with.

Unwanted, intrusive, and spontaneous

Aggressive Scripts

Although the content of an aggressive script is similar to AITs (i.e., thoughts about harming or injuring another person), whether the script is experienced 'intrusive' is currently unclear. The concept of intrusiveness has not been directly examined with respect to aggressive scripts, and thus little is known about how offenders may experience thoughts that may be classified as 'intrusive'.

In a qualitative study of violent offenders which examined the function and features of aggressive scripts and violent fantasies, Patel (2015) found that several respondents identified that their experience of an aggressive script was intrusive. This was identified by the respondents' descriptions of their 'disturbing, unwanted and unpleasant experiences' that were followed by suppression or distraction strategies (Patel, 2015, p. 161). Respondents' aggressive script experiences included some similar characteristics to AITs including that they were: disturbing, unwanted, and unpleasant; however, some features still remain unclear. These include whether the aggressive scripts were experienced recurrently, spontaneously, as distressing, and were experienced as ego-dystonic. These findings from Patel (2015) do argue against a potential misconception that violent thoughts are experienced pleasantly by forensic samples (Veale et al., 2009), as Patel (2015) identified that the presence of an aggressive script was sometimes described by offenders as unwanted and unpleasant. Patel's (2015)

study is one of the first to show an association between aggressive scripts and intrusiveness. However, as noted by Patel (2015), it is currently unclear whether offenders did, in fact, experience intrusive thoughts similar to those described in OCD literature, as respondents were not provided with a description of what an intrusive thought consists of (e.g., a thought that suddenly appear in our minds, interrupts what we are doing, is difficult to control, is upsetting, unpleasant, and disturbing) — which has proved to be important in measuring and characterising this phenomena (Clark & Purdon, 1995; Pascual-Vera et al., 2019). It is therefore difficult to disentangle whether (2015) sample of offenders experienced aggressive scripts as 'intrusive', akin to what is described with intrusive thought phenomena in OCD, or if they experienced their aggressive scripts as phenomena reflecting other cognitive processes (e.g., intrusive memories in PTSD or rumination).

Whether aggressive scripts enter the mind spontaneously or whether they are summoned into the mind deliberately has received some attention in the forensic psychology literature. While limited, there has been some empirical research that suggests aggressive script rehearsal may be experienced spontaneously. Sheldon and Patel (2009) conducted qualitative interviews with 25 male patients with violent convictions, who had been admitted to a high security forensic hospital in the United Kingdom. Sixteen of the 25 assessed offenders identified an aggressive script, and thematic analyses identified some offenders experienced them as spontaneous, where others described their rehearsal of aggressive scripts as deliberately self-generated. A limitation of these findings was that offenders did not explicitly report their violent thoughts as spontaneous, rather inferences were made by the authors in the thematic analyses of the findings. The deliberate generation of violent thoughts that were reported by offenders presents as a stark contrast to what is commonly seen within OCD. Nevertheless, these findings highlight how aggressive scripts can be experienced both

spontaneously and deliberately self-generated, but whether these features influence differences in behavioural outcomes remains unclear.

4.4.2 Frequent, recurrent and not easily dismissed Aggressive Intrusive Thoughts

The frequency of AITs has been investigated in several studies of OCD (Belloch et al., 2004; Clark & Purdon, 1995; Purdon & Clark, 2001), where it has been found that individuals with OCD experience their intrusive thoughts more frequently than non-clinical subjects. Rachman and De Silva's (1978) seminal paper on the differences between abnormal and normal obsessions compared the experiences of intrusive thoughts between a clinical sample of obsessional patients and a non-clinical sample of undergraduate, postgraduate, and professional individuals. Results from a clinical structured interview with respondents found that intrusive thoughts experienced by the individuals with OCD lasted longer in one's mind and appeared more frequently than intrusive thoughts in the non-clinical sample. These findings are consistent with cognitive models of OCD, where despite intrusive thoughts occurring in the general population (Radomsky et al., 2014), the implications these thoughts have on an individual's mood is dependent on the way the thought is appraised (Salkovskis, 1985). It is this appraisal process that has been found to influence the frequency with which individuals experience intrusive thoughts, as well as the likelihood of them recurring and persisting (Moulding, Coles, et al., 2014; Salkovskis, 1985).

The frequency of AITs specifically has received limited attention in the literature where inconsistencies currently exist regarding how often individuals may experience intrusive thoughts about harming or injuring another person. This is due to the different measures used across the literature to examine AITs. Investigations of the frequency of AITs have found that approximately 75% of students (N=64) reported ever experiencing an intrusive thought related to harm, aggression, or sexual impulses (Rowa & Purdon, 2003). In

contrast, Radomsky et al. (2014) found that students across 13 different countries (N = 777) endorsed AITs far less frequently than other intrusive thought contents (e.g., doubts). This discrepancy in frequency of AITs speaks to the difficulty in understanding not only how common these thoughts are in the general population, but also at a cross-cultural and individual level (i.e., how often individual's experience these thoughts on a day-to-day basis).

In studies of AITs, DeLapp et al. (2018) found that in a sample of inmates (n = 78) and college students (n = 103), the frequency of AITs did not differ across groups, although they did not provide detail regarding how frequent these experiences were for respondents. This finding aligns with the idea that intrusive thoughts are a universal experience (Purdon & Clark, 1994b), however they are inconsistent with O'Neill et al. (2009) who found inmates endorsed intrusive thoughts significantly less frequently than students. DeLapp et al. (2018) suggests that they discrepancy with their results are due to their use of a novel and more sensitive measure of AITs that was developed for the study which provided respondents with a list of AITs across a range of aggressive content themes. This is seen in developed measures of intrusive thoughts such as the Questionnaire of Unpleasant Intrusive Thoughts which provides respondents with detail on the type and form of intrusive thought being explored (Pascual-Vera et al., 2019). This suggests that future research should use AIT measures that comprehensively define and provide respondents with information regarding their key characteristics (e.g., intrusiveness, unwanted, spontaneous). It is anticipated that this process will allow for a more accurate identification of the AIT phenomena.

Frequent, recurrent, and not easily dismissed

Aggressive Scripts

The forensic literature has extensively studied the impact that frequent thoughts about harming or injuring another person have on an individual, particularly the relationship with aggressive behaviours (Grisso et al., 2000). Several studies have explored the frequency in

which aggressive scripts occur in the general population (Grisso et al., 2000; Kenrick & Sheets, 1993), but limited research exists examining how frequently these scripts are experienced at an individual level. Kenrick and Sheets (1993) investigated the duration that aggressive scripts remain in one's mind, finding that males reported thoughts of harming or injuring another person lasted longer in their mind (i.e., a few minutes) than did females, who reported their aggressive script as fleeting and lasting only a few seconds. Gender differences in the frequency of aggressive scripts was also explored by Auvinen-Lintunen et al. (2015) in a university student sample, where 28% of males and 15% of females (N = 617) reported experiencing a thought about harming/killing someone in the past week, while the majority of the sample reported no experiences of aggressive thoughts in the past 2 months.

The relevance of aggressive scripts to aggressive behaviours is said to be influenced by the frequency with which these scripts are rehearsed in one's mind. Specifically, Grisso et al. (2000) found that hospitalised forensic patients who reported experiencing recent aggressive scripts were more likely to perpetrate violence after discharge. This finding is consistent with current understandings of script rehearsal where ongoing engagement in thoughts of harming others is purported to strengthen and/or maintain the aggressive tendencies in a person's repertoire (Huesmann, 1988). These conceptualisations of script rehearsal demonstrate how the feature of frequency plays a significant role in the development of aggressive behaviour, where the frequency of aggressive scripts appears to condition an individual towards the normalisation of aggressive or violent behaviours. This is in stark contrast to what is seen with aggressive intrusive thought (AITs) phenomenology, where the frequency of AITs is not believed to see an individual's propensity for violence or aggressive increase (Veale et al., 2009).

4.3.3 Distressing versus pleasant subjective experiences Aggressive Intrusive Thoughts

The subjective experience associated with intrusive thoughts has been extensively investigated in the OCD literature. One feature that distinguishes normal intrusions, which are those commonly experienced by the general population, from clinical obsession in OCD is the extent to which the intrusion is associated with distress or discomfort (Rachman & de Silva, 1978). The concept of distress, as it is understood in OCD, appears to relate to thoughts containing unpleasant or disturbing contents, as well as the personal meaning individuals derive from the thought experience (Belloch et al., 2004). For example, a person who has thoughts of harming others may believe these thoughts signify that there is a bad, immoral or dangerous hidden aspect of their persona, and thus contribute to a distressed affect (Aardema et al., 2017; Ferrier & Brewin, 2005).

Why some thoughts are deemed more upsetting or distressing than others has been investigated within relevant literature, specifically how thoughts classified under the domain of repugnant/unacceptable thoughts are associated with distress (Rowa et al., 2005). Rowa et al. (2005) investigated why certain intrusive thoughts are experienced as upsetting to individuals with OCD (N = 28). It was found that 43% of the sample identified their most upsetting intrusion to include content relating to aggressive, religious, or sexual themes, and that the occurrence of these intrusions were related to an increase in distress. Participants also appraised these intrusions as indicating that they needed to exert control over their thoughts, as well as believing the presence of the thought in mind signified importance and thus should be attended to (Rowa et al., 2005). Similar conclusions were also found in a three-part study conducted by Lee and Kwon (2003) who compared the differences in subjective experience between repugnant thoughts (*autogenous* obsessions) and non-repugnant intrusive thoughts (*reactive* obsessions). It was found that autogenous thoughts were rated as being more disturbing and they were associated with greater feelings of unacceptability and guilt. Lee and Kwon's (2003) measurement of subjective experience did not ask participants about their

levels of distress associated with their own repugnant thoughts, which therefore creates difficulty in comparing similar studies that have operationalised distress measurement. Additionally, Lee and Kwon (2003) did not measure the subjective experience of specific autogenous thoughts (e.g., aggressive/harm intrusive thoughts), but rather a cluster of repugnant obsessions measured collectively. Measuring repugnant obsessions collectively does not allow for specific associations to be made between aggressive intrusive thoughts and subjective experiences. It is therefore unknow whether respondents in Lee and Kwon's (2003) study felt guilty after noticing their thoughts of harming others, or whether guilt was related to thoughts of other content themes (e.g., sexual or immoral thoughts). Being able to differentiate between the level of distress across different intrusive thought contents will allow for better understandings of how different content themes may implicate one's subjective experience.

While there has been much literature on intrusive thoughts and their associated subjective experiences, literature that has specifically focused on aggressive intrusions and their impact on one's emotional experience is limited. This trend is also observed in forensic psychology research, where several studies have overlooked the emotional reaction that people have to their thoughts of harming or injuring others. The assumption that aggressive thoughts reported by forensic samples are associated with positive emotional experience requires further scrutiny and validation as limited empirical evidence exists confirming this presumption. More recently, DeLapp et al. (2018) explored the association between distress and AITs in a non-clinical (n = 103) and violent incarcerated sample (n = 78). This study indicated that there were no differences in distress levels between the two samples, where increases in distress were observed when the frequency of AITs increased. The similarities in distress levels for intrusive thoughts across student and incarcerated samples was also observed by O'Neill et al. (2009), where no differences were found in the level of distress

associated with intrusive thoughts or impulses across groups. However, it is currently unclear from DeLapp et al.'s (2018) and O'Neill et al.'s (2009) studies whether offender populations experience distress from intrusive thoughts due to their content, level of intrusiveness, or other related factors akin to those that are seen in non-clinical and OCD populations. Further research is required to examine how differences in distress levels may be related to the content type of an intrusive thought as well as to other factors such as level of intrusiveness, frequency, and appraisal.

Distressing versus pleasant subjective experiences

Aggressive Scripts

To the best of the authors' knowledge, there is limited understanding of the subjective states that occur in forensic populations when aggressive scripts are experienced. Some studies have identified the negative impact aggressive fantasy rehearsal has on subjective well-being (Poon & Wong, 2021), where other studies have highlighted the increase in positive affect that occurs from rehearsing thoughts of harming others (Patel, 2015). This conflict in research findings is further complicated by the minimal research that has been dedicated towards exploring the relationship between the emotional experiences associated with aggressive scripts.

In a community sample of participants (N = 113), Poon and Wong (2021) explored the effects of prolonged rumination of aggressive fantasies on participants' subjective well-being, as measured by their responses on the Satisfaction with Life Scale and the Positive and Negative Affect Schedule. Findings revealed that participants who fantasised about harming another person reported lower levels of subjective well-being than participants who did not engage in aggressive script rehearsal. The implications of aggressive script rehearsal on one's emotional experience has also been investigated by Patel (2015), where a sample of 48 offenders from a Forensic Mental Health Hospital in the United Kingdom were examined to

identify qualities of their aggressive scripts and aggressive fantasies. The exploratory nature of Patel's (2015) study meant that offenders were not provided with a pre-determined definition of what constitutes an aggressive thought or aggressive fantasy, but rather respondents were given the opportunity to define and describe their understanding of the concepts in their own words. Qualitative analyses identified that offenders experienced aggressive scripts as being more distressing than aggressive fantasies. Offenders operationalised aggressive scripts as thoughts embedded in reality that were provoked by environmental stressors (e.g., being threatened or bullied). Aggressive fantasies on the other hand, were described as creations of fiction and associated with enjoyment and pleasure, for example, causing harm to others as part of revenge. Offenders reported not wanting to experience an aggressive script, as they were characterised as provoking anxiety and feelings of unease. Offenders described that this emotional response was resultant from the belief that aggressive thoughts were part of reality, and they thus were perceived as having an increased risk regarding the thought becoming true.

Patel's (2015) findings align with current understandings of intrusive thoughts and distress, where cognitive models highlight the influence that interpretations of thoughts can have on the subsequent emotional experience (Rachman, 1997). In contrast to the understanding of obsessions in OCD research was the finding that some offenders who experienced aggressive scripts believed that they were more likely to act on them. This aligns with conceptualisations of aggressive scripts where the rehearsal and the continual engagement with aggressive thoughts increases their connections with aggressive behaviours (Huesmann, 1998). It is currently unclear whether the anxiety and feelings of unease described by offenders as occurring when aggressive scripts were experienced decreased the likelihood of aggressive acts, as this was not examined by Patel (2015) or any other studies to date. Patel (2015) noted that only a few offenders reported an increased likelihood of acting

on their aggressive thoughts, and it is not clear whether they reported negative subjective experiences (e.g., anxiety and unease) when these thoughts were rehearsed initially. Further examining the role that negative affect and subjective experiences may play in the rehearsal of aggressive scripts would allow for a clearer understanding of how these thoughts may be distinguished from AITs. A point worthy of consideration when interpreting Patel's (2015) findings is the difficulty in ascertaining whether the experiences of aggressive scripts that were considered unpleasant, distressing, and anxiety provoking were in fact intrusive thoughts (i.e., like those commonly reported by non-clinical and OCD populations) or aggressive scripts (i.e., those commonly investigated in forensic research). Thus, further research is warranted in identifying whether aggressive intrusive thoughts experienced by offender populations are associated with negative subjective states.

Recent research has explored the emotional sequalae to aggressive script rehearsal, where Hosie et al. (2021) identified in a sample of 131 prisoners that anger was the most common emotion associated with script rehearsal, followed by hate, fear, sadness, confusion and annoyance. Hosie et al. (2021) also revealed that offenders who had been physically aggressive towards another person more than 10+ times were found to be more likely to feel excited and more in control when rehearsing their aggressive scripts, than offenders who had been physically aggressive towards another person no more than nine times. Auvinen-Lintunen et al. (2015) compared the differences between genders in emotional responses to aggressive script and found that females reported experiencing stronger negative emotions (i.e., anger, anxiety, feeling confused and afraid) and males reported feeling a sense of humour towards their aggressive scripts. When interpreting these findings it is important to note that Auvinen-Lintunen et al. (2015) measured aggressive thoughts by asking respondents "Have you ever had homicidal fantasies?", which was described by Auvinen-Lintunen et al. (2015) as a limitation of their study. The use of these terms may have coloured

respondents' subjective experiences, as the terms 'homicidal' and 'fantasies' may be sensitive areas of exploration for respondents. Additionally, the use of the term 'fantasies' may explain the differences in the emotional reactions that males and females reported, where males may have experienced humorous feelings by perceiving fantasies as fictional and only as imagined situations, which aligns with current definitions of fantasies (Burgess et al., 1986). The findings from the aforementioned studies that explored the relationship between aggressive scripts and emotional experiences have identified how one's subjective experience appears to be a relative phenomenon that is dependent on how aggressive scripts are interpreted or perceived at an individual level. For some, aggressive scripts cause significant anxiety and trepidation, and for others they appear to be sources of pleasure or amusement. Exactly why these differences may exist requires further exploration.

4.4.4 Control and neutralising strategies

Aggressive Intrusive Thoughts

Phenomenological explorations of AITs have established that such thoughts are experienced as uncontrollable. Uncontrollability, alongside factors of content themes and subjective distress have been found to relate to the experience of aggressive intrusions (Rachman, 1981). Cognitive behavioural theories of OCD suggest that control and neutralising strategies are resorted to by an individual in order to prevent an unwanted thought from becoming true and to reduce the individual's negative affect (Salkovskis, 1989). Differentiating neutralisation strategies from compulsions, which are a central component of OCD phenomenology, has proved difficult. Clark (2004) suggested that compulsions are ritualistic, repetitive behaviours that are generally fixed ways of responding to an unwanted intrusion (e.g., repetitively washing one's hands), whereas neutralisation is described as a flexible way of behaving that aims to remove the unwanted thought from mind and to prevent the perceived consequences of the thought. Neutralisation is conceptualised as a way of

cancelling out or undoing the perceived effect of the unwanted thought (e.g., replacing the image of harming another person with a pleasant positive thought).

Several studies have been conducted examining the different control strategies used to neutralise repugnant intrusions, however little research has examined AITs specifically. Berman et al. (2012) examined the different control strategies used when harm intrusions were experienced by students, specifically when AITs that centred on a vulnerable victim were evoked. Participants were placed into two groups, a 'strong condition' where they were asked to identify an able-bodied victim, and a 'weak condition' where they were asked to identify a vulnerable victim (i.e., elderly person aged between 60 -100). The AIT was evoked by participants reading and completing the following sentence with their specified victim: "I hope gets into a car accident this week and ends up in critical care". Participants who thought of a vulnerable victim experienced greater distress, increased feelings of moral wrongness, and a need to neutralise the thought experience. Berman et al.'s (2012) findings also provide support for the different appraisals used in interpreting unacceptable thought experiences, whereby participants in the 'weak condition' believed they needed to exert control over their thoughts as well as believing that there was a greater possibility for the thought becoming true. With regards to control strategies, over half of the study sample engaged in neutralisation of thoughts, where no differences were found between groups in the frequency of neutralisation strategies. Covert strategies such as engaging in a mental neutralisation techniques (e.g., counting, thinking about something opposite, praying) were most common across both groups. Berman et al. (2012) interpreted the frequent use of covert neutralisation strategies as a possible way for participants to avoid judgement from the experimenter. Additionally, given the intrusive thought experience was induced by experimenters in this study, it raises issues regarding the ecological validity of the thoughts compared to spontaneous intrusive thoughts. It is also worth noting the relevance of these

findings to individuals with OCD whereby Parsons et al. (2017) found that the thought task of imagining a vulnerable victim did not elicit distress that was unique to OCD. These findings suggests that this thought task may evoke distress in participants beyond what is seen in individuals with OCD. Nevertheless, Berman et al.'s (2012) findings provide some indication of how thoughts that include harm towards a vulnerable victim are more likely to influence negative subjective states and promote the use of neutralisation strategies.

Similarly to Berman et al.'s (2012) findings, Belloch et al. (2004) found that in order to control their most upsetting intrusive thought, a student sample (N = 334) used a variety of control strategies with the three most common including reasoning with self, covert distraction, and covert neutralisation. Differences between gender were identified, where women were found to employ all the aforementioned strategies more frequently than men. The findings of both these studies show how individuals respond to experiences of unwanted intrusive thoughts. Strategies employed to manage intrusive thought experiences have been vastly researched in the OCD literature and it has been elucidated that the type of strategies used by individuals is highly idiosyncratic, does not always follow a linear trend, and is dependent on factors including the frequency, distress, and appraisal associated with the thought (Freeston et al., 1995; Freeston et al., 1991; Moulding, Coles, et al., 2014; Purdon & Clark, 1994a). It has long been proposed that the use of control and neutralisation strategies may directly influence the recurrence and frequency of intrusive thoughts, in addition to serving the immediate function of relief and reduction in one's emotional distress (Clark, 2004; Salkovskis, 1989). Although, the efficacy of these strategies in reducing the recurrence and frequency of intrusive thought through has been debated (Clark, 2004). In the context of AITs, Salkovskis (1985) describes neutralisation as an individual's attempt to reduce or avoid the responsibility of harm occurring to others. There are several types of strategies, both overt and covert, that individuals may employ to control, neutralise or supress their unwanted

intrusive thoughts, with the five most common including: (1) distraction (e.g., thinking about something else), (2) social control (e.g., talking to a friend about the thought), (3) worry (e.g., dwelling on other worries), (4) punishment (e.g., punishing self for thinking the thought), and (5) reappraisal (e.g., reinterpreting the thought; Luciano et al., 2006).

Amir et al. (1997) explored differences in the types of thought control strategies utilised by individuals with OCD (n = 55) and a non-clinical sample (n = 27). Using the original Thought Control Questionnaire (TCQ) developed by Wells and Davies (1994), Amir et al. (1997) found that participants with OCD used more punishment, worry, reappraisal, and social control strategies to manage their intrusive thoughts, than did the non-clinical participants. The use of distraction was most frequently reported by non-clinical participants. Amir et al. (1997) also found that the use of punishment was a clear discriminator between participants with OCD patients and the non-clinical sample, where punishment strategies were used less frequently by non-clinical participants. The use of punishment, as well as worry, was also found to significantly correlate with the severity of obsessive-compulsive symptoms, which is consistent with previous findings that have highlighted the impact that punishment has on stress, anxiety and psychopathology (Wells & Davies, 1994). These findings seem to suggest that the use of certain thought control strategies may be maladaptive, in turn contributing to the return of unwanted intrusive thoughts rather than eradicating them from one's mind.

Although neutralisation and control strategies reduce the discomfort associated with the unwanted intrusive thought, some negative long-term consequences may arise from these behaviours. A study by Ahern et al. (2015) provided support for the immediate implications of using neutralisation strategies, where significant reductions in distress were evident when a covert neutralisation strategy was used subsequent to an unwanted intrusive thought.

However, consistent with previous research in this area (Clark, 2004; Wells & Davies, 1994),

when an unwanted intrusive thought was evoked at a later stage, respondent's level of distress and the need to neutralise increased (Ahern et al., 2015). These findings suggest that the neutralisation may leave an individual vulnerable to further distress when an intrusion is experienced at a later point in time. Additionally, the evidence remains unclear whether the use of thought control strategies effectively manages the experience of unwanted intrusive thoughts in the long term.

Control and neutralising strategies

Aggressive Scripts

The investigation of the use of control and neutralisation strategies for aggressive scripts is limited. Nagtegaal et al.'s (2006) study explored the association between aggressive scripts and the use of different thought control strategies in a sample of students (N = 72). Approximately 60% of the sample reported aggressive script rehearsal, and thought suppression, distraction, and cognitive reappraisal were found to be frequently used by students, when compared with social coping, worry, and punishment strategies. Some thought control strategies were significantly associated with aggression, as measured by the Aggression Questionnaire, where punishment, suppression, worrying, and reappraisal were positively linked with hostility and general anger displays, whereas distraction was found to be negatively linked with aggression (i.e., reduced the likelihood of anger displays). The finding that individuals using control strategies other than distraction to suppress aggressive scripts are more likely to experience aggression is consistent with the notion that most techniques commonly reported as thought control strategies are not effective in controlling or eradicating aggressive thought experiences.

Similar to the AIT research, thought control methods that employ a punishment technique appear to produce negative outcomes (Wells & Davies, 1994), whether it be an increase in one's negative affect or leaving one vulnerable towards aggressive behaviours

(Bushman, 2002). Nagtegaal et al.'s (2006) finding that the use of distraction acts as a functional strategy for supressing aggressive scripts is consistent with previous research on distraction techniques which has found this method to be an effective emotional regulator (Bushman, 2002). Specifically, distraction has been found effective in regulating angry affect; however, no association has been found with reducing aggressive tendencies (Bushman, 2002).

Few studies have directly investigated the motivations behind using thought control strategies for aggressive scripts. Patel's (2015) qualitative study of violent offenders (N = 48) explored the way in which violent thoughts and fantasies were managed. It was found that in general, managing violent thoughts were motivated by reducing the risk of violent behaviour, which was prompted by the need to be careful. Offenders described the management of these thoughts through methods such as avoidance (e.g., physically removing self from provoking situation), distraction (e.g., engaging in an activity that alters focus from provoking situation), and social coping (e.g., sharing the violent thought with staff to receive help and support with the negative experience). These findings share similarities to the motivations seen by individuals with OCD where the use of control strategies is motivated by the belief that the aggressive thought may become true.

It was also highlighted by Patel (2015) that some offenders felt an urge to act out the violent thought, with the belief that this would help eliminate or eradicate the thought experience. Other offenders also identified a belief that directing the violence towards themselves through deliberate acts of self-harm would be productive in managing the expression of the violent thought. These findings share similarities with Amir et al.'s (1997) results, where individuals with OCD were found to frequently report using punishment techniques as a primary method for thought suppression. However direct comparisons should be cautioned as the unwanted intrusive thoughts individual suppressed in (1997) study were

not specific to aggressive contents, and thus the motivations for punishment strategies may serve different purposes to that which is observed in offender samples. Whether similarities exist in the thought control strategies employed when thoughts of harm or injury to others are experienced by individuals with OCD and offenders currently remains unclear.

4.4.5 Self-Themes: ego-dystonicity & the feared self Aggressive Intrusive Thoughts

Several cognitive models of OCD have been created to explain the impact different self-themes and individual self-perceptions may have on the experience of intrusive thoughts within OCD (Doron & Kyrios, 2005; Doron et al., 2008). This section will review the existing literature on different self-themes that have been implicated in obsessive-compulsive symptoms, specifically the role of ego-dystonicity, feared self-perceptions, and sensitive self-domains.

A feature that has been used in OCD research to characterise clinical obsessions is ego-dystonicity. Ego-dystonicity refers to the extent that the content of an intrusive thoughts is inconsistent with one's sense of self, contradicting an individual's values, ideals, and morality (Clark, 2004). An ego-dystonic thought may include an intrusion about harming a loved one, where the individual experiences this thought as reflecting inconsistencies with their sense of self, and outside the thoughts they would expect to have. A systematic review by Jaeger et al. (2021) revealed OCD-relevant intrusive thoughts and obsessions were ego-dystonicity, and that the experience of beliefs that imply a feared or immoral self are strongly related to the presentation of obsessive-compulsive symptoms. Ego-dystonicity aligns with Rachman's (1997) cognitive model of obsessions where it is proposed that unacceptable thoughts that encompass moral wrongness (e.g., aggression, sex, blasphemy), become problematic and may escalate to clinical obsessions when the individual appraises these thoughts as representing an unknown or abhorrent part of themselves. The individual in turn

assigns personal significance to the content and meaning of the intrusive thought (Rachman, 1997). This interpretation becomes problematic for the individual. Ferrier and Brewin (2005) found that individuals with OCD interpreted their unwanted obsession as reflecting negative inferences of their personality. These ego-dystonic intrusive thoughts evoke distress in the individual and are met with resistance which precipitates the need to use control and neutralising strategies. This is particularly relevant to intrusive thoughts that concern harm, violent, sexual, or abhorrent themes, whereby one is likely to interpret these intrusions are repugnant and immoral, and experienced as alien to their sense of self (Clark, 2004).

The Ego Dystonicity Questionnaire (EDQ) is the primary measure of ego-dystonicity. The EDQ comprises items that measure the extent to which an intrusive thought is experienced to the individual as undesirable (e.g., do not want it to come true), irrational and inconsistent with one's personality (e.g., not the kind of thought I would expect), and immoral (e.g., against what is right). In a sample of individuals with OCD (N = 28), Purdon et al. (2007) found that the experience of ego-dystonic thoughts surprisingly was negatively related to obsessive compulsive symptom (OCS) severity. This finding suggests that a non-linear relationship between ego-dystonicity and OCD may exist, whereby changes in levels of OCS may occur when more chronic and frequent ego-dystonic thoughts are experienced (Purdon et al., 2007). Purdon et al.'s (2007) findings are consistent with previous accounts of ego-dystonicity, where Purdon and Clark (1999) have suggested that one may become accustomed to their ego-dystonic thought, incorporating what they appraise from the thought into their personal self-view.

Self-themes have received increased attention within OCD research, where interest lies in the way that individuals' self-perceptions are linked to the frequency, and maintenance of intrusive thoughts (Doron et al., 2008). It has been proposed that "sensitive" self-domains, which are areas of life where one lacks confidence, may be areas in which the individual is

vulnerable to thoughts that are experienced as ego-dystonic and unacceptable (Doron & Kyrios, 2005; Doron et al., 2008). This vulnerability to unwanted thoughts may also be seen in individuals who hold an ambivalent self-view which includes the individual being preoccupied with a self-view that is dichotomised as being either "good" or "bad" (Bhar & Kyrios, 2007). Ambivalence towards the self-concept renders an individual vulnerable to intrusive thoughts that are appraised as immoral or sensitive. In addition to these sensitive self-themes, a construct known as the "feared self" has been explored in relation to the appraisal process of unwanted intrusive thoughts (Aardema et al., 2017). Specifically, the feared self includes attributes that an individual fears becoming and/or possessing, which are believed to sit hidden beneath one's personality or self-concept (Aardema et al., 2017). When the feared self is implicated, individuals may appraise their unwanted intrusive thoughts as representing components of their self-concept that are bad, immoral, or dangerous, which contribute to distress and negative obsessive compulsive symptoms (Melli et al., 2016). Investigations of the feared self have identified that intrusive thoughts with repugnant and unacceptable content themes such as aggressive intrusions relate most strongly with the feared self construct (Aardema et al., 2017; Aardema et al., 2013). Wong et al. (2020) found that by inducing feared self perceptions in a virtual reality-based paradigm, participants experienced aggressive intrusive thoughts and the urge to neutralise their thoughts more frequently than those participants who did not have induced feared self-perceptions. Further, Melli et al. (2016) found that the feared self was strongly associated with high levels of importance placed on repugnant thoughts, as well as the belief that the thought needs to be controlled. For example, an individual who experiences an AIT about harming a loved one or friend may appraise this intrusion as revealing a negative aspect of their persona; the feared self (e.g., I am a dangerous person). Further, the individual may assign over importance to the presence of this intrusion and engage in neutralising strategies to control the thought or to

prevent the perceived consequences of the thought. Additionally, this individual may also experience fear of who they are or might become, based on the abhorrent content of the AIT itself (Rachman, 1997). Research on the feared self suggests that not only is the content of the thought experienced as ego-dystonic, but the inferences individuals make appear to implicate sensitive self-domains (Ferrier & Brewin, 2005; Rachman, 1997).

Self-themes: ego-dystonicity & the feared self

Aggressive Scripts

Within the context of aggressive scripts, there has been little empirical investigation into self-domains and their implications for the experiences of aggressive and violent thinking. Although associations have identified between thoughts of harming others and selfperceptions within OCD research (e.g., ego-dystonicity and the feared self), it is unclear whether these self-themes (i.e., ego-dystonicity and the feared self) are also implicated in aggressive script rehearsal. In studies of the self within offender research, the role that negative self-perceptions may have on the experience of aggressive scripts has not been directly explored, rather the effect that self-views have on desistance (i.e., abstaining from crime) has been highlighted (Paternoster & Bushway, 2009). Offender research appears to focus on self-views after offending has occurred, postulating that offenders view their selfconcept as criminal, and the feared self plays a role in motivating change towards a prosocial desired self (Paternoster & Bushway, 2009). It is highlighted that offenders wanting to abstain from crime may experience aversion towards future crime due to the negative inferences made from their feared self-perceptions. (Paternoster & Bushway, 2009). As such, current understandings of the feared self in forensic research suggests that the effect of negative self-perceptions may be a motivator behind reducing recidivism; however, whether the feared self-influences the experience of aggressive scripts and their rehearsal has not been directly examined. Similar to what is experienced when ego-dystonic thoughts become

accustomed to an individual's self-view after frequent experiences of the thought, Patel (2015) suggests the possibility of violent thoughts becoming a part of an offenders self-identity when they serve a function to the individuals' needs, such as self-regulation. Patel (2015) highlights that the experience of violent thoughts and fantasies have origins in childhood experiences. This early experience may overtime normalise the presence of violent thoughts and fantasies in mind, which in turn would contribute to the incorporation of these thoughts into one's self-view. Similar to the process that is observed with intrusive thoughts, Patel (2015) identified that some offenders acknowledged their violent thoughts as socially and morally inappropriate. However, while there are potential similarities, one difference between intrusive thoughts in OCD and aggressive scripts in offender samples is the functional properties that aggressive thinking serves the individual (e.g., emotional regulation). It appears that for offenders this functional aspect is as an important factor in their engagement with their aggressive thoughts and fantasies (Patel, 2015).

There has been some research with OCD that has identified the relationship between repugnant thoughts (e.g., those that concern harm, sexual, or blasphemous themes) and feared self-perceptions (Aardema et al., 2013; Aardema et al., 2021; Aardema et al., 2019), but how these feared self-beliefs related to aggressive thoughts specifically remains unclear. Currently, the feared self is measured using a revised 8-item version of Fear of Self Questionnaire (FSQ; Aardema et al., 2013) where items explore the negative self-perceptions one may hold resultant from the experiences of unwanted intrusions (e.g., I am afraid of the kind of person I might become). Whether these perceptions are applicable to the experience of aggressive script rehearsal requires investigation. Current understandings of how offenders perceive their self-concept suggest that the concept of feared or possible selves may be implicated, particularly regarding re-offending and future risk (Paternoster & Bushway, 2009). It is therefore worth investigating whether feared self-perceptions are related to the

experience of aggressive script rehearsal to further understand the role these self-perceptions may have in deterring individuals from engaging in future acts of aggression. By examining the relationship the feared self may have in offender samples using a validated measure of this construct is therefore warranted.

4.5 Summary

In this review, the features of aggressive intrusive thoughts (AITs) have been explored and compared with aggressive scripts. Despite the commonality that exists with the content of AITs and aggressive scripts, it has been suggested in this review that not all features of aggressive scripts can be viewed from the lens of intrusive thoughts of OCD literature. This is due to the lack of empirical investigations of these features within the forensic literature and specifically with aggressive scripts, as well as the complexities in the measurement of aggressive scripts and AITs more generally. What has been made clear by reviewing the literature is the possibility of offender populations experiencing aggressive scripts, to some extent, in similar ways to intrusive thoughts in OCD. The following table provides a summary of the dimensions of aggressive intrusive thoughts and how they are comparable to aggressive scripts.

Table 1. Differentiating Characteristics of Aggressive Intrusive Thoughts and Aggressive Scripts

Aggressive Intrusive Thoughts	Aggressive Scripts
Unwanted & intrusive	• Unwanted? Intrusive? (unclear)
• Spontaneous	• Rehearsed
• Frequent & recurrent	• Frequent & recurrent
 Distressing 	• Distressing? Emotional regulation function
Compulsions or control strategies	• Thought control strategies

- Externally triggered
- No evidence of past aggressive behaviours or experiences
- Influenced by provoking environments
- Influenced by past experiences of aggression

The similarities appear within the subjective experience of aggressive scripts, specifically the way these thoughts of harming others occur frequently and spontaneously, induce distress, and discomfort, and motivate the use of different thought control behaviours. There is also some evidence that suggests negative self-perceptions and feared self-domains may also be implicated in offender populations, however the direct relationship with aggressive scripts remains unknown. Further research is required within the area of aggressive thinking, where the subjective experience of aggressive thoughts either in clinical or forensic domains have implications for the individual who presents with them but also for the clinicians treating them. Future research may be guided by the lack of empirical research that directly compares AITs and aggressive scripts based on their features. Additionally, empirical research is warranted in areas that investigate issues in measurement of these constructs, where current measures appear unable to clearly differentiate between the two phenomena.

PART II –EMPIRICAL RESEARCH: OVERVIEW, METHODOLOGY AND FINDINGS

CHAPTER 5 – EMPIRICAL RESEARCH INCLUDED IN THIS THESIS

The previous chapter provided a critical review of the literature that explored common features of intrusive thoughts in OCD. The chapter integrated known features of intrusive thoughts and compared these features of aggressive script rehearsal, using literature from forensic research domains. Further, to address the overarching research thesis aims and contribute towards rectifying gaps in the literature as outlined in chapter one, two empirical studies were conducted. The research examined the experience of AITs and aggressive script rehearsal, specifically exploring the prevalence amongst non-clinical and clinical populations. Additionally, relationships between AITs, aggressive script rehearsal, and dysfunctional cognitions such as obsessive beliefs and violence supportive beliefs were examined. The association between AITs, aggressive script rehearsal, and a history of aggressive behaviour were also examined. Finally, the subjective experiences associated with AITs and aggressive script rehearsal were explored. The manuscripts of these empirical studies are described below including each of the manuscripts aims and hypotheses.

5.1 Empirical Study One

The first empirical study aimed to address thesis aim one; to explore the similarities and differences in the phenomenology of AITs and aggressive scripts. The first aim of the study was to investigate the differential cognitive predictors of AITs and aggressive scripts using a non-clinical sample. More specifically, general and specific cognitive predictors related to OCD, and general beliefs related to violent attitudes and aggression were investigated to address this aim. The contributions of the feared self, thought control strategies, and ego-dystonicity were also examined. Behavioural outcomes measured by a retrospective recollection of a history of aggression, and obsessive compulsive symptoms

were examined to determine their relationships with maladaptive beliefs. The following hypotheses were made:

- (1) It was hypothesised that ego-dystonicity, self-ambivalence, and the fear of self would predict AITs.
- (2) It was hypothesised that violent attitudes, anger rumination, and life history of aggression would predict aggressive scripts.
- (3) It was hypothesised that general OCD beliefs and thought control strategies would predict the experience of obsessive-compulsive symptoms.
- (4) It was also hypothesised that anger rumination and violent attitudes would predict a life history of aggression.

5.2 Empirical Study Two

The second empirical study aimed to address thesis aim two, to explore the whether the experience of AITs and aggressive scripts is comparable, for both clinical and forensic populations. The first aim of the study was to examine the subjective experience associated with AITs and aggressive scripts by examining the role associated beliefs and features (e.g., intrusiveness, spontaneousness, ego-dystonicity, feared self, level of distress) have on the experience of these phenomena. Due to the exploratory nature of this study, no specific hypotheses were made. The following research questions were used to guide the study development and completion:

- 1. What are the phenomenology of AITs and aggressive scripts?
- 2. What are the similarities and differences in phenomenology between AITs and aggressive scripts?
- 3. What are the emotional and behavioural outcomes associated with AITs and aggressive scripts?

CHAPTER 6 – METHODOLOGY

6.1 Overview

This chapter details the research design and methods used in the two empirical studies. It is acknowledged that each empirical paper presented in the following chapters contains their own methodology relevant to each of the studies. However, as each empirical paper has been prepared for publication, word restrictions apply, and thus methodology covered within each paper is limited. This expanded methodology section provides a more complete account of the methodology to assist the reader to fully appreciate the different methods. There is therefore, some inevitable repetition across chapters regarding the methods used. The purpose of this chapter is to provide more specific detail surrounding decisions made for the research design, sample selection, and ethical considerations. Additionally, the recruitment process, measures used, data collection procedures, and data analyses employed are described.

6.2 Research Design

The two empirical studies utilised a cross-sectional design, with study one using predominately quantitative methods, and study two using a mixed-method design of both quantitative and qualitative methods. By using quantitative methods, study one explored the predictors of aggressive script rehearsal and AITs, inclusive of dysfunctional beliefs, self-related beliefs, obsessive-compulsive symptoms, and a history of aggressive behaviour. Utilising both quantitative and qualitative methods in study two allowed for both symptom and subjective experiences to be explored, with qualitative questioning enabling further examination of aggressive thoughts, and their association with other factors (e.g., emotional reactions, ego-dystonicity).

6.3 Participant Selection

Study one involved the recruitment of individuals from the general population. Given that the experience of intrusive thoughts is considered a common phenomenon, occurring in approximately 92% of individuals at some stage in their life (Julien et al., 2007), study one was interested in examining how the experience of intrusive thoughts relates to aggressive scripts and associated features.

Study two involved the recruitment of a group of individuals diagnosed with OCD, and a group of individuals receiving treatment for violent offending or anger issues through a Community Forensic Mental Health Service. While the intetion was to recruit participants with OCD for study two given the commonality of AITs in this clinical population (Rachman, 1997) this proved too challenging given COVID-19 disruptions. It is suggested that AITs in OCD are not related to aggressive behaviour (Veale et al., 2009), and thus the intention to recruit from this population group was to explore the potential factors that may protect against aggressive behaviour by exploring the subjective experiences associated with AITs. Participants with a history of interpersonal violence or anger issues were recruited for study two given the increased likelihood that aggressive scripts could be reported (Daff et al., 2015). There is a general acceptance that a history of violence, and the presence of violence supportive beliefs are related to the experience of aggressive script rehearsal (Hosie et al., 2021), and thus empirical study two was interested in examining the experiences of aggressive scripts in individuals with relevant backgrounds as well as understanding the subjective experiences associated.

When planning the recruitment processes for study two, we acknowledged that difficulties may arise in the recruitment of participants with OCD who report experiencing AITs, as it is likely that participants with OCD are reluctant to report their AITs for fear of how they may be interpreted by others, and the perceived consequences of this. It was also

acknowledged that certain questions within the interview protocols are considered sensitive topics (e.g. I've had a mental intrusion about injuring or harming someone close to me), and may cause discomfort to some participants. To support participants who become uncomfortable or distressed, several procedures were put in place, including providing participants with support services resources (e.g., Lifeline, Lifeline International, Beyond Blue; refer to Appendix J).

Additionally, given the nature of the data collected, it was acknowledged and made clear to participants at the beginning of the interviews, that the researcher was bound by ethical guidelines pertaining to breaches of confidentiality and mandatory reporting, and that if there was a reasonably belief that risk was imminent to the participant or any other persons, reporting to relevant authorities was required.

6.4 Ethical Procedures

Full approval for study one was granted by the Swinburne University Research Ethics Committee (Ref: 20190386-1937). See appendix A for supporting ethical approval documents. Full approval for study two was granted by Swinburne University Research Ethics Committee (Ref: 20215556-7961) and the Forensicare Operational Research Committee. See appendix B and appendix C for supporting ethical approval documents.

Prior to commencing the online survey for study one, participants were provided with an advertisement of the study, an explanatory statement that detailed information about the study, consent processes, withdrawal procedures, confidentiality, data storage, and dissemination of study results (see appendix D, E, F). Participants consented to participate by answering yes or no within the survey. At the end of the survey participants were provided with a debriefing statement, which reiterated the purpose of the study and provided participants with crisis line numbers if required (see appendix G). Participants were given the option to provide their emails to sign up for the gift voucher prize draw.

Prior to commencing interviews for study two, participants were given the advertisement of the study either by the student researcher, or by their treating clinician (see appendix H and I). Participants were also provided with the explanatory statement that detailed information about the study, what it involved, consent processes, data storage and dissemination procedures (see appendix J). Participants provided consent by signing a statement of informed consent (see appendix K). Before commencing the interview, participants were reminded that should any of these responses raise questions about their safety, or the safety of others, relevant treating clinicians or authorities would be notified. Participants consented on grounds of these terms. No identifying information, including consent forms, were stored with completed interview transcripts or survey responses. All interview documents were individually assigned a unique code and thus were de-identified. Participants were informed that any of their responses to the questions in the interview or the surveys would not be disclosed to their treating clinicians or support staff. Participants were provided an AUD\$30 gift voucher for their participation.

Funding was provided to the student researcher and author of this thesis by Swinburne University through the Research Training Program Stipend.

6.5 Procedure

6.5.1 Recruitment

Study one recruitment occurred between June 2020 and October 2020. Participants were recruited using an undergraduate psychology research experience program from an Australian University. The study advertisements were uploaded via the student research experience program portal, and students chose to participate based on their interest in the study. Students participated in the study in exchange for course credit. Participants were also recruited through advertisements on social media (Gumtree, Whirlpool, Twitter, and Reddit),

and additional participants were obtained via "snowball" methods initiated by those who participated. Only participants recruited via social media streams were offered the opportunity to enter into a draw for one of four AUD\$100 gift vouchers.

Study two recruitment occurred between August 2021 and July 2022. During the second half of the year in 2021, the state of Victoria, where the study was conducted, underwent a protracted COVID-19 lockdown impacting recruitment processes. As participants were being sought from the Systematic Treatment of Obsessive-Compulsive Phenomena (STOP) OCD group program via the Swinburne University of Technology Psychology Clinic (STOP OCD Group program) and the Community Forensic Mental Health (CFMHS; Handling Anger Wisely Program), these programs were postponed until early 2022. During the time between August 2021 and January 2022, the student researcher contacted treating clinicians within CFMHS and clinicians in community private practice, providing them with the advertisement and statement of information of the study to pass on to potential participants. Participants recruited via their treating clinicians involved participants following a link on the advertisement flyer where they were asked to provide their contact details to have the student researcher contact them to discuss the project, consent processes, and to organise an interview.

Amendments to the Swinburne University Research Ethics Committee and Forensicare's Operational Research Committee applications were made to accommodate for online video or telephone interviews to be conducted. Despite these amendments, community treatment clinicians involved in supporting the recruitment of required participants were unable to identify participants who, (a) met the eligibility criteria for the study, or (b) were comfortable in sharing their experiences of AITs or aggressive scripts. It is worth noting that the student researcher was able to liaise with community OCD treatment clinicians, and it was identified that a common barrier for research participation in individuals diagnosed with

OCD and who experience AITs was the overwhelming fear associated with disclosing their aggressive thoughts, and the perceived repercussions associated with this disclosure.

Data collection began in early 2022 when both group programs were scheduled to resume. Recruitment of participants from the STOP OCD Group program involved facilitators providing potential participants with the advertisement flyer and obtaining consent from the participant to have their contact details passed on to the student researcher. The student researcher then contacted each interested participant, introduced the study, and explained the consent process. Despite having several individuals (n= 3) from the STOP OCD group program and community practices interested in participating, these individuals did not experience AITs and were therefore not eligible for the study.

Once group programs commenced in 2022, the student researcher contacted group facilitators of the Handling Anger Wisely Program to organise a time to present the study to group participants. The student researcher attended a scheduled group session and distributed advertisement flyers, introduced the purpose of the study, eligibility criteria, interview process, and reimbursement to the participants. If interested in participating, participants were given the statement of information and consent form and were given the opportunity to ask the student researcher any questions about the project. Participants were given the option to take the statement of information and consent form home to review, or review and sign the consent form at that moment. A day and time to participate was organised with the participant.

Between January and July 2022, four individuals from the CFMHS participated in the study. Data collection remained open until July 2022, until the decision was made by the student researcher and supervisory team to cease data collection.

6.5.2 Data Collection Process

Study one involved the administration of online self-report questionnaires to the general population. The survey battery consisted of a consent statement, questions related to basic demographics such as age, gender, place of residence, as well as questionnaires that measured unpleasant intrusive thoughts including AITs, aggressive script rehearsal, obsessive and feared self beliefs, obsessive-compulsive symptoms, violence supportive beliefs, a history of aggressive behaviour, anger rumination, ego-dystonicity, thought control strategies, and socially desirable responding. It took approximately 30-45 minutes to complete the online survey.

Study two involved the collection of basic demographic questions including age, gender, ethnicity, and behavioural history (e.g., violent offending or OCD diagnosis). A semi-structured was conducted using the Schedule of Imagined Violence (SIV), followed by further questioning depending on answers given to the SIV. The semi-structured interview also involved administration of the Questionnaire of Unpleasant Intrusive Thoughts, which was also followed by further questions depending on answers collected. A battery of self-report questionnaires were also administered measuring anger rumination, violence supportive beliefs, a history of aggressive behaviour, ego-dystonicity, obsessive and feared self beliefs, obsessive-compulsive symptoms, thought control strategies, and socially desirable responding. It took approximately 120 minutes to complete the semi-structured interviews and self-report questionnaires. Throughout the interview participants were reminded that they could take breaks if required, and ask questions as necessary. Only one participant refused to participate in the whole interview once recruited. Once interviews were completed, participants were sent the gift voucher via email, mail, or given to their corresponding psychology clinic for pick up.

The measures used in study one and two are explained below.

6.6 Materials/Measures

6.6.1 Participant Demographics

Participant demographics were obtained to provide a general characterisation of the sample used. Participants were asked about their age, gender, ethnicity, and in study two, additional questions about their background (i.e., violent criminal history, anger problems) were explored.

6.6.2 Socially Desirable Responding

Socially desirable responding is defined as a respondents tendency to answer self-report questionnaires in a way that presents them in a positive light (Paulhus, 2017). This becomes problematic as respondents who are responding in a socially desirable way are likely to be responding in this way across other measures administered concurrently (Paulhus, 2017). Therefore for the current thesis, it was important to consider socially desirable responding in the self-reports of respondents.

The Social Desirability Scale (SDS-17) is a 17 item self-report measure, devised from the Marlowe-Crowne Scale (Lück & Timaeus, 1969), that assess the extent to which participants engage in social desirable responding (for a copy of the SDS, please see Appendix L). Participants respond to items such as "I sometimes litter" on a dichotomous scale of true or false. The SDS has demonstrated adequate internal consistency ($\alpha = 0.72$), and good test-retest reliability over a 4-week period (r = 0.82; Stöber, 2001).

6.6.3 Aggressive Script Rehearsal

Several studies have utilised the Schedule of Imagined Violence (SIV; Daff et al., 2015; Grisso et al., 2000; Hosie et al., 2021) as a measure of aggressive script rehearsal, and associations between aggressive script rehearsal and aggressive behaviour have been identified across several population groups (Grisso et al., 2000). Aggressive script rehearsal

is most commonly measured using the Schedule of Imagined Violence (SIV; Grisso et al., 2000), and the current thesis utilised an adapted version of the original SIV (see Appendix N). The original SIV comprised a set of eight items administered in a semi-structured interview format to measure details related to participants' experience of a daydream or thought about physical violence. Participants who answer positively to the first item (i.e., "Have you ever had daydreams or thoughts about physically hurting or injuring some other person?") are invited to answer an additional seven questions. Each remaining item asks participants to consider different aspects of their thought experience: form of thought (Item 1a), content of thought (Item 1b), frequency (Item 2), chronicity (Item 3), similarity/diversity in type of harm (Item 4), degree of seriousness (Item 5), proximity to target (Item 6), and context (Item 7).

Empirical study one of this thesis utilised question one of the SIV which asks respondents to report how frequently they rehearse their aggressive scripts, "How often do you have thoughts about hurting or injuring other people?". Empirical study two utilised the full modified version of the SIV that explores several facets of aggressive script rehearsal. Only the frequency item of the the SIV was used as the SIV has been shown to reliably associate with different levels of aggression, and several studies have confirmed that a greater frequency of aggressive script rehearsal is associated with greater likelihood of aggressive behaviour (Daff et al., 2015; Grisso et al., 2000; Hosie et al., 2021). A copy of this modified version is available in appendix M.

6.6.4 Aggressive Intrusive Thoughts

The Questionnaire of Unpleasant Intrusive Thoughts (QUIT) is a self-report measure of unpleasant intrusive thoughts. Derived from earlier measures of intrusive thoughts (Clark et al., 2014; Garcia-Soriano et al., 2011; Purdon & Clark, 1993), the QUIT assesses the experience of specific themes of intrusive thoughts. Five intrusive theme domains comprise

the QUIT: unpleasant content, physical appearance, diet and physical exercise, health-related, and relationship-related. For the purposes of the current project, only the unpleasant content domain was used (for a copy of the QUIT, please see Appendix N). Participants are provided with a preamble before questions are presented which defines what an intrusive thought is, their features and characteristics. The following description is read out to participants: "This questionnaire includes a list of somewhat upsetting, unpleasant, or even disturbing thoughts that most people have had at some time. These thoughts SUDDENLY APPEAR IN OUR MINDS against our will, and INTERRUPT what we were doing or thinking at that moment. They can often be DIFFICULT TO CONTROL; that is, it can be difficult to get them out of our minds, stop them, or keep them from appearing, no matter how hard we try. Furthermore, they are UPSETTING, UNPLEASANT, and sometimes even DISTURBING or UNACCEPTABLE. These types of thoughts are called "MENTAL INTRUSIONS" or "SUDDEN, UNINVITED THOUGHTS", and they can appear in our minds in one or more of these ways: 1. As IMAGES, that is, like photographs that suddenly appear in our minds; 2. As an IMPULSE or STRONG AND URGENT NEED to do or say something; 3. As an unpleasant physical SENSATION; 4. Or, simply as THOUGHTS or DOUBTS about something."

The first part of the unpleasant content domain contains 12 intrusive thought examples, where participants rate items such as "While holding a sharp object like a knife, I have had mental intrusions about injuring or harming a person close to me" on the frequency and level of discomfort associated. Frequency is measured on a 7-point Likert scale ranging from 0 (*never*) to 6 (*always*), and discomfort is measured on a 5-point Likert scale ranging from 0 (*does not bother me at all*) to 4 (*is extremely disturbing*).

The second part of the unpleasant content domain asks participants to consider the type of intrusion, from the 12 items rated in part one, that was experienced in the past 3

months, which was the most disturbing/unpleasant or was bothersome due to its frequency. Participants are asked to indicate which intrusion is being referred to, and then are asked a series of questions which measure the form and recency of the intrusion, the context of its occurrence, and the subjective experience associated. Participants' subjective experience are rated using items such as "I try not to think about the intrusion; I try to mentally suppress it", on a 5-point Likert scale ranging from 0 (never/not at all) to 4 (always/frequently). Only the frequency item was used for the QUIT as it allowed for consistency and the most appropriate comparison between measures of aggressive thoughts.

The psychometric properties of the full QUIT have been assessed cross-culturally (Pascual-Vera et al., 2019) and it has demonstrated good to excellent internal consistency across different countries ($\alpha = 0.80 - 0.92$).

6.6.4 Anger Rumination

Several research studies have found an association between anger rumination and anger (Bushman, 2002), as well as anger rumination and aggressive behaviour (Peled & Moretti, 2009). The Anger Rumination Scale (ARS) was used in the current thesis to assess thoughts around anger-provoking situations (Sukhodolsky et al., 2001). The ARS comprises 19 items and four subscales; Angry Afterthoughts, Thoughts of Revenge, Angry Memories, and Understanding of Causes (for a copy of the ARS, please see Appendix O). The current thesis only utilised the Thoughts of Revenge subscale, which measures thoughts about anger and retribution after provoking situations have occurred. This subscale was selected as the researchers were interested in exploring how perseverative thinking with regards to retribution and provocation related to other aggression constructs. Participants rate items such as "I have long living fantasies of revenge after the conflict is over" on a 4-point Likert scale ranging from 1 (almost never) to 4 (almost always). The original ARS has demonstrated

adequate internal consistency ($\alpha s = 0.72 - 0.86$) and good test-retest reliability r = 0.77 for a one month period (Sukhodolsky et al., 2001).

6.6.5 History of Aggressive Behaviour

History of aggressive behaviour was measured using the Life History of Aggression (LHA) scale revised by Coccaro et al. (1997). Earlier versions of the LHA such as those used by Brown et al. (1979) were initially developed as a measure of lifetime history of aggression to establish associations with several biological factors in military personnel. Using items from the Brown et al. (1979) assessment, Coccaro et al. (1997) modified the LHA developing the measure into a semi-structured interview style format. The LHA has three subscales: the aggression subscale, the consequences and antisocial behaviour subscale, and the self-directed aggression subscale. The aggression subscale contains five items that measure the number of occurrences of aggressive behaviours since the age of 13. The items contain details of overt aggressive behaviour including verbal aggression, temper tantrums, destruction of property, fighting, and physical assaults. The current thesis has only utilised the aggression subscale in a self-report format (for a copy of the LHA, please see Appendix P). Participants rate items such as "Temper Tantrums" on a 5-point Likert scale ranging from 0 (no occurrences) to 5 (more events than can be counted).

Psychometric properties of the self-administered LHA are not available. However, the LHA has been found to correlate highly with the interview version (Dunne et al., 2018). Additionally, the aggression subscale has demonstrated good internal consistency ($\alpha = 0.87$), and good test-retest reliability ($\alpha = 0.80$; Coccaro et al., 1997).

6.6.6 Violence Supportive Beliefs

Violence supportive beliefs were measured using the Measures of Criminal Attitudes and Associates (Mills et al., 2002). The full measure of the MCAA consists of a two-part

self-report questionnaire. Part A quantifies the individuals association with individuals who have criminal involvement. Part B includes 46-items, across four subscales that measure differential criminal attitudes. The four subscales consist of Violence, Entitlement, Antisocial Intent, and Associates. The current thesis only utilised 13 items pertaining to the Violence subscale as the research was interested in exploring respondents attitudes supportive of violence only. Items such as "There is nothing wrong with beating up someone who asks for it" are rated on a dichotomous scale of agree/disagree (for a copy of the MCAA, please see Appendix Q). The violence subscale has demonstrated good internal consistency ($\alpha = 0.80$) and test-retest reliability ($\alpha = .73$)

6.6.7 Ego-dystonicity

Ego-dystonicity has been consistently found as a factor that differentiates obsessional intrusive thoughts from normal thought phenomena (Purdon et al., 2007). The Ego Dystonicity Questionnaire (EDQ) was originally created by Purdon et al. (2007). In an adaptation study with a Spanish sample, Belloch et al. (2012) reduced the measure to 27 items (for a copy of the EDQ-Revised, please see Appendix R). The self-report measure assesses the extent to which one believes the content of their thoughts is inconsistent with their self-beliefs, values, and moral attitude. For the purposes of the current thesis, the EDQ-R has been modified to ask participants to focus on their most upsetting Aggressive Intrusive Thought and/or their aggressive thought of wanting to harm or injure another person, whilst providing their ratings. Participants rate items such as "Thought is immoral" on a 7-point Likert scale ranging from 1 (strongly agree) to 7 (strongly disagree). The EDQ-R has demonstrated good internal consistency (α s = 0.94).

6.6.8 Thought Control

Disparate research on AITs and aggressive scripts have identified that a range of thought control strategies are often employed to help participants manage the distress associated with the thought, as well as the thought experience and frequency (Belloch et al., 2004; Nagtegaal et al., 2006). The original Thought Control Questionnaire (TCQ) comprises 30 items (Wells & Davies, 1994) which measure the use of different control strategies (for a copy of the TCQ, please see Appendix S). The five factors measured by the TCQ include: distraction, social control, worry, punishment, and reappraisal. A recent psychometric study conducted by Luciano et al. (2006) confirmed a five-factor model with 16 items. This 16-item version was used in the current thesis. Participants rate items such as "I punish myself for thinking the thought" on a 4-point Likert scale ranging from 1 (never) to 4 (almost always). The 16-item version of the TCQ has demonstrated adequate internal consistency in the current study ($\alpha = 0.68$)

6.6.9 Obsessive Beliefs

Maladaptive beliefs have been extensively studied within OCD research (Obsessive Compulsive Cognitions Working Group, 2003). The Obsessive Beliefs Questionnaire (OBQ-20) is a short form version of the original OBQ (OCCWG, 2005), and was developed by Moulding et al. (2011). A 20-item self report measure, the OBQ assesses four obsessive beliefs identified through factor analyses: (1) Threat, (2) Responsibility, (3) Importance of Thoughts, and (4) Perfectionism (for a copy of the OBQ-20, please see Appendix T). Participants rate items such as "If I'm not absolutely sure of something, I'm bound to make a mistake" on a 7-point Likert scale ranging from 1 (Disagree very much) to 7 (Agree very much). The OBQ-20 has demonstrated excellent internal consistency across all subscales $\alpha = 0.80 - 0.82$ (Moulding et al., 2011).

6.6.10 Self Ambivalence

Specific beliefs concerning uncertainty and preoccupations with one's self-worth and moral compass has received increased attention in OCD research (Bhar & Kyrios, 2007). The Self Ambivalence Measure (SAM) consists of 19-items and measures the degree to which individuals hold beliefs concerning uncertainty towards their self-concept, and dichotomous self-evaluations about the self (for a copy of the SAM, please see Appendix U). Participants rate items such as "I have mixed feelings about my self-worth" on a 5-point Likert scale ranging from 0 (not at all) to 4 (agree totally). The SAM has demonstrated excellent internal consistency ($\alpha = 0.93$; Aardema et al., 2013)

6.6.11 Feared Self

Self-themes, included feared self beliefs, have been found to relate with obsessive-compulsive symptoms (Aardema et al., 2013), specifically with repugnant obsessions such as AITs (Aardema et al., 2019). The FSQ is an 8-item self report measure that assess beliefs pertaining to covert aspects of one's personality (for a copy of the FSQ, please see Appendix V). Participants rate items such as "I often question my own character" on a 6-point Likert scale ranging from 1 (strongly disagree) to 6 (strongly disagree). The FSQ-8 has demonstrated good internal consistency ($\alpha = 0.88$), and good convergent validity with other measures of self-related beliefs and constructs (i.e., Self-Ambivalence Measure r = 0.68; Inferential confusion Questionnaire r = 0.72; Aardema et al., 2013).

6.6.12 Obsessive Compulsive Symptoms

Obsessive compulsive (OC) symptoms are common characteristics of OCD (Leonard & Riemann, 2012), but have also been reported in non-clinical samples (Radomsky et al., 2014). The Obsessive Compulsive Inventory – Revised (OCI-R; Foa et al., 2002; for a copy see Appendix W) is an 18-item self-report measure that assesses a range of obsessive

compulsive symptoms, such as "I find it difficult to control my own thoughts". Participants rate items on a 4-point Likert scale ranging from 0 (not at all) to 4 (extremely). The OCI-R has demonstrated excellent internal consistency ($\alpha = 0.90$; Radomsky et al., 2014).

6.6.13 Depression, Anxiety and Stress

Research has confirmed an association between depression, anxiety and stress symptoms with OCD, particularly with the presence of aggressive obsessions (Ching et al., 2017). Given this association, the Depression, Anxiety Stress Scale (DASS-21; Lovibond & Lovibond, 1995) was used to control for these symptoms (for a copy of the DASS-21, please see Appendix X). The DASS-21 is a 21-item self-report measure comprising of three subscales: Depression, Anxiety, and Stress. The DASS-21 measures a respondents emotional well-being, and participants rate items such as "I felt that life was meaningless" within a reference period of the past seven days. Participants rate items on a 4-point Likert scale ranging from 0 (did not apply to be at all) to 3 (applied to me very much or most of the time). The DASS-21 has demonstrated excellent internal consistency across all three subscales (Depression: $\alpha = 0.91$; Anxiety: $\alpha = 0.80$; Stress: $\alpha = 0.84$; Sinclair et al., 2012).

6.6.14 Construct Phenomenology

A semi-structured interview was developed to distinguish AITs and aggressive scripts for empirical study two. The interview schedule was derived from existing measures of AITs and aggressive scripts, as well the differential features pertinent to intrusive thoughts and aggressive script phenomenology. Items were developed by the research team and have been based on previous self-report and interview schedules from OCD and aggressive script research areas (Grisso et al., 2000; Pascual-Vera et al., 2019). The interview schedule is available in Appendix Y.

By using a semi-structured interview format, it allowed for the researchers to employ follow-up style questions to participant responses, particularly responses to the self-report questionnaires. For example, participants who reported rehearsing aggressive scripts were asked descriptive questions to explore their subjective experiences; such whether the aggressive scripts was experienced as intrusive, spontaneous, and distressing.

A mixed methods approach was used in empirical study two, where both quantitative and qualitative data were collected during interviews with participants. Both self-report measures and a semi-structured interview were used to contextualise the subjective experiences of script rehearsal.

6.7 Participant Sample

The original sample for study one consisted of 460 participants. After validity checks were conducted, the final sample included 412 English-speaking non-clinical subjects aged between 18 and 69 (M = 31.96; SD = 11.02; 73% females) who did not fail attentional control task questions and completed all questionnaires. Attentional control task questions were distributed throughout the survery, and asked participants to answer a question a specific way (e.g., select option A for this question). Attentional control task questions were used to determine if participants were answering questions in an appropriate manner, and not randomly selecting answers. Participants resided in Australia (n = 401), China (n = 1), Indonesia (n = 1), Japan (n = 1), Lebanon (n = 1), Russia (n = 1), Sri Lanka (n = 1), Switzerland (n = 1), and USA (n = 2). Two participants did not disclose where they were residing.

For study two, the total sample comprised five male participants from the Community Forensic Mental Health Service (CFMHS). Participants were aged between 23 and 49 (M = 36.50; SD = 10.76). One participant from CFMHS was excluded from the study as they refused to complete the full interview and thus complete data was not collected.

6.8 Data Preparation and Analysis Methods

The following section explores the data entry and analyses processes used for both empirical studies.

6.8.1 Data Preparation and Analyses for Empirical Study One: Exploring
Predictors of Aggressive Intrusive Thoughts and Aggressive Scripts: Similarities
and Differences in Phenomenology

6.8.1.1 Data Inspection

Data collected via Qualtrics was uploaded into IMB SPSS (Statistical Package for Social Sciences) version 27 for statistical analyses. Preliminary data inspection revealed that five items were missing from the survey across three variables (one item from OBQ-20; two items from MCAA, and two items from TCQ) due to administration error (n = 223). The survey was amended to include all missing items, and the amended survey was used in all subsequent data collection. Once data collection was completed and an additional 237 participant responses were recorded (N = 460), an Expectation-Maximisation (EM) algorithm was applied to the data set to impute the missing item scores from the initial survey. The utilisation of an EM algorithm for the missing items was justified as Little's MCAR test indicated that the data was Missing Completely at Random (MCAR): [OBQ-20; χ^2 (68, N = 460) = 57.98, p = .80; MCAA; χ^2 (49, N = 460) = 532.12, p = .97; and TCQ; χ^2 (68, N = 460) = 86.97, p = .06]. A missing values analysis was performed on the entire data set and this revealed that several scales also contained missed data. Little's MCAR tests were performed on these scales which indicated that data was MCAR, and thus missing data was imputed using the EM algorithm.

Further data inspection revealed that several scales were significantly skewed, which is a common trend in non-clinical samples. Transformations on skewed scales were

performed to normalise the distributions, however, some of the scales remained skewed. The assumptions for regression analyses were met. Further details of the analytical procedure in empirical study one is detailed in the following chapter.

6.8.1.2 Main Analyses

Correlational analyses, using Pearson's correlation coefficient, were conducted to analyse the association between variables. Four, two step regression analyses were conducted on the predictors of a history of aggressive behaviour, obsessive-compulsive symptoms, aggressive script rehearsal, and AITs. The regression analyses also assessed the unique contribution of anger rumination, violence supportive beliefs, ego-dystonicity, thought control, obsessive beliefs, and self-ambivalence have on the aforementioned predictors. The regression analyses allowed for depression, anxiety, and stress to be controlled in these relationships.

Examination of the regression analyses identified that certain relationships indicated a suppression effect had occurred, as the direction of the relationships were contrary to prior research (e.g., association between OBQ and QUIT was negative). Repeated regression analyses were conducted, excluding each variable in the model to determine which factors were causing the suppression effect.

6.8.2 Analyses for Empirical Study Two: Exploring the Experiences of
Aggressive Intrusive Thoughts and Aggressive Scripts in an OCD and Forensic
Sample - A Case Study

Qualitative analysis for study two involved the use of inductive thematic analysis on the interview transcripts, across both sample groups. As highlighted by Braun and Clarke (2006), inductive or 'bottom up' thematic analysis, involves identifying themes that are

derived from the data, as opposed to a theoretical or analytic interest guiding the identification of themes (i.e., deductive or 'top down' thematic analysis).

Interview audio recordings were transcribed user Otter.ai software. Transcripts were reviewed approximately four times by the student researcher, and errors in transcriptions corrected accordingly. Transcripts were then uploaded to NVivo qualitative data analysis software package. Raw psychometric data collected during interviews were entered into an Excel spreadsheet by the student researcher. The data set were then uploaded into SPSS for analysis.

Before codes were established, the student researcher reviewed the transcripts and developed a reflective journal noting any preliminary themes, opinions, biases, or processes identified during the interview stage. This allowed the student researcher to transparently note their experiences of the interviews and interactions with participants that may influence coding and theme development. Transcripts were coded by the student researcher and last author (MN), and the supervisory team was consulted frequently to discuss coding options and processes. Once initial codes were generated on all transcripts, themes were established. The student researcher consulted the supervisory team to review and define generated themes. Themes were then refined and named which were included in the final empirical study paper.

6.8.2.1 Reflective Statement

To ensure transparency is maintained throughout the research process, it was deemed necessary to develop a reflective journal throughout the data collection and analysis phase.

This would allow the student researcher to explore their role, any biases they encountered which in turn may have influenced the data collection processes and overall findings of the study. A reflective statement is provided below by the student researcher.

My interest in this study developed out of clinical curiosity in being able to differentiate, with confidence, aggressive scripts, which are commonly reported in forensic samples, from AITs, a common symptom of OCD. My understanding of the two phenomena were that aggressive scripts are anecdotally assumed to be ego-syntonic, and are consistently associated with aggressive behaviour, whereas AITs are widely accepted as thoughts that are highly ego-dystonic and have no relationship with overt acts of aggression or violence. At the time of the study development, I was completing a doctorate placement in a forensic service which became a forensic clinical setting that proved useful as a recruitment strategy for the study. By being a member of staff, I became familiar with the processes of the service, the clientele, and risk assessment considerations. Due to COVID-19 restrictions in Victoria, whilst completing my placement with the service, I was unable to begin data collection due to logistic and group program rescheduling issues. This avoided my engagement in dual roles as both a student clinician and PhD researcher, which was beneficial. This may then have influenced participants to be more forthcoming with their responses and experience, given I was not directly associated with the service as a clinician during the data collection phase. Despite this, it was imperative the participants were aware of the boundaries to the research interviews, including the limits to confidentiality and duty to warn processes - these boundaries were made apparent at the outset of each interview. Participants were also informed that any content discussed during the interview would not be described to their treating clinicians, and only included in the analysis and report of the study.

The level of detail provided by participants varied across interviews. Some participants were willing to share detailed experiences of their aggressive thinking including the antecedents, consequences, and details of the content of these thoughts. However, occasionally some participants were more reserved with their responses, which would require more direct prompting and questioning from myself. I found myself concerned with not

wanting the participants to feel they had to share things there were uncomfortable with, but also, I wanted to be able to fully understand their experiences for the purposes of the data collection. Reflecting on my interview style, my questions for participants may have been too complex or require time to think deeply about before responding, which was evidenced by some participants taking their time to respond. This may have impacted on the data collection process as participants may have been given more time to answer questions, which left less time to explore other details of their experiences due to time constraints of the interview.

CHAPTER 7 – EMPIRICAL RESEARCH PAPER ONE

7.1 Preamble to Empirical Research Paper One

This chapter presents the first empirical research study of this thesis. Empirical study one addresses the second thesis aim, to elucidate the similarities and differences in the phenomenology of AITs and aggressive script rehearsal. Extending from questions raised from the critical review in chapter four, which highlighted that although there are similarities in features between AITs and aggressive scripts, some features require further investigation (e.g., level of intrusiveness, ego-dystonicity, associated beliefs) and issues with measurement need to be considered. This paper is one of the first to measure AITs and aggressive scripts concurrently, and although a non-clinical sample has been used, inferences made from the study's findings have implications for the measurement and assessment of AITs and aggressive scripts.

This research paper was intended to provide an empirical investigation of commonly used OCD measures, and aggression-related measures with the aim to identify differentiating features between AITs and aggressive scripts. This paper provides a basis for exploring the second and third thesis aims, which are explored in empirical research paper two in chapter eight.

Empirical study one, Exploring Predictors of Aggressive Intrusive Thoughts and Aggressive Scripts: Similarities and Differences in Phenomenology has been published in Aggressive Behavior. Aggressive Behavior is a peer-reviewed journal that publishes empirical papers that focus on the factors underlying or influencing aggressive behaviour, with a current impact factor of 3.047 (Clarivate, 2021).

Fernandez, S. J., Daffern, M., Moulding, R. & Nedeljkovic, M. (2022). Exploring Predictors of Aggressive Intrusive Thoughts and Aggressive Scripts: Similarities and Differences in Phenomenology. *Aggressive Behavior*.1-3. http://doi.org/10.1002/ab.22061

7.2 Author Indication Form for Empirical Research Paper One



Swinburne Research

Authorship Indication Form

For PhD by Publication candidates

NOTE

This Authorship Indication form is a statement detailing the percentage of the contribution of each author in each published 'paper'. This form must be signed by each co-author and the Principal Coordinating Supervisor. This form must be added to the publication of your final thesis as an appendix. Please fill out a separate form for each published paper to be included in your thesis.

DECLARATION We hereby declare our contribution to the publication	on of the 'naner' entitled:
	d Aggressive Scripts: Similarities and Differences in Phenomenology
Expansing predictors of riggressive intersect moderns and	a riggressive compatibilities and biliterates and mineral sign
First Author	A)
Name_Stephanie Jane Fernandez	Signature:
Percentage of contribution: 85 %	Date: 20/02 /2023
Brief description of contribution to the 'paper' and y	our central responsibilities/role on project:
Reviewing literature, research design, data collect manuscript	ction, statistical analysis, prepared
Second Author	
Name: Michael Daffern	Signature:
_ Percentage of contribution: _ 5_%	Date: <u>21/02/_23</u>
Brief description of your contribution to the 'paper': Supervision on research design, critically reviewe	
Third Author	
Name: Richard Moulding	Signature:
Percentage of contribution: _5%	Date: 20 / 02 / 2023
Brief description of your contribution to the 'paper':	:
Supervision on research design, assistance with	statistical analysis, critically reviewed and edited manuscript
Fourth Author	
Name: Maja Nedeljkovic	Signature: Ollicy
Percentage of contribution: _5_%	Date: 20 / 02/ 2023
Brief description of your contribution to the 'paper': Primary supervision at all stages, critically revie	wed and edited manuscript

Principal Coordinating Supervisor:	
Name:Maja Nedeljkovic	Signature: Alley
Date: 20 / 02 / 2023	

Exploring Predictors of Aggressive Intrusive Thoughts and Aggressive Scripts: Similarities and Differences in Phenomenology

Stephanie Jane Fernandez^{a,b}, Michael Daffern^{a,b}, Richard Moulding^c, and Maja Nedeljkovic^a

^aDepartment of Psychological Sciences, Swinburne University of Technology, John Street,

Hawthorn, Melbourne, Australia, ^bCentre for Forensic Behavioural Science, Swinburne

University of Technology, 582 Heidelberg, Alphington, Melbourne, Australia, ^cThe

Cairnmillar Institute, Melbourne, Australia

Corresponding author: Stephanie Fernandez, Centre for Forensic Behavioural Science, 582

Heidelberg, Alphington, Victoria, Australia, 3078; Email: sfernandez@swin.edu.au

Contact number: +61 0421 325 795

Acknowledgements

Funding: The first author received funding support from the Australian Government in the form of an Australian Government Research Training Program Scholarship

Declarations of interest: none

Declaration: This manuscript has not been published elsewhere and has not been submitted simultaneously for publication elsewhere.

Data availability: The data that support the findings of this study are available from the corresponding author upon reasonable request.

Abstract

Experiencing a thought about harming or injuring another person is commonly reported by the general population. Aggressive intrusive thoughts (AITs) and aggressive scripts are two constructs commonly used to define the experience of thinking about harming another person. However, they are generally investigated separately and with two significantly different population groups; respectively, individuals with Obsessive Compulsive Disorder and people with a history of violent behaviour. AITs and aggressive scripts are assumed to have very different implications for violence risk assessment, but conceptual overlap and an absence of empirical research renders this assumption premature. Using a battery of selfreport measures, this study aimed to investigate the differential predictors of AITs and aggressive script rehearsal in a non-clinical sample. Additionally, using regression analyses, the predictors of self-reported aggressive behaviour were explored in a sample of 412 adults (73% females; $M_{\text{age}} = 31.96$ years, SD = 11.02). Violence supportive beliefs and frequency of anger rumination predicted the frequency of aggressive script rehearsal, and aggressive script rehearsal, anger rumination, and violence supportive beliefs predicted a history of aggressive behaviour. In contrast, obsessive beliefs were predictive of AITs, and only AITs were related to ego-dystonicity. Both AITs and aggressive script rehearsal were related to the use of thought control strategies. These findings support the contributions that maladaptive beliefs have in the experience of aggressive scripts and AITs. Beliefs about violence, a history of aggressive behaviour, and ego-dystonicity appear to differentiate aggressive scripts from AITs.

Keywords: Aggressive intrusive thoughts; aggressive scripts, aggressive behaviour, obsessive beliefs, ego-dystonicity

Exploring Predictors of Aggressive Intrusive Thoughts and Aggressive Scripts: Similarities and Differences in Phenomenology

The outcomes associated with thoughts about inflicting harm or injury to others differ significantly depending on the population group that experience these thoughts. In some instances, aggressive thoughts are associated with aggressive acts and violence (Gilbert & Daffern, 2017; Grisso et al., 2000), and in other cases, individuals will go to significant lengths to prevent harm occurring to others (Pascual-Vera et al., 2019; Rowa & Purdon, 2003). In particular, in people diagnosed with Obsessive Compulsive Disorder (OCD), aggressive intrusive thoughts (AITs) are purportedly *not* associated with subsequent acts of aggression (Veale et al., 2009), whereas in offender populations thoughts with similar content, typically referred to as aggressive scripts, *are* associated with aggressive behaviour⁵ (Daff et al., 2015). Historically, these phenomena have been investigated separately and within these two diverse population groups, and the associated features of these two constructs have not been compared, conceptually or empirically. Such an examination is overdue for improving risk assessment and intervention by clinicians.

AITs are a common feature of OCD, with approximately 58% experiencing aggressive obsessions as one of their main symptoms (N = 485; Pinto et al., 2008b); however, the unwanted, distressing, and ego-dystonic (i.e., contradict an individuals' sense of self) features of these thoughts are said to protect against acts of violence (Veale et al., 2009). Rather, AITs induce significant fear and apprehension in those who experience them, influencing compulsive behaviours that reduce one's distress and avert the perceived

-

⁵ The authors acknowledge that AITs are symptoms that appear in other psychiatric conditions such as anxiety disorders and psychotics disorders (see: Abramowitz et al., 2003; Grisso et al., 2000). However, for the scope of this study, only AITs relevant to OCD have been included. The rationale for not including other psychiatric conditions alongside OCD, particularly psychotic disorders, was due to differences in the appraisal process involved during aggressive thought experiences (see: Clark, 2005; Link et al., 1997).

consequences (e.g., ensuring loved ones are safe and not at risk of harm; Rachman, 1997; Veale et al., 2009). Contrastingly, aggressive scripts are defined as thoughts or daydreams about physically harming or injuring another person and they are used to guide behaviour and to regulate emotions (Hosie et al., 2021). When studied in violent offender samples, aggressive script rehearsal have been shown to relate to aggressive behaviour; specifically, the more one rehearses their aggressive scripts in mind, the more likely they are to act aggressively (Gilbert & Daffern, 2017).

Cognitive Models of Aggressive Intrusive Thoughts and Aggressive Scripts

According to the cognitive behavioural model of OCD, AITs are interpreted through beliefs that cause these thoughts to be viewed as abhorrent, dangerous, or threatening (Moulding et al., 2011; Rachman, 1997; Radomsky et al., 2014). These beliefs include thought action fusion—that a thought about harming another person is equivalent to the imagined action (e.g., 'thinking about hurting my loved ones is the same as actually hurting them'; Shafran & Rachman, 2004), and feared self beliefs, where the individual believes they possess bad, dangerous, or immoral characteristics as a result of experiencing unwanted thoughts (e.g., 'I must be a dangerous person for thinking about harming another person'; Aardema & O'Connor, 2007; Jaeger et al., 2021; Moulding et al., 2011; Shafran & Rachman, 2004). Both Veale et al. (2009) and Fairbrother et al. (2022) suggest that there should be no concern regarding whether a person with OCD will carry out their aggressive intrusions, as they are highly ego-dystonic to the individual and are associated with significant distress and trepidation.

Research exploring aggressive script rehearsal is based upon social-cognitive developmental models, like the General Aggression Model (GAM) and Script Theory, which suggest that once aggressive scripts are created, either by observing aggression or acting aggressively, they are maintained through positive beliefs about aggression (Bushman &

Anderson, 2002; Huesmann, 1998). These beliefs include attitudes that endorse aggressive behaviour (e.g., 'Sometimes you have to fight to keep your self-respect'; Mills et al., 2002, p. 249), and these beliefs have been found to associate with aggressive scripts rehearsal and aggressive behaviour, and are common in some offender populations (Gilbert & Daffern, 2017; Huesmann, 1988; Kelty et al., 2011). Further, alternative cognitive models, such as the Multiple Systems Model (Denson (2013), propose that understanding how and why individuals engage in 'angry rumination' (a related but not synonymous construct; Hosie, Simpson, et al., 2022) may identify the precipitants of aggressive behaviour (Denson, 2013). As emphasised by Hosie, Simpson, et al. (2022), anger rumination is concerned with perseverative thinking over experiences of anger, which may also include ruminations regarding past provocations. However, Hosie, Simpson, et al. (2022) argue that anger rumination that includes thoughts of retaliation (i.e., one preparedness and plans for revenge) is synonymous with some aggressive scripts (although some other aggressive scripts are rehearsed outside of the context of angry rumination, for instance, in a more pleasurable planful state).

The Appraisal Process of AITs and Aggressive Scripts

Aggressive Intrusive Thoughts

Current understandings of AITs in OCD highlight how beliefs underly the appraisal of these intrusions (Rowa & Purdon, 2003). For example, Rowa and Purdon (2003) found that AITs are associated with beliefs of responsibility and the need to control thoughts (N = 64). When explained in the context of experiencing AITs, these beliefs: (1) influence the preoccupation with one's responsibility over the content of the thought (e.g., 'I must ensure my loved ones are safe from harm'), and (2) influence one to believe they should control their thoughts (e.g., 'I must not have thoughts with abhorrent contents and should control my

thoughts'). These forms of appraisal have been found to promote neutralising or thought control behaviours (e.g., distraction strategies; or checking) that are dysfunctional, and that in turn maintain the experience of Obsessive Compulsive (OC) symptoms like AITs (Amir et al., 1997; Jacoby et al., 2015).

Within OCD, the content of the intrusive thought is experienced as ego-dystonic to the individual, and the initial appraisal of the thought includes a sense of disbelief about thinking unpleasant or unacceptable content (Lee & Kwon, 2003; Purdon et al., 2007).

Purdon et al. (2007) found that in an OCD sample, experiencing thoughts classed as repugnant was related to these thoughts being experienced as ego-dystonic and inconsistent with one's morals. Purdon et al. (2007) also identified that intrusive thoughts initially appraised as ego-dystonic can, over time, become accommodated into one's self concept and be re-appraised as ego-syntonic. Similarly, Bhar and Kyrios (2007) identified that individuals who demonstrate ambivalence about their sense of self, morality, or lovability (e.g., 'I question whether I am a moral person'; Bhar & Kyrios, 2007, p. 1855) are likely to experience obsessional thoughts and behaviours. Beliefs concerning self-perceptions, including the feared self, have been proposed to influence the appraisal and reoccurrence of AITs. It is highlighted that individuals with feared self beliefs are likely to perceive themselves as "immoral", "dangerous" or "insane" for experiencing thoughts with unacceptable contents (Aardema et al., 2017; Ferrier & Brewin, 2005).

Aggressive Scripts

Beliefs related to aggressive script rehearsal are centred on the acceptability of the imagined aggressive behaviours (Gilbert & Daffern, 2017). In a non-clinical sample of students and community participants, Kelty et al. (2011) found that individuals who endorse violent and aggressive beliefs were likely to rehearse aggressive scripts and engage in aggressive behaviour. The maintenance of aggressive scripts is also moderated by one's past

experiences with aggression and violence, suggesting that a person's life history with aggressive behaviours (either through observation or direct experiences) is highly predictive of a person's aggressive tendencies (Gilbert & Daffern, 2017).

Fewer studies have focused on the role an individual's subjective experience has in aggressive script rehearsal and their appraisal. Hosie et al. (2021) investigated the emotional sequalae associated with aggressive script rehearsal in a sample of incarcerated offenders (*N* = 131). The most common emotion associated with aggressive script rehearsal was anger, followed by hate, fear, sadness, confusion, and annoyance. It was also identified that offenders with a greater history of aggressive behaviour were more likely to experience feelings of excitement when rehearsing aggressive scripts, when compared to offenders with less significant histories of aggression. These findings highlight variability in the emotional reactions associated with aggressive script rehearsal, and also the relationship between aggressive script rehearsal and aggressive behaviour.

Neutralisation and Thought Control Strategies

There is limited research that has examined the thought control methods employed to manage AITs or aggressive scripts, specifically. Studies of OCD have found that a range of control strategies can be employed to manage unacceptable intrusive thoughts, such as AITs, including self-punishment, avoidance, and seeking reassurance (Belloch et al., 2004; Jacoby et al., 2015; Lee & Kwon, 2003). However, the effectiveness of these control strategies has been questioned, as research suggests that certain methods (e.g., self-punishment) can increase the severity and frequency of intrusive thoughts (Jacoby et al., 2015). With regards to aggressive scripts, Nagtegaal et al. (2006) found that in an undergraduate student sample (N = 72), in which 60% reported rehearsal of aggressive scripts, distraction and reappraisal techniques were the most common thought control method utilised. Nagtegaal et al. (2006) also found that self-punishment techniques were associated with hostility and aggression.

These findings across intrusive thought and aggressive script research highlight that different thought control strategies may be utilised to manage different thought experiences. However, the research also indicates that certain methods may have negative implications on the thought experience, and one's affect.

Current Study

To our knowledge, no study has concurrently examined the relationships between cognitive predictors of AITs and aggressive script rehearsal, nor have features of these phenomena been empirically investigated in the same study. As such, in the present study, cognitive predictors related to OCD symptoms, and beliefs related to violent attitudes and aggressive behaviour, were investigated. The contributions of the feared self, ego-dystonicity, and thought control strategies were also examined. It was hypothesised that ego-dystonicity, self-ambivalence, and the fear of self would predict AITs. The association between thought control strategies with AITs and aggressive scripts was explored. It was hypothesised that violent attitudes, anger rumination, and a history of aggressive behaviour would predict aggressive script rehearsal. It was hypothesised that general OCD beliefs and thought control strategies would predict the experience of OC symptoms. It was also hypothesised that anger rumination and violent attitudes would predict past aggressive behaviour. Given the relationship that exists between OCD and depressive symptoms, and the influence that depression and anxiety can have on aggressive thoughts (Ching et al., 2017), depression, anxiety, and stress were controlled for the in the present study.

Method

Participants

The original sample comprised 460 participants, and after validity checks were conducted, the final sample included 412 English-speaking non-clinical subjects aged

between 18 and 69 ($M_{age} = 31.96$; SD = 11.02; 73% females) who did not fail attentional control questions and completed all questionnaires. Majority of participants resided in Australia (n = 401), with the remaining in other countries (n = 9). Two participants did not disclose where they were residing. Review of clinical measures (i.e., Depression, Anxiety Stress Scale – 21; Obsessive Compulsive Inventory-Revised) indicated that approximately 31% of participants experienced moderate to severe levels of stress symptoms. Depression, anxiety, and OC symptoms were within normal range.

Participants were recruited using an undergraduate psychology research experience program from an Australian University. Students participated in the study in exchange for course credit. Participants were also recruited through advertisements on social media (Gumtree, Whirlpool, Twitter, and Reddit), and additional participants were obtained via "snowball" methods initiated by those who participated. Participants recruited via social media streams were offered the opportunity to enter into a draw for one of four AUD\$100 gift vouchers.

Measures

Anger Rumination Scale: Thoughts of Revenge Subscale (ARS; Sukhodolsky et al., 2001). The present study only utilised the Thoughts of Revenge subscale of the ARS, which measures thoughts about anger and retribution after provoking situations. Participants rate items such as "I have long living fantasies of revenge after the conflict is over" on a 4-point Likert scale ranging from 1 (almost never) to 4 (almost always). The original ARS has demonstrated adequate internal consistency (α s = 0.72 – 0.86) and good test-retest reliability (r = 0.77) for a one month period (Sukhodolsky et al., 2001). In the current study, the Thoughts of Revenge subscale demonstrated good internal consistency (α = 0.75).

Depression Anxiety Stress Scales Short Form (DASS-21; Lovibond & Lovibond, 1995). The DASS is a 21-item self-report measure that assess emotional states of depression,

anxiety, and stress symptoms, and comprises three subscales; Depression, Anxiety, and Stress. Participants rate items such as "I felt that life was meaningless" with reference to the past week, on a 4-point Likert scale ranging from 0 (did not apply to me at all) to 3 (applied to me very much or most of the time). In the current study, the DASS-21 demonstrated excellent internal consistency ($\alpha = 0.94$), and good internal consistency across the three subscales ($\alpha = 0.86 - 0.92$).

Ego-Dystonicity Questionnaire- Reduced Version (EDQ-R; Belloch et al., 2012). The EDQ is a 27 item self-report measure that assesses the extent to which one believes the content of their thoughts is inconsistent with their self-beliefs, values, and moral attitude. For the present study, the EDQ was modified to ask participants to focus on their most upsetting 'aggressive intrusive thought' whilst providing their ratings. Participants rate items such as that the "Thought is immoral" on a 7-point Likert scale ranging from 1 (strongly agree) to 7 (strongly disagree). The EDQ-R demonstrated good internal consistency in the present study $(\alpha s = 0.94)$.

Fear of Self Questionnaire (FSQ; Aardema et al., 2013). The FSQ is an eight item self-report measure that assess beliefs pertaining to covert aspects of one's personality. Participants rate items such as "I often question my own character" on a 6-point Likert scale ranging from 1 (strongly agree) to 6 (strongly disagree). In the current study the FSQ-8 demonstrated good internal consistency ($\alpha = 0.88$).

Life History of Aggression (LHA; Coccaro et al., 1997). The LHA, as revised by Coccaro et al. (1997), is a self-report measure that assesses the number of occurrences of aggressive behaviours since the age of 13. The present study only utilised the Aggression subscale which measures overt experiences of aggressive behaviour. Participants rate items such as "Temper Tantrums" on a 5-point Likert scale ranging from 0 (no occurrences) to 5

(more events than can be counted). In the current study, the Aggression subscale demonstrated good internal consistency ($\alpha = 0.79$).

Measures of Criminal Attitudes and Associations (MCAA; Mills et al., 2002). The complete MCAA consists of a two-part self-report questionnaire, comprising Violence, Entitlement, Antisocial Intent, and Associates subscale. For the present study only 13 items pertaining to the Violence subscale were used, where items such as "There is nothing wrong with beating up someone who asks for it" are rated on a dichotomous scale of agree/disagree. In the current study, the subscale of Violence demonstrated good internal consistency ($\alpha = 0.81$).

Obsessive Beliefs Questionnaire (OBQ-20; Moulding et al., 2011). The OBQ-20 is a short form of the Obsessive Beliefs Questionnaire (OCCWG, 2005). A 20-item self-report measure, the OBQ assesses four obsessive beliefs identified through factor analyses: (1) Threat, (2) Responsibility, (3) Importance of Thoughts, and (4) Perfectionism. Participants rate items such as "I should be upset if I make a mistake" on a 7-point Likert scale ranging from 1 (Disagree very much) to 7 (Agree very much). In the current study, the OBQ-20 demonstrated excellent internal consistency ($\alpha = 0.91$), and good internal consistency across the four subscales ($\alpha = 0.77 - 0.85$).

Obsessive Compulsive Inventory – Revised (OCI-R; Foa et al., 2002). The OCI-R is an 18-item self-report measure that assesses obsessive compulsive symptoms and associated distress associated. Participants rate items such as "I find it difficult to control my own thoughts' on a 4-point Likert scale ranging from 0 (not at all) to 4 (extremely). In the current study, the OCI-R demonstrated good internal consistency across the six subscales (α s = 0.65 – 0.87).

Schedule of Imagined Violence (SIV; Grisso et al., 2000). The original SIV is a set of eight items that explore details relating to participants' experience of a violent thought,

including subsequent aggressive actions. The current study only utilised the frequency item of the SIV ("How often do you have thoughts about hurting or injuring other people?"). Participants rated their responses to this item on a 7-point Likert scale ranging from 0 (never) to 7 (several times a day). This item has been used in prior research to measure the frequency of one's aggressive script rehearsal (Daff et al., 2015; Hosie, Simpson, et al., 2022; Podubinski et al., 2017).

Self-Ambivalence Measure (SAM; Bhar & Kyrios, 2007). The SAM is a 19-item measure of self-ambivalence—which encompasses beliefs regarding uncertainty towards the self, and dichotomous perceptions about one's self-concept. Participants respond to items such as "I have mixed feelings about my self-worth" on a 5-point Likert scale ranging from 0 (not at all) to 4 (agree totally). In the current study, the total scale of the SAM was used, and it demonstrated excellent internal consistency ($\alpha = 0.93$).

Thought Control Questionnaire (TCQ; Luciano et al., 2006). A recent psychometric study conducted by Luciano et al. (2006) confirmed a five-factor model of the TCQ with 16 items, which the current study utilised. Participants rate items such as "I punish myself for thinking the thought" on a 4-point Likert scale ranging from 1 (never) to 4 (almost always). The 16-item version of the TCQ demonstrated adequate internal consistency in the current study ($\alpha = 0.67$).

Questionnaire of Unpleasant Intrusive Thoughts (QUIT; Pascual-Vera et al., 2019). Derived from earlier measures of intrusive thoughts (Clark et al., 2014; Garcia-Soriano et al., 2011; Purdon & Clark, 1993), the QUIT assesses the experience of specific themes of intrusive thoughts. Only the unpleasant content domain was used in the current study, and analyses involving the QUIT only used the frequency item of unwanted aggressive intrusive thoughts. Participants rated their responses on a 7-point Likert scale ranging from 0 (never) to 6 (always).

Procedure

After participants provided consent, the online questionnaire was administered with responses recorded anonymously via Qualtrics. After completing demographic questions (e.g., age, gender, place of residence, education level) the measures were presented in random order.

Results

Data Inspection

Data analysis was conducted using IBM SPSS Statistics 27 for PC. Preliminary data inspection revealed that five items were missing from the survey across three variables (one item from OBQ-20; two items from MCAA, and two items from TCQ) due to administration error (n = 223). The survey was amended to include all missing items, and the amended survey was used in all subsequent data collection. Once data collection was completed and an additional 237 participant responses were recorded (N = 460), an Expectation-Maximisation (EM) algorithm was applied to the data set to impute the missing item scores from the initial survey. The utilisation of an EM algorithm for the missing items was justified as Little's MCAR test indicated that the data was Missing Completely at Random (MCAR): [OBQ-20; χ^2 (68, N = 460) = 57.98, p = .80; MCAA; χ^2 (49, N = 460) = 532.12, p = .97; and TCQ; χ^2 (68, N = 460) = 86.97, p = .06]. A missing values analysis was performed on the entire data set and this revealed that several scales also contained missed data. Little's MCAR tests were performed on these scales which indicated that data was MCAR, thus missing data was imputed using the EM algorithm. Data inspection identified several scales were significantly skewed, which is a common trend in non-clinical samples. Transformations on skewed scales were performed to normalise the distributions; however, some of the scales remained skewed. The assumptions for regression analyses were met.

Correlational Analyses

Pearson's correlations were performed between variables (Table 1). Obsessive-compulsive symptoms (i.e., OCI-R scores) were related significantly to general OCD beliefs (OBQ), thought control (TCQ), anger rumination (ARS), history of aggressive behaviour (LHA), and violence supportive beliefs (MCAA) at a weak-to-moderate level, with the strongest correlation with general OCD beliefs. Aggressive intrusive thoughts (QUIT) were significantly related to anger rumination, life history of aggression, criminal attitudes, ego-dystonicity (EDQ), feared self beliefs (FSQ), and self-ambivalence (SAM), at a weak-to-moderate level. Aggressive intrusive thoughts related at a weak-to-moderate level with all OCI-R symptom dimensions except for the washing dimension. Aggressive intrusive thoughts related to all dimensions of obsessive beliefs (OBQ), except for the responsibility, and importance/control of thoughts dimension. Aggressive intrusive thoughts related to all dimensions of thought control (TCQ), except for the worry, and social dimensions. Finally, aggressive scripts (SIV) were found to relate in a significant positive direction with a history of aggressive behaviour, anger rumination, and violence supportive beliefs at a moderate level. Aggressive scripts correlated with AITs, feared self-beliefs, and symptom dimensions

Table 2. Pearson's Correlations for all variables

							OBQ					OCI-R								TCQ							
	ARS	DASS	EDQ	FSQ	LHA	MCAA	Total	1. T	2. R	3. I/C	4. P/U		1. W	2. Ob	3. H	4. C	5. Or	6. N	SAM		1. D	2. P	3. W	4. R	5. S	QUIT_F S	IV_F
ARS	1						OBQ					OCI								TCQ							
DASS	.382**	1																									
EDQ	.058	.250**	1																								
FSQ	.435**	.538**	.233**	1																							
LHA	.390**	.235**	.022	.306**	1																						
MCAA	.417**	.157**	042	.206**	.309**	1																					
OBQ	.417	.157	042	.200	.509	1																					
Total OBQ	.219**	.523**	.358**	.540**	.145**	.085	1																				
1. Threat	.362**	.525 .551**	.336 .251**	.629**	.272**	.171**	.796**	1																			
	.065	.328**	.349**	.029	.091	050	.746**	.433**	1																		
 Responsibility I/C 	.085	.335**	.349	.352**	027	050 .017	.746	.433	.416**	1																	
4. P/U	.225**	.335 .438**	.251**	.352 .458**	.123*	.133**	.839**			.505**	1																
4. P/U	.225	.438	.251	.458	.123	.133	.839	.610	.469	.505	1																
Total OCI-R	.466**	.605**	.287**	.536**	.247**	.237**	.528**	.542**	.296**	.359**	.469**	1															
	.466	.005	.136**	.270**		.129**	.382**		.274**			.477**	1														
1. Washing		.275 .654**	.201**	.548**	.101*	.129	.382 .464**	.333** .528**	.256**		.304	.783**	210**	1													
2. Obsessing	.433**				.271**								.310**	1	4												
3. Hoarding	.301**	.251**	.194**	.294**	.193**	.178**	.234**	.246**	.121*	.165**	.208**	.659**	.293**	.346**	1	4											
4. Checking	.267**	.339**	.176**	.323**	.086	.166**	.338**		.154**		.268**		.428**	.428**	.327**	1 410**	4										
5. Ordering	.393**	.518**	.253**	.422**	.232**	.159**	.470**							.615**			1	4									
6. Neutralising	.254**	.307**	.168**	.335**	.101*	.176**	.358**					.613**						1	4								
SAM	.376**	.636**	.275**	.753**	.259**	.130**	.638**	.669	.355	.415	.5/1	.552	.253	.569**	.269	.286**	.467**	.323**	1								
TCQ	047	454**	407**	4.42**	056	004	242**	470**	405**	240**	402*	240**	247**	404**	442*	402**	4.62**	402**	400**	4							
Total TCQ	.017	.151**	.187**	.142**	.056	.004	.213**	.170**	.195**			.219**						.183**	.190**	1	4						
1. Distract	129**	154**	.128**	182**	046	023	031	059	.022	.032	076	.002	.124*	048	.026	.053	.007		153**	.623**	1						
2. Punish	.176**	.389**	.381**	.417**	.132**	.051	.414**	.344**		.352**						.241**			.481**		017	1					
3. Worry	.207**	.423**	.132**	.332**	.201**	.095	.302**		.177**		.241**					.218**		.166**	.408**	.447**	.051	.367**	1				
4. Reappraise	092	029	023	021	060	150**	.004	021	.103*	030		056	.085	076		026		008	01	.667**	.293**	.045	.061	1			
5. Social	.019	.040	030	.078	.044	.086	.077	.081	.071	.138**	018	.104*	.167**	.064	.031	.116*	.032		.06	.600**	.126*	.127*	.137**	.340**	1		
QUIT	.310**	.280**	.165**	.282**	.142**	.204**	.125*	.258**	021	.007	.144**	.302**	.050	.329**	.189**	.112*	.258**	.209**	.281**	060	103*	.114*	.054	119*	048	1	

SIV	.485**	.246**	.021	.326**	.426**	.374**	.130**	.272**	.037	044	.146**	.333**	.070	.351**	.242**	.124*	.261**	.213**	.249**	073	077	.112*	.079	147**	092	.377**	1
Mean	.813	5.53	126.55	25.90	8.90	14.46	69.51	15.98	21.83	3.48	18.78	3.57	1.24	3.79	1.55	1.18	1.72	.854	40.59	33.65	9.94	.635	3.93	7.27	2.80	.660	1.68
Std.	.135	2.27	29.35	10.88	5.69	2.41	21.22	6.34	6.73	.888	7.35	1.23	.913	2.74	.883	.916	.773	.867	16.46	5.26	2.40	.143	1.29	2.06	.303	1.07	1.13
Min	.600	0	27	8	0	12	19.71	5	5	2.24	4.71	0	0	0	0	0	0	0	4	19.30	4	.480	2	3	2	0	1
Max	1.20	10.95	189	53	23	23.25	131.02	34	35	5.83	35	7.07	3.32	12	3.46	3.46	3.46	3.32	73	51.85	16	1.08	8	12	4	5	7

Note. N = 410. *p < 0.05 **p < 0.01 ***p < 0.001.

ARS = Anger Rumination Scale; DASS = Depression Anxiety Stress Scale; EDQ = Ego Dystonicity Questionnaire; FSQ = Fear of Self Questionnaire; LHA = Life History of Aggression; MCAA = Measure of Criminal Attitudes and Associates; OBQ = Obsessive Beliefs Questionnaire; OCI-R = Obsessive Compulsive Inventory-Revised; SAM = Self Ambivalence Measure; SIV = Schedule of Imagined Violence - Frequency; TCQ = Thought Control Questionnaire; QUIT = Questionnaire of Unpleasant Intrusive Thoughts - Frequency

of the OCI-R except the washing dimension, at a weak-to-moderate level. All dimensions of the OBQ related to aggressive scripts at a weak-to-moderate level, except the responsibility, and importance/control of thoughts dimension. Aggressive scripts related to all dimensions of thought control, except for the distraction, worry, and social dimensions.

Regression Analyses

Regression analyses are presented in Table 2. Depression, anxiety, and stress were entered as a total at stage 1, and all other variables at stage 2. Regression analyses examined the predictors of AITs (QUIT) and aggressive script rehearsal (SIV). The regression analysis also examined the predictors of life history of aggression, and OC symptoms. The assumptions for regression analyses were met.

Examining the differential predictors of aggressive intrusive thoughts (QUIT), the DASS at stage 1 explained 8% of the variance in QUIT, and the remainder 10 predictors explained an additional 15%, F change (10, 398) = 7.85, p < .001. In the final model, the SIV , EDQ, OBQ, and TCQ were all significant predictors of QUIT. Examining predictors of aggressive script rehearsal (SIV), the DASS explained 6% of the variance in aggressive scripts, with the additional 10 predictors explaining an additional 32% of the variance in the SIV after controlling for psychological well-being, F change (10, 398) = 20.73, p < .001. In the final model the ARS, LHA, QUIT, MCAA, and TCQ were all significant predictors of aggressive script rehearsal. Examination of regression weights identified that the influence of obsessive beliefs on QUIT was negative. This result indicated a suppression effect, as the OBQ is a univariate positive predictor of QUIT frequency. To identify which variables were suppressing the OBQ, repeated regression analyses were conducted where each variable in the model was excluded in turn from the analysis. These analyses indicated that the DASS

Table 3. Beta coefficients (t statistics) for regression analyses predicting AITs, Aggressive scripts (SIV), obsessive-compulsive symptoms (OCI), and life history of aggression (LHA).

	QUIT		SIV		OCI-R		LHA	•
	Step 1	Step 2	Step 1	Step 2	Step 1	Step 2	Step 1	Step 2
DASS	0.28 (5.89***)	0.089 (1.42)	0.253 (5.27***)	-0.030 (-0.53)	0.607 (15.44***)	0.296 (6.30***)	0.232 (4.81***)	0.051 (0.82)
ARS		0.059 (1.01)		0.233 (4.54***)		0.187 (4.16***)		0.154 (2.67**)
EDQ		0.126 (2.63*)		-0.02 (-0.45)		0.077 (2.03*)		-0.026 (-0.54)
FSQ		0.029 (0.41)		0.107 (1.68)		0.098 (1.77)		0.092 (1.31)
MCCA		0.064 (1.28)		0.120 (2.69**)		0.036 (0.92)		0.137 (2.77**)
OBQ		-0.152 (-2.46*)		-0.038 (-0.69)		0.201 (4.20***)		-0.032 (-0.52)
SAM		0.148 (1.89)		-0.041 (-0.58)		0.009 (0.14)		0.06 (0.77)
TCQ		-0.092 (-1.98*)		-0.101 (-2.44*)		0.108 (2.99**)		0.061 (1.32)
OCI-R		0.098 (1.53)		0.125 (2.18*)		-		-0.036 (0.57)
LHA		-0.068 (-1.36)		0.214 (4.87***)		-0.023 (-0.57)		-
SIV		0.255 (4.68***)		-		0.095 (2.19*)		0.263 (4.87***
QUIT		-		0.204 (4.68***)		0.06 (1.54)		-0.067 (-1.36)
R^2 ΔR^2	0.078	0.230 0.152**	0.064	0.384 0.321***	0.369	0.530 0.161***	0.054	0.241 0.187***

Step 1: DASS; Step 2: ARS, EDQ, FSQ, MCAA - Violence, OBQ, SAM, TCQ, OCI, SIV. Standardised beta coefficients; t statistics in parentheses.

Note. *p < 0.05 **p < 0.01 ***p < 0.001.

Note. P 40.03 P 40.01. P 40.001. Note. N = 410. ARS = Anger Rumination Scale; DASS = Depression Anxiety Stress Scale; EDQ = Ego Dystonicity Questionnaire; FSQ = Fear of Self Questionnaire; LHA = Life History of Aggression; MCAA = Measure of Criminal Attitudes and Associates-Violence Subscale; OBQ = Obsessive Beliefs Questionnaire; OCI-R = Obsessive Compulsive Inventory-Revised; SAM = Self Ambivalence Measure; SIV = Schedule of Imagined Violence – Frequency; TCQ = Thought Control Questionnaire; QUIT = Questionnaire of Unpleasant Intrusive Thoughts – Frequency

OCI-R, FSQ and SAM were suppressing the OBQ, with obsessional beliefs non-significantly associated with QUIT (β = .003, p >.05) when they were removed from the equation.

Repeating the analysis for OC symptoms, the DASS explained 37% of the variance and a further 16% of the variance was predicted by the other variables, F change (10, 398) = 13.60, p < .001. In the final model, the DASS, OBQ, ARS, TCQ, SIV, and the EDQ were significant predictors. For a history of aggressive behaviour, the DASS explained 5% of the variance, and the additional 10 predictors were entered at stage 2, the total variance explained by the entire model was 24%, F (11, 398) = 11.47, p < .001. The inclusion of 10 predictors explaining an additional 19%, F change (10, 398) = 9.80, p < .001. Overall, the SIV, ARS, and the MCAA were significant.

Discussion

This study investigated the association between different OCD-relevant beliefs, self-themes, and aggression-related beliefs with AITs and aggressive script rehearsal. As hypothesised, general OCD beliefs and thought control strategies predicted OC symptoms. Unexpectedly, anger rumination and the frequency of aggressive script rehearsal also predicted OC symptoms. It was also found, as hypothesised, that ego-dystonicity predicted the experience of AITs. Contrary to expectations, the feared self was not identified as a unique predictor of AITs or aggressive script rehearsal. As hypothesised, violence supportive beliefs, a history of aggressive behaviour, anger rumination, and the use of thought control strategies predicted aggressive script rehearsal. It was also found, as hypothesised, that anger rumination, violence supportive beliefs, and the frequency of aggressive script rehearsal predicted a history of aggressive behaviour. No relationship was found between aggressive scripts and ego-dystonicity. These findings are unpacked in detail below.

Firstly, and consistent with prior research (Moulding, Coles, et al., 2014; Purdon & Clark, 1994b), general OCD beliefs and thought control strategies were found to predict OC symptoms. Cognitive appraisal models of OCD postulate that maladaptive beliefs influence the appraisal of intrusive thought experiences (Radomsky et al., 2014) which motivate the use of compulsive behaviours, inadvertently perpetuating the intrusive thoughts (Belloch et al., 2004; Brakoulias et al., 2014; Purdon & Clark, 1993; Salkovskis, 1985; Wheaton et al., 2010) The current study found significant correlations between the thought control strategies of punishment and worry with OC symptoms, suggesting that certain thought control strategies may be less effective than others at controlling the reoccurrence of symptoms; similar findings have been highlighted by Jacoby et al. (2015) who found that using self-punishment as a means of controlling intrusive thoughts, was associated with frequent repugnant intrusive thoughts. Contrary to expectations, at a multivariate level, anger rumination and aggressive script rehearsal were also found to predict the experience of OC symptoms. An explanation for this finding may concern the measurement of anger rumination and aggressive script rehearsal, which enquires generally about aggressive thinking and may share similarities to measurements of AIT in OCD. It is possible that current measurements of anger rumination, aggressive script rehearsal, and AITs in OCD are unable to clearly differentiate the phenomena, and thus interrelationships across constructs are being identified.

Secondly, ego-dystonicity and self-themes including the feared self and self-ambivalence, were examined. At a univariate level, ego-dystonicity, feared self beliefs, and self-ambivalence all were significantly related to the experience of AITs. However, at a multivariate level, ego-dystonicity was found to be the only self-related predictor of AITs. This aligns with previous research on ego-dystonicity (Purdon et al., 2007), where experiencing thoughts of harming another person that do not reflect one's intentions, are likely to be interpreted as abhorrent to the self (Lee & Kwon, 2003). Obsessive beliefs were a

positive predictor of AITs only when certain measures (e.g., DASS, OCI-R, FSQ, and SAM) were removed from the model, with analyses indicating a suppression effect due to these variables. One explanation for this finding is that by controlling for depression, anxiety, and OC symptoms, the relationship between AITs and obsessive beliefs no longer identifies the symptomatic elements of intrusive thought experiences. It is unclear how the self-measures of FSQ and SAM influenced the suppression effect, however one may speculate that these measures may contain anxiety related elements, and when controlled for, the relationship between AITs and obsessive beliefs is no longer able to identify these elements. As these analyses involved a post-hoc exploration it is therefore essential that these findings be confirmed in further studies of AITs and obsessive beliefs.

Thirdly, our results suggest an association between the frequency of aggressive script rehearsal and anger rumination, violence supportive beliefs, a history of aggressive behaviour, thought control strategies, AITs, and contrary to expectations, also OC symptoms. No relationship was found between aggressive script rehearsal and ego-dystonicity. The current findings support the notion that the presence of violence supportive beliefs and prior acts of aggression increase the likelihood of aggressive script rehearsal (Hosie, Simpson, et al., 2022; Kelty et al., 2011). Contrary to expectations, thought control strategies were found to have a negative association with aggressive script rehearsal when considered alongside other predictors. This may suggest that the use of thought control strategies for aggressive scripts may act as a protective factor by distracting or occupying one's thoughts on something other than aggression. This aligns with Nagtegaal et al. (2006) who found that distraction, when used as a control strategy, reduced the likelihood of aggressive behaviour in participants examined. Additionally, the association between AITs and aggressive script rehearsal can be explained by similarities in measurement instruments. Both measures of AITs and aggressive script rehearsal ask respondents whether they have ever experienced a

thought of harming another person. Although it is asked in different ways in instruments used to measure phenomena thought to be related to OCD and aggressive behaviour, the overarching content appears to be the same and extant instruments may not have been designed with consideration given to differences in phenomena (AITs and aggressive scripts) across different populations. This appears to demonstrate a critical issue with the measurement of these constructs, and a refinement of these measurement instruments is required. Based on the findings of this study, AITs and aggressive script rehearsal may be differentiated by a history of aggressive behaviour, endorsement of violence supportive beliefs, and ego-dystonicity. Further clinical explorations of these factors is warranted for the purposes of risk assessment and phenomenological understandings of these constructs.

The lack of association between ego-dystonicity and aggressive script rehearsal may be explained by the anecdotal assumption that individuals with a history of aggressive behaviour or who endorse violence supportive beliefs, may experience aggressive thoughts as ego-syntonic (i.e., consistent with one's self-concept). As highlighted by Purdon et al. (2007), thoughts that are initially appraised as ego-dystonic can over time be accommodated for into one's self concept, leading that thought to be interpreted as ego-syntonic. It may be that a measure of ego-syntonicity may be more sensitive towards identifying whether an association exists between one's self concept and the process of aggressive script rehearsal—which was not investigated in the present study.

Fourthly, anger rumination, aggressive script rehearsal, and violence supportive beliefs were significant predictors of one's propensity to have acted aggressively in the past. This finding is consistent with prior research which suggests that the more one ruminates on aggressive altercations or acts of revenge, the more likely aggressive behaviour becomes a part of one's repertoire (Daff et al., 2015; Denson, 2013). Life history of aggression was not

predicted by AITs which further confirms the understanding that OCD related AITs are not associated with overt acts of violence (Veale et al., 2009).

The present study has several strengths, including its novel approach of concurrently measuring AITs and aggressive scripts, as well as exploring relevant features of these phenomena with the aim to better differentiate these constructs from each other. Since this study is one of the first to examine AITs and aggressive script rehearsal concurrently, it has brought attention to the complex task of differentiating AITs from aggressive script rehearsal, and through this, has identified issues with measurement instruments of these constructs. The present study's findings also suggest there are differences in the phenomenology of these constructs, where it appears AITs and aggressive script rehearsal differ particularly on factors concerning a history of aggressive behaviour, endorsement of violence supportive beliefs, and the experience of ego-dystonicity. These features warrant consideration when assessing violence risk, specifically when determining which elements of aggressive thoughts indicate one's propensity to act aggressively, when compared to others.

Nevertheless, the present study's findings should be considered in light of several limitations. The study utilised a cross-sectional design with self-report measures, and a limited sample size which prevents causal assumptions being made regarding the relationships found. Similarly, common method variance introduced with self-report instruments may have influenced the relationships found, and thus results should be interpreted with caution. The use of a non-clinical sample of participants, though a common practice in OCD research (Abramowitz et al., 2014), influences the severity and extent to which OC symptoms and obsessive beliefs are reported. Although the current study was able to identify differential predictors of AITs and aggressive script rehearsal, it should be acknowledged that a limited number of participants endorsed such cognitions, and thus relationships that consider these phenomena should be interpreted accordingly.

The present study supports the relationship that violence supportive beliefs and a history of aggressive behaviour have with aggressive script rehearsal. Whether egosyntonicity is an important feature of aggressive script rehearsal remains unclear, so further exploration is required. This study has also demonstrated the potential overlap that exists between current measurements of AITs and aggressive script rehearsal, and further explorations of these phenomena concurrently is warranted, where the refinement of these measures would have implications for the clinical utility of these instruments. This study provides implications for risk assessments, as preliminary features that distinguish AITs from aggressive script rehearsal were identified empirically: a history of aggressive behaviour, violence supportive beliefs, and ego-dystonicity. Further, examining these phenomena with clinical population groups may prove beneficial in understanding how these constructs are experienced and maintained by those who frequently report them.

CHAPTER 8 – EMPIRICAL RESEARCH STUDY TWO

8.1 Preamble to Empirical Research Paper Two

This chapter presents the second empirical research study of this thesis. Empirical study two addresses the second and third thesis aims, to examine the similarities and differences between AITs and aggressive scripts, and to explore the subjective experiences of aggressive thoughts, respectively. Extending from the findings of empirical study one, empirical study two will further explore features pertinent to aggressive script rehearsal, including the subjective experiences associated with aggressive thinking.

8.2 Recruitment Context for Empirical Study Two

Recruitment for study two was scheduled to begin in June 2021. Due to the global COVID-19 pandemic and Victorian Government lockdown restrictions, in person data collection was unable to begin in 2021. Specifically, data collection strategies for study two involved recruitment of participants from group treatment programs (STOP OCD Program, and the Handling Anger Wisely Program), and both these groups were cancelled in 2021 and rescheduled to resume in early 2022.

Amendments to the Swinburne University Research Ethics Committee and Forensicare's Operational Research Committee applications were made to accommodate for online video or telephone interviews to be conducted. Despite these amendments, community treatment clinicians involved in supporting the recruitment of required participants were unable to identify participants who, (a) met the eligibility criteria for the study, or (b) were comfortable in sharing their experiences of AITs or aggressive scripts. It is worth noting that the student researcher was able to liaise with community OCD treatment clinicians, and it was identified that a common barrier for research participation in individuals diagnosed with

OCD and who experience AITs was the overwhelming fear associated with disclosing their aggressive thoughts, and the perceived repercussions associated with this disclosure.

Data collection resumed in early 2022 when both group programs were scheduled to resume. Despite having several individuals from the STOP OCD group program interested in participating, these individuals did not experience AITs and were therefore not eligible for the study. Between January and July 2022, four individuals from the Community Forensic Mental Health Service participated in the study. Data collection remained open until July 2022, until the decision was made by the student researcher and supervisory team to cease data collection. The decision to abandon the recruitment of the OCD cohort was based on: a) low likelihood of further recruitment due to the ongoing impact of COVID-19 on clinical services, which were experiencing significant demand for services in the face of considerable staffing shortages due to COVID and had thus limited capacity to engage in any research activities; b) the limited timeline for the project; c) already extensive literature on the nature and experience of AITs in OCD; and d) the findings emerging from the forensic population sample which provided a rich and informative perspective on the different experience of both AITs and aggressive scripts. Given the limited parameters of the current project it was deemed that the level of detail obtained was appropriate for an initial exploration for the area, and provided a basis for further exploration and potential comparisons to OCD populations in future research.

8.3 Authorship Indication Form for Empirical Research Paper Two



Swinburne Research

Authorship Indication Form

For PhD by Publication candidates

NOTE

This Authorship Indication form is a statement detailing the percentage of the contribution of each author in each published 'paper'. This form must be signed by each co-author and the Principal Coordinating Supervisor. This form must be added to the publication of your final thesis as an appendix. Please fill out a separate form for each published paper to be included in your thesis.

We hereby declare our contribution to the publication of the ' Exploring the Experiences of Aggressive Scripts in a Forensic	
First Author	-
Stephanie Jane Fernandez Name	Signature:
Percentage of contribution: _80_%	Date: 20 / 02 / 2023
Brief description of contribution to the 'paper' and your centra	
Reviewing the literature, study design, data collection, data	апагузіз, рієраїєй піапизстірі
Second Author	
Name: Michael Daffem	Signature:
Percentage of contribution:5_%	Date: 21/02/23_
Brief description of your contribution to the 'paper': Supervision on study design and recruitment processes,	ritically reviewed and edited manuscrint
Richard Moulding Name: 5 w	Signature: Olloy
Percentage of contribution:5 %	Date: 20 / 02 / 2023
Brief description of your contribution to the 'paper':	
Critically reviewed and edited manuscript	
Fourth Author	
Name: Maja Nedeljkovic	Signature:
Percentage of contribution: 10 %	Date: 20 / 02 / 2023
Brief description of your contribution to the 'paper':	
Primary supervision at all stages, involvement in data analy	vsis critically reviewed and edited manuscrint
· mary caper roles at all caages, morrolless in data and	icio, cinacany icinamacana ana canacana manacana
Principal Coordinating Supervisor:	
Maia Nadalikavia	ignature: Ollkey
Date: 20/02 / 2023	
la the case of more than favor suthern places attack and	

In the case of more than four authors please attach another sheet with the names, signatures and contribution of the authors.

Exploring the Experiences of Aggressive Script Rehearsal in a Sample of Adult Males

Recruited from a Forensic Mental Health Service

Stephanie Jane Fernandez^{a,b}, Michael Daffern^{a,b}, Richard Moulding^c, and Maja Nedeljkovic^a

^aDepartment of Psychological Sciences, Swinburne University of Technology, John Street,

Hawthorn, Melbourne, Australia, ^bCentre for Forensic Behavioural Science, Swinburne

University of Technology, 582 Heidelberg, Alphington, Melbourne, Australia, ^cThe

Cairnmillar Institute, Melbourne, Australia

Corresponding author: Stephanie Fernandez, Centre for Forensic Behavioural Science, 582

Heidelberg, Alphington, Victoria, Australia, 3078; Email: sfernandez@swin.edu.au

Contact number: +61 0421 325 795

Acknowledgements

Funding: The first author received funding support from the Australian Government in the form of an Australian Government Research Training Program Scholarship

Declarations of interest: none

Declaration: This manuscript has not been published elsewhere and has not been submitted simultaneously for publication elsewhere.

Abstract

Aggressive script rehearsal is commonly reported by people with a history of violence and is also a key treatment target in violence intervention programmes. Intrusive thoughts about violence are also a common symptom of Obsessive Compulsive Disorder (OCD) and are experienced frequently, but are not associated with a history of violence. Within forensic practice, the rehearsal of aggressive scripts informs violence risk assessment with repeated rehearsal increasing the potential for violence, particularly when these thoughts are pleasurable and/or when they contain thoughts of seriously harmful behaviours. The features of aggressive scripts experienced by people with a history of violence have not been the focus of much research. This study explored aggressive scripts in a sample of adults with a history of aggression or with problematic expression/control of anger (N = 4). All participants were recruited from a community forensic mental health service. Both qualitative and quantitative data were collected during semi-structured interviews, and an inductive thematic analysis was performed to identify themes in interview transcripts. Quantitative data provided context to explore relationships within the thematic analysis. Participants described several features related to their aggressive thinking with five main themes identified: precipitants, negative impacts, negative experiences, positive experiences, and management. The emotional experiences associated with aggressive thoughts was dependent on the thought content, and a range of thought management strategies were explored. While only having a limited sample size due to the difficulties in accessing this population, this study highlights the importance of exploring several facets of aggressive thinking, beyond beliefs that may support violence, and provides information on the subjective experiences associated with aggressive thoughts.

Keywords: Aggressive scripts; history of violence; thought management; subjective experiences

Exploring the Experiences of Aggressive Script Rehearsal in a Sample of Adult Males Recruited from a Forensic Mental Health Service

Experiencing thoughts about harming another person is a normal phenomenon, commonly reported within the general population (Kenrick & Sheets, 1993; Rowa & Purdon, 2003). However, aggressive thoughts are also reported by people with various mental health disorders, including Obsessive Compulsive Disorder (OCD) where they are experienced as intrusive, repetitive, distressing, and difficult to control (Moulding, Aardema, et al., 2014a; Rachman, 1997). In the OCD field these thoughts are commonly referred to as aggressive intrusive thoughts (AITs) or obsessions, and they are generally not associated with a history or risk of aggression (Veale et al., 2009). However, thoughts about harming another person are often included in risk assessment protocols within forensic settings, as these thoughts are commonly reported by people with a history of violence or who have problems with the regulation of anger (Daff et al., 2015; Hosie et al., 2021). Within forensic settings, thoughts about harming another person are often referred to as aggressive scripts, and repeated rehearsal of aggressive scripts, particularly when they relate to more serious acts of violence and/or when they are associated with pleasurable emotions, are associated with aggressive behaviour (Daff et al., 2015; Gilbert et al., 2013; Hosie, Dunne, et al., 2022).

Aggressive scripts act as mental templates for aggressive behaviour and are created through observations of aggressive actions (Huesmann & Eron, 1984). Social cognitive models of aggression posit that an individual's propensity to act aggressively can be attributed to how frequently they mentally rehearse aggressive scripts (Bushman & Anderson, 2002; Gilbert et al., 2013; Huesmann, 1988), as well as the extent to which they hold antisocial attitudes, and have experienced aggressive behaviour in their lifetime (Andrews et al., 2011; Coccaro et al., 1997; Mills et al., 2002). As highlighted by Gilbert and Daffern (2017), individuals with a history of violent offending typically report more frequent

rehearsal of aggressive scripts and that these scripts are more established and accessible in mind, thus increasing the likelihood of future aggressive actions. In contrast, AITs are defined as intrusive, repetitive, and unwanted thoughts about harming another person, and are experienced as significantly distressing, abhorrent, and inconsistent with one's view of self (Rachman, 1981, 1997). Individuals with OCD who report these intrusions are often reluctant to seek treatment or speak about their thoughts as they are ashamed about what these thoughts may mean about them, as well as fearing being judged negatively (e.g., 'having these thoughts means I'm a terrible, dangerous person'; Veale et al., 2009). Findings from DeLapp et al. (2018) suggest that AITs can exist in individuals with a history of violence, with results revealing that the frequency of AITs did not differ between individuals incarcerated for violent offending (n = 78) and a student sample (n = 103). It is important to highlight that both AITs and aggressive scripts can be reported by individuals with a history of violence, and therefore being able to differentiate between these two phenomena has implications for the assessment and treatment of these constructs. Although AITs and aggressive scripts share similarities in terms of thought content, there appears to be a clear difference in the behavioural outcomes associated. This may be attributed to the distinct population groups these constructs are investigated in, but this may also indicate differences in phenomenological processes involved in these constructs. The features that separate AITs and aggressive scripts are not clearly understood. Further, exploring the features associated with aggressive script rehearsal in individuals with a violent history may prove beneficial in understanding the factors that influence levels of violence risk.

Cognitive Explanations of AITs and Aggressive Scripts: Consideration of Self-themes

AITs in OCD research are widely regarded as ego-dystonic, distinguishing them from aggressive scripts in offender populations (Fernandez et al., 2022), and furthermore, separating them from other normative cognitions (e.g., rumination, negative thoughts in

depression; Belloch et al., 2012). A thought is considered ego-dystonic when it is perceived by the individual as being inconsistent with their sense of self, beliefs, and past behaviour (Purdon et al., 2007). Further, whether a thought is considered ego-dystonic is also dependent on the thought content and the degree to which the individual experiences the thought content as being personally salient (Purdon & Clark, 1999). With regards to repugnant obsessions (i.e., intrusive thoughts encompassing harm towards other people and aggression, sexual themes, and contents that are personally repugnant or immoral), Aardema et al. (2013) identified that ego-dystonicity appeared as a unique predictor of these type of intrusions.

With regards to aggressive scripts, the General Aggression Model and Script Theory postulate that the frequent rehearsal of aggressive scripts is associated with acts of violence (Bushman & Anderson, 2002; Daff et al., 2015), and aggressive scripts are maintained because they are seen as acceptable to the individual, with the frequency of aggressive script rehearsal correlating with the strength of endorsement with violence supportive beliefs (Gilbert & Daffern, 2017). Further, cognitive models of aggressive thinking such as the Multiple Systems Model (Denson, 2013) posit that the precursors to aggression likely concern the extent to which individuals engage in anger rumination—perseverative thinking concerning experiences of anger, and ruminations of past provocations. Research has identified the role that normative beliefs about violence have on the experience and maintenance of aggressive thinking over time (Gilbert et al., 2013; Hosie et al., 2021), but little is understood about how these thoughts are appraised by those who experience them (Hosie, Dunne, et al., 2022; Patel, 2015).

A recent study by Fernandez et al. (2022) found that in a non-clinical sample (N=410), aggressive scripts and AITs may be differentiated by features related to ego-dystonicity, violence supportive beliefs, and a life history of violence. In support of the current understanding of scripts (Hosie, Dunne, et al., 2022; Kelty et al., 2011), Fernandez et

al. (2022) identified that the presence of violence supportive beliefs and a history of aggressive behaviour was related to the rehearsal of aggressive scripts. In line with prior research (Veale et al., 2009), violence supportive beliefs and a history of aggressive behaviour were not associated with AITs. Consistent with phenomenological underpinnings of repugnant intrusive thoughts (Moulding, Aardema, et al., 2014a; Purdon et al., 2007), Fernandez et al. (2022) identified that AITs were related to ego-dystonic beliefs, and no association was found between the experience of ego-dystonic beliefs and aggressive script rehearsal. Given there is little empirical examination of the subjective experiences associated with aggressive script rehearsal, anecdotal assumptions purport that an individual with a history of aggression, who holds normative beliefs about aggression, is likely to experience their aggressive script rehearsal as ego-syntonic—i.e., consistent with one's beliefs, intentions, or past experiences (Belloch et al., 2012). Empirical investigations exploring the subjective experiences associated with aggressive scripts is warranted, as understanding the features associated with aggressive scripts may have implications for assessment and treatment of these phenomena.

Purdon et al. (2007) highlighted how an intrusive thought that is initially experienced as ego-dystonic, can over time, be appraised as ego-syntonic as it becomes accommodated within a person self-view rather than resisted. This change in self-view is said to include feared self-perceptions, where the individual may view themselves as being dangerous, bad, or immoral (Purdon et al., 2007). While this change in self-view may influence an ego-dystonic AIT to be perceived as ego-syntonic, it is important to highlight that in OCD although individuals may no longer resist the thought, compulsive behaviours are still used to reduce the perceived consequences of the AIT, and one's intention to act aggressively or in accordance with their AITs does not appear to change (Veale et al., 2009). Changes in self-view have also been explored with reference to aggression and violent behaviour in forensic

populations, however, in contexts where individuals rehearse aggressive scripts repeatedly, the perpetration of aggressive behaviour becomes normalised, as this behaviour becomes assimilated as being part of one's self-view (Patel, 2015). Further examination of the experiences associated with aggressive thinking will prove useful for risk assessment and treatment, as well as the potential for using subjective differences to differentiate aggressive thinking from other similar thought phenomena.

Differential Features of AITs versus aggressive scripts

Research has identified the key features that differentiate intrusive thoughts from other anxiety and mood related phenomena (Clark, 2004). These features include level of frequency, distress, intrusiveness, unwantedness, ego-dystonicity, disruption in functioning, and spontaneity (Clark, 2004; Moulding, Aardema, et al., 2014a). Whether these features can be used to differentiate AITs from aggressive scripts requires further empirical exploration. In a qualitative investigation of individuals with a history of violent behaviour, Patel (2015) found that individuals were likely to describe their aggressive thinking as disturbing and unwanted, and that distraction strategies were used to manage these thought experiences. Research examining the use of thought control strategies to manage aggressive thoughts have revealed that the use of certain thought control strategies such as distraction may reduce aggression (Nagtegaal et al., 2006). The features identified by Patel (2015) indicate that aggressive thoughts in individuals with a history of aggressive behaviour may be experienced similarly to AITs in OCD, specifically how these thoughts are described as disturbing and unwanted, and the use of thought control to manage the thought occurrence. However, the similarities between these constructs leads to questions regarding what features differentiate these phenomena.

Further, the frequency of script rehearsal and emotional experience associated with the rehearsal of aggressive scripts in forensic populations has received some attention (Hosie et al., 2021), with results showing that the frequency of aggressive script rehearsal was related to greater life history of violence. Hosie et al. (2021) also identified that the most common emotional experience associated with aggressive script rehearsal was anger, followed by hate, fear, sadness, disgust, confusion, and annoyance. Also, individuals with a greater history of violence were more likely to report that their aggressive thinking was associated with excitement (Hosie et al., 2021). Hosie et al. (2021) also showed that common precipitants of aggressive script rehearsal included themes around family protection/betrayal and feeling belittled and disrespected by others. Hosie et al. (2021) highlighted the need for further research to consider the identification of themes associated with aggressive script rehearsal, as this may likely improve understanding of this construct including the nature and purpose of script rehearsal.

Aggressive script rehearsal serves planning and emotional regulation functions (Hosie, Dunne, et al., 2022; Patel, 2015). Emotion regulation is the process by which individuals affect the way they experience and express their emotions (Gross, 1998). In a study exploring relationships between emotion regulation difficulties, aggressive script rehearsal, and aggressive behaviour in an incarcerated male sample (N = 129), Hosie, Dunne, et al. (2022) found that the frequency of aggressive script rehearsal was significantly positively correlated with emotional regulation difficulties, specifically impulse control difficulties and experiencing limited confidence in using emotional regulation strategies. Further, Hosie, Dunne, et al. (2022) suggest that that aggressive script rehearsal may serve as a cognitive response modulation strategy, alleviating negative emotional experiences, or used as a way to plan an individual's retaliation against the perceived causes of negative affect. The emotional regulation function of aggressive scripts has been considered in previous research, where Patel (2015) identified, in a qualitative analysis of offenders' aggressive thoughts, that themes related to power and control, and coping, were associated with

emotional regulation functions. Further exploration of the function of aggressive thinking is warranted as understanding what influences individuals to engage in the rehearsal of these thoughts may help elucidate features of this phenomena and its functions.

Current Study

In recent years the importance of aggressive script rehearsal to aggressive behaviour has been highlighted, suggesting implications for violence risk assessment. At the same time, questions have been asked about phenomenological overlap with AITs, which, despite sharing aggressive content, seem unrelated to aggressive behaviour. This study aimed to explore the subjective experience of individuals with a history of violent behaviour or problematic anger. Specifically, the study aimed to elucidate the cognitive, emotional and behavioural features of aggressive scripts and AITs in a forensic sample, and to examine the role that specific characteristics, including intrusiveness, spontaneity, ego-dystonicity, feared self, and level of distress, have on the experience of these scripts and AITs. The specific research questions of the study were:

- 1. What are the subjective experiences associated with aggressive thoughts (both in the form of scripts and AITs) in a forensic sample?
- 2. Are there specific features, beliefs, or experiences associated with aggressive thoughts in a forensic sample?
- 3. What are the emotional and behavioural outcomes associated with aggressive thinking in a forensic sample?

Method

Research design

Qualitative data was collected to identify participants experiences of aggressive thinking, with quantitative data describing symptom severities, and clinical cut-offs used to contextualise the subjective experiences of aggressive thoughts. As highlighted by Yardley

and Bishop (2007), the integration of qualitative and quantitative data should follow a pragmatism approach with the goal of data inquiry not only based on identifying the truth in human experience, but the richness of the experience of individuals by combining idiosyncratic exploration and scientific approaches. This pragmatic viewpoint was used as a basis for the integration of qualitative and quantitative data in this study. Given the absence of prior research examining participants' descriptions of their personal experience of aggressive scripts and AITs in verbatim dialogue, it was decided prior to data collection that greater significance would be given to the qualitative data rather than the quantitative data. Further, given the exploratory nature of the study, precedence was given to the quality and richness of interviews, rather than sample size. Nevertheless, the small sample also did not permit the use of inferential statistics when analysing the quantitative data.

Participants

Participants were recruited from a Community Forensic Mental Health Service (CFMHS), a state-wide forensic mental health service in Victoria, Australia. All participants had a history of violence or were reported or considered, by their treating psychiatrist or psychologist, to have problems with anger or aggression. Participants had to be over 18 years and able to consent voluntarily, could not have a past or present experience of psychosis, and were required to have English language skills that would allow for conversation during interviews. All genders were invited to participate. Participants with a history of interpersonal violence, aggression, or anger problems were sought given the increased likelihood that aggressive thoughts are reported in individuals with such histories. Participants were offered a AUD\$30 gift voucher (excluding alcohol purchases) in appreciation of their participation. Four males agreed to participate, and they were aged between 23 and 49 (M = 36.50; SD = 10.76).

Materials

Symptom measures

Several self-report and symptom measures were administered to provide descriptive statistics about the sample and to inform the semi-structured interview questions discussed below. These measures included the (1) Anger Rumination Scale - Thoughts of Revenge Subscale (ARS; Sukhodolsky et al., 2001), (2) Depression Anxiety Stress Scales - Short Form version (DASS-21; Lovibond & Lovibond, 1995), (3) Ego-Dystonicity Questionnaire-Reduced Version (EDQ-R; Belloch et al., 2012), (4) Fear of Self Questionnaire (FSQ; Aardema et al., 2013), (5) Life History of Aggression – Aggression Subscale (LHA; Coccaro et al., 1997), (6) Measures of Criminal Attitudes and Associations – Violence Subscale (MCAA; Mills et al., 2002), (7) Obsessive Beliefs Questionnaire (OBQ-20; Moulding et al., 2011), (8) Obsessive Compulsive Inventory Revised (OCI-R; Foa et al., 2002), (9) Self-Ambivalence Measure (SAM; Bhar & Kyrios, 2007), (10) Social Desirability Scale (SDS; Stöber, 2001), (11) Thought Control Questionnaire (TCQ; Luciano et al., 2006), (12) Schedule of Imagined Violence – Frequency Item (SIV; Grisso et al., 2000), and (13) Questionnaire of Unpleasant Intrusive Thoughts – Unpleasant Content Domain (QUIT; Pascual-Vera et al., 2019). For brevity, details on the psychometric properties of these measures are included in supplementary material S1 (Appendix Z).

Semi-structured interview

This study used a semi-structured interview format, created for the purpose of investigating the research questions of the study. The interview schedule was developed by the authors and was derived from existing measures of AITs (Questionnaire of Unpleasant Intrusive Thoughts [QUIT]; Pascual-Vera et al., 2019) and aggressive script rehearsal (Schedule of Imagined Violence [SIV]; Grisso et al., 2000). A semi-structured interview format allowed the researcher to employ follow-up style questions after participant responses,

including their responses to the self-report measures. For example, participants who reported rehearsing aggressive thoughts were asked to describe their experiences such as to whether the aggressive thoughts were experienced as intrusive, spontaneous, or distressing.

Procedure

Recruitment and data collection were ongoing from September 2021 to August 2022. After informed consent was obtained, voluntary participation, and the right to withdraw were discussed, the SIV and QUIT were administered alongside the semi-structured interview and were followed by a battery of self-report questionnaires. Responses to the questionnaires were followed with further semi-structured questioning and probing. All interviews were audio recorded using Otter.ai software. Interviews were transcribed by (author initials redacted) using Otter.ai software. The interview transcripts were then imported into NVivo software package for qualitative analysis. Quantitative data obtained from self-report questionnaires were analysed in IBM SPSS Statistics 27 for PC to generate descriptive statistics for the sample.

Qualitative and Quantitative Analysis

Qualitative analysis involved the use of inductive thematic analysis on the interview transcripts, across both sample groups. As highlighted by Braun and Clarke (2006), inductive or 'bottom up' thematic analysis involves identifying themes that are derived from the data, as opposed to a theoretical or analytic interest guiding the identification of themes (i.e., deductive or 'top down' thematic analysis). Table 4 presents the procedure for thematic analyses used in the current study, derived from the six phases of thematic analysis by Braun and Clarke (2006).

.

Table 4. The six phases of thematic analysis used in the current study derived from Braun and Clarke (2006)

Phase	Description of what occurred
1. Familiarising yourself with your data	The first author conducted all interviews and reviewed transcriptions
	uploaded to Otter.ai.
2. Generating initial codes	The first and last author generated initial codes (together and
	individually) using NVivo 11, and met frequently to discuss code conten
	and process
3. Generating initial themes	During discussions with the first and last author, codes were
	organised on piece of paper and arranged into preliminary
	themes. Several codes were discarded or reorganised.
4. Reviewing identified themes	Through team discussions, codes were reviewed under each preliminary
	themes. Codes were reorganised or discarded accordingly. The
	relationship between codes and themes were explored as a team.
5. Defining and naming themes	Theme names were reviewed, and cross checked with code content during
	team discussions. Final theme names were decided during team
	discussions.
6. Producing the report	A draft of the report was written by the first author using identified
	themes and certain extracts. Further drafts received feedback by all
	authors.

Total scores for the self-report measures were calculated to provide supplementary findings to the qualitative results. As the study utilised a small sample, inferential analyses could not be completed. Quantitative and qualitative data were analysed separately and integrated during the data interpretation phase.

Results

Quantitative findings

Table 5 presents psychometric scores for each participant from the self-report measures and clinical cut-off scores for the clinical measures from Lovibond and Lovibond (1995) and Abramovitch et al. (2020), respectively (i.e., for the DASS-21, the OCI-R). According to scores on the clinical measures, these scores indicate that most participants reported experiencing moderate to extreme-severe levels of depression, anxiety and stress symptoms (Lovibond & Lovibond, 1995). Results also indicate that all participants reported at least moderate levels of obsessive-compulsive symptoms (Abramovitch et al., 2020). Further, scores on the SDS suggest that all participants except for Participant B were likely to

respond to questions posed in the self-report questionnaires and interview in a socially desirable way.

Table 5. Demographics and Self-Report Measure Scores for Each Participant

	Participants			
	Participant A	Participant B	Participant C	Participant D
Aggression Measures				
ARS-Revenge	15	16	9	12
LHA-Aggression	21	25	8	21
MCAA-Violence	24	24	18	17
Clinical Symptoms				
Total DASS	52	62	86	38
DASS-Depression	14 (moderate)	20 (moderate)	16 (moderate)	6 (normal)
DASS-Anxiety	14 (moderate)	10 (moderate)	36 (extremely severe)	20 (extremely severe
DASS-Stress	24 (moderate)	32 (severe)	34 (extremely severe)	12 (normal)
OCI	40 (severe)	31 (severe)	35 (severe)	16 (moderate)
Other				
Total OBQ	83	89	94	92
OBQ-T	28	24	21	23
OBQ-R	14	17	27	27
OBQ-I/C	12	15	23	19
OBQ-P/U	29	33	23	23
EDQ	147	129	44	63
TCQ	26	25	38	38
FSQ	36	23	43	43
SAM	29	31	55	63
SDS	8	2	10	10
Age	49	35	39	23
Ethnicity	Australian	Australian	Australian	Australian

Note. ARS - Revenge = Anger Rumination Scale – Thoughts of Revenge subscale; DASS = Depression Anxiety Stress Scale; EDQ = Ego Dystonicity

Questionnaire; FSQ = Fear of Self Questionnaire; LHA-Aggression = Life History of Aggression – Aggression subscale; MCAA - Violence = Measure of

Criminal Attitudes and Associates- Violence subscale; OBQ = Obsessive Beliefs Questionnaire; OBQ- T = Obsessive Beliefs Questionnaire – Threat subscale;

OBQ- R = Obsessive Beliefs Questionnaire – Responsibility subscale; OBQ- I/C = Obsessive Beliefs Questionnaire – Importance and Control of Thought

subscale; OBQ- P/U = Obsessive Beliefs Questionnaire – Perfectionism and Uncertainty subscale; OCI-R = Obsessive Compulsive Inventory-Revised; SAM

= Self Ambivalence Measure; SDS = Social Desirability Scale; SIV = Schedule of Imagined Violence - Frequency; TCQ = Thought Control Questionnaire;

QUIT = Questionnaire of Unpleasant Intrusive Thoughts - Frequency

Analysis of the total scores from the self-report measures indicated that all participants reported engaging in anger rumination, held violence supportive beliefs, and had a history of aggression. Psychometric data also revealed that all participants experienced, to varying extents, maladaptive obsessive beliefs and beliefs concerning the self. Further, participants A and B's psychometric scores on the EDQ indicated that they experienced elements of ego-dystonicity with regards to their aggressive thought experiences.

Qualitative findings

Five major themes derived from the semi-structured interviews and illustrative participant descriptions are presented below. During the interviews and data analysis, difficulties arose when attempting to differentiate whether respondents were describing AITs or aggressive scripts. Therefore, no assumptions have been made with regards to which construct participants were describing, and this is justified on the basis that both positive and negative emotions were identified when rehearsing aggressive thoughts. Results from the thematic analysis revealed that the rehearsal of aggressive thoughts is a complex process, involving many facets, and can precipitate both positive and negative experiences for those who experience them.

Theme 1: Precipitants. Most participants said that the precipitants to their aggressive thinking was related to being provoked by another person or being exposed to certain objects (e.g., such as a knife). The type of provoking situation was idiosyncratic, and each participant described unique precipitants (i.e., road rage). Most participants reported that their experience of trauma and abuse were likely the cause of their propensity to rehearse aggressive thoughts.

Most participants noted that their aggressive thought rehearsal was dependent on the mood they were in, and that negative affect most often induced this type of thinking. Some participants acknowledged that experiencing anger would often perpetuate the aggressive thinking.

Sometimes anger, anger will help spur it along. (Participant B)

I'm basically put under a lot of stress, and that causes me to go into that thinking mode. (Participant D)

Theme 2: Negative impacts. Participants described several negative impacts associated with the rehearsal of aggressive thoughts. The main impact noted was that the aggressive thought impacted a person's attention and concentration, as they would become

fixated on the aggressive thought. Participants also described that the thoughts may often appear frequently and spontaneously, and that this process can be disruptive and impact on any activity at hand. Some participants described that the thoughts would become "all consuming" and that it was experienced as "invasive and intrusive". Several participants noted that the graphic content of the aggressive thoughts was what drew their attention to the thought, and this was described by most participants as being a negative experience. One participant noted that the aggressive thought was significantly distracting to them, impacting their interaction in social settings.

Nothing else is going in my head. It's just constant thinking about what I could do to the individual or object or whatever it might be. It's all consuming.

(Participant A)

Through exploring the subjective experiences associated with aggressive thinking, it was identified that some participants reported both positive and negative experiences. These experiences are detailed below across two themes.

Theme 3: Negative experiences. Most participants described the emotional impact of aggressive thinking as unpleasant, unwanted, and/or distressing. Whether participants considered their aggressive thought as unpleasant, unwanted, and/or distressing was dependent on the content, and whether this content was salient to the individual. Several participants acknowledged that aggressive thoughts that were related to people they cared about, were experienced as very distressing. Additionally, some participants noted that the distressing aspect of the thought was centred on how they were perceived by others.

If it's someone who I really love, dear, and care for I get very distressed.

(Participant A)

Probably the fact that I never wanted to be an angry person yet, unfortunately,

I've always been an angry person and I'm very well known for it unfortunately...The violence isn't what I want you know, it shouldn't be needed sort of thing. And even though I've had some very aggressive thoughts over a while now, I've never wanted them. (Participant D)

The concept of ego-dystonicity appeared from participants' description of their emotional experiences associated with their aggressive thoughts. Similar to how certain thoughts were considered distressing, whether the aggressive thought experience was ego-dystonic to the individual was highly subjective. This was dependent on their level of discomfort associated with the thought content, and their interpretation of the thought.

Distressed is probably the best one. So as I'm heading up, adrenaline excited, and if it's someone that I care about I get very distressed afterwards. (Participant A)

It's not normal for a normal person to think that they can, they want to break someone's arm or rip someone's head off especially if it's over something petty.

(Participant C)

All participants acknowledged that the potential outcomes associated with enacting their aggressive thoughts was something they were highly aware of and were keen to avoid. Most participants noted that going to jail or having a criminal record was a negative outcome they did not wish to experience.

Well, that's the part that keeps me out of jail...I value my freedom too much.

(Participant B)

I don't want to go to jail. That's a life ruiner you know...if I got a criminal record, I would not be allowed to work...(Participant D)

A subtheme of negative experiences was centred around negative self-perceptions resulting from their aggressive thought experiences.

I've always considered myself to be damaged goods (Participant A)

Yeah I am [fearful of what the thoughts reflects about me] and like it really scares me that I am...It just important for me going forward in life because if I'm a bad person, then I deserve what I get. (Participant C)

I've always been known as a reactive angry person. (Participant D)

Some participants acknowledged recognising that their aggressive thinking was a source of them feeling like they were different to others or people they knew.

I've always been different, different amongst my friendship groups different amongst my family. (Participant A)

I mean, I've always wondered why I fought and acted differently like my whole life.

(Participant B)

Theme 4: Positive experiences. Some participants noted that the rehearsal of aggressive thinking was associated with pleasant emotions including feeling amused, experiencing enjoyment, and excitement, with two participants describing aggressive thinking generating a positive and energising physiological response, an "adrenaline rush". One participant described that the pleasant experience associated with aggressive thinking was influenced by the physiological sensations in their body which precipitated sexual arousal.

I get super excited and it's like an absolute adrenaline rush (Participant A)
...It's normally a sensation of, of feeling quite good. (Participant B)
...If I'm totally honest, I sort of felt on top of the world in my own little world
(Participant D)

Most participants recounted deliberately engaging with or elaborating on their aggressive thinking, suggesting an active role in this process.

Because what's going on in my head is just going over and I'm refining, you know,

I'll go for more, I'll pick up that steak knife and jam it in their throat to going oh no I

could have more fun doing it this way or that way. (Participant A)

...I'll run through a scenario in my mind where it'll be like, so it's not just like a still picture, but it's almost like watching a move where I'll see it all unfold but like everything, like it could be the whole situation...running through my head frame by frame. (Participant B)

So I was sort of fuelling my own little fantasy with violent thoughts. (Participant D)

Nearly all participants described that their aggressive thinking, which was deemed a pleasant experience, reflected consistency with their sense of conduct, history of violence, attitudes, how they perceived themselves, or content that they enjoyed thinking about.

If it's something that I don't, someone or something that I don't care about I get super excited absolute adrenaline rush. (Participant A)

...mainly you know, people I deem as being unworthy or trash basically, which I kind of sometime lump everyone into that category...I suppose things that have helped mould and shape my mind in different ways...Oh [aggressive thoughts] don't bother me, as long as I can understand them. (Participant B)

...the aggressive thoughts, definitely are something that was a part of who I was in the last few years. (Participant D)

A participant described that their aggressive thinking served a function of providing a place to imagine using violence as a solution to their problems.

...wanting to fix the problems myself with violence...because I imagined it in my mind, you know, figuring out the problem with violence would clear up the problem and therefore not exist...yeah the problem is solved. (Participant D)

Theme 5: Management. Participants described several ways they managed their aggressive thoughts, with the most common involving some form of experiential avoidance. By deliberately avoiding aggressive thoughts, participants described feeling that they were engaging in "healthier" ways of thinking and thereby reducing the likelihood of being in risky situations.

I can't get rid of the thoughts unless I've got other coping mechanisms around me.

(Participant A)

If I'm not around knives, it's one less way or avenue for me to get into trouble...

(Participant B)

...submerge myself into escape basically, that is my form of escapism. (Participant D)

Additionally, some participants described the benefit of engaging in cognitive restructuring, to help modify the way they experienced and interacted with their thoughts.

I'm just trying to come up with healthier ways basically to, to let the thought come and go and to think of something else. (Participant B)

Whereas now it's, you know ... now it doesn't really, it's not something that I hang on to because if I don't let it go, then it eats me up. (Participant D)

All participants described that receiving support from others, including family, loved ones, or professionals, helped manage their aggressive thinking. Some participants acknowledged setting limits to their thinking, where they were aware of certain people or experiences that they did not want their aggressive thinking contents to revolve around, and thus attempts to manage the thought were made.

Discussion

The aim of the study was to examine the subjective experiences associated with aggressive thinking in a sample of community forensic mental health clients who had a history of violent offending or who were considered by treating professionals to have

problems with the experience or regulation of anger. Findings highlight that both positive and negative experiences are associated with aggressive thinking, the negative impacts that develop as the thoughts are experienced, as well as the types of circumstances that precipitate the thoughts and what people sometimes do to manage negative experiences and impacts.

Experiences Associated with Aggressive Thinking

Negative Impacts

This study revealed that aggressive thinking may include several negative impacts related to the disruption of functioning, intrusiveness, frequency of thought, and level of distraction associated with the thought. This is consistent with Patel (2015) and Hosie, Dunne, et al. (2022) where the experience of aggressive thinking was sometimes associated with feelings of intrusiveness, and that participants are often seeking to manage their thoughts, seemingly to avoid acting on them aggressively, but also because they sometimes dislike the self-perception that experiencing these thoughts reflects poorly on them, and that experiencing these thoughts means they are a 'bad person'. These experiences are not all that different to those reported by individuals with AITs in OCD, where the frequent and intrusive thoughts precipitate thought control strategies and these thoughts are often associated with negative self-appraisals (Belloch et al., 2007; Moulding, Aardema, et al., 2014a). These findings suggest that the negative impacts associated with AITs in OCD, and aggressive thoughts experienced by individuals with a history of aggressive behaviour, are similar and thus may not prove as useful features for differentiation between these phenomena.

Thought Management

The use of management strategies to control aggressive thoughts or to distract oneself from their thoughts was highlighted in the current study, and findings revealed that the form of strategies employed to manage aggressive thinking was different for each participant.

Consistent with prior findings, the current study highlights the use of methods such as experiential avoidance, cognitive restructuring, and social coping (i.e., seeking support from others such as loved ones or professionals) to manage aggressive thinking (Nagtegaal et al., 2006; Patel, 2015). Findings from Fernandez et al. (2022) highlight that the use of thought control strategies was negatively associated with aggressive script rehearsal, suggesting that these methods may be useful in reducing the frequency of these thoughts. Whether the methods identified in the current study, such as experiential avoidance, cognitive restructuring, and distraction are effective in the long-term management of aggressive thoughts requires further examination, as previous studies have questioned the long-term benefit of thought control (Nagtegaal et al., 2006; Stokes et al., 2022). Within the context of OCD, research findings examining the use of thought suppression and thought control for the management of intrusive thoughts are mixed, where the use of certain methods, such as thought suppression, have been found to exacerbate obsessive compulsive symptoms (Belloch et al., 2004; Purdon & Clark, 2001; Wegner et al., 1987). However, research has also identified that methods such as cognitive restructuring may be effective in reducing obsessive compulsive symptoms (Clark, 2004; Shingler, 2009). The current study's findings highlighted the potential benefit of using certain thought control strategies in the management of aggressive thinking, however the long-term efficacy of these methods requires further exploration.

Precipitants of Thoughts

The current study identified various precipitants to aggressive thinking and consistent with the Multiple Systems Model developed by Denson (2013), perservative thinking of anger or provoking situations appear to intensify and prolong one's engagement with their aggressive thoughts. In this study respondents reported that they were likely to engage in aggressive thinking as a means of emotional regulation, or that the thinking was triggered by

a certain emotional state (e.g., anger or frustration). This finding aligns with previous research by Hosie, Dunne, et al. (2022), where engagement with aggressive thinking has been proposed to regulate negative affect, helping individuals alleviate an unpleasant emotional state or provide rehearsal of plans for retaliation to provocation. Overall, these results suggest that although the precipitants to aggressive thinking may involve an emotional component, specific precipitants for aggressive thinking remain idiosyncratic and likely reflect subjective situations that induce aggressive or anger states in the individual. Within the context of OCD, repugnant thoughts such as AITs are not used as a means of emotional regulation (Veale et al., 2009), their occurrence is said to be attributed to internal processes, and identifiable precipitants are less clearly understood (Lee & Kwon, 2003). These findings suggest that the precipitants and rehearsal of AITs and aggressive thoughts may be features useful in differentiating between these phenomena, however, further exploration of these features is required.

Negative and Positive Experiences

Our findings suggest an association between aggressive thinking and negative experiences, namely features pertinent to unpleasantness and unwantedness, distress, concern regarding consequences, and some elements of ego-dystonicity. The results support previous findings which have demonstrated that aggressive thinking in individuals with a history of aggression can be experienced as unwanted and disturbing (Patel, 2015). An unexpected finding was that aggressive thinking may be associated with elements of ego-dystonicity (i.e., experienced as inconsistent with one's self-view) in a forensic population. Extending the findings from Fernandez et al. (2022), the current study demonstrates that ego-dystonicity may relate to aggressive thinking only in specific contexts. Although participants had a history of violence or problems with anger, whether their aggressive thought experiences were deemed ego-dystonic was dependent on the thought content type (e.g., if it concerned

loved ones or people they cared for), how they interpreted the thought (e.g., believing they are not normal for thinking aggressively, or perhaps believing the person was deserving of provocation), and the emotional reaction associated. Research on ego-dystonicity in forensic samples is limited, and therefore the literature explaining these findings is limited to studies conducted amongst OCD populations. Purdon and Clark (1999) propose that whether a thought is deemed to be ego-dystonic is dependent on how salient the thought is to the individual, and this may concern areas related to though content, appraisal, and emotional reactions. Therefore, even in samples prone to violence, aggressive thoughts may be experienced as ego-dystonic to the individual depending on thought content, and thus may be associated with different levels of violence risk. Further exploration of ego-dystonicity in the context of aggressive thinking in forensic samples may elucidate the role this concept has on influencing or limiting aggressive behaviour over time, and it may explain desistance processes in individuals who have a history of violence but who are now committed to nonviolence. Additionally, the current study identified that certain negative self-perceptions were described in relation to aggressive thought experiences. Our findings reflect that individuals may engage in negative self-appraisals when experiencing aggressive thoughts, including seeing themselves as different to others, or being fearful of what their thinking reflects about them. These findings are similar to what is observed in individuals with OCD who report AITs, where negative self-perceptions develop from the experience of ego-dystonic thoughts. However, whether these features influence levels of violence risk remains unclear, as research exploring self-perception and aggressive thoughts, in individuals with a history of violence is limited.

Positive experiences were also identified as a feature associated with aggressive thinking for some participants. Our findings suggest that the engagement with aggressive thinking may be related to elements of pleasant emotional experiences, ego-syntonicity, and

deliberate elaboration and refinement of aggressive scripts. These findings are consistent with prior research which suggest that there are different emotional reactions to aggressive thinking (Hosie et al., 2021), and that some individuals may experience their aggressive thoughts as pleasurable and exciting (Patel, 2015). These findings also present the possibility that individuals who experience their aggressive thinking with positive emotions are likely to interpret them as aligning with their sense of self, attitudes, and behaviour. This is consistent with prior findings that suggest aggressive thinking is influenced by one's attitudes towards violence, and history of violent behaviour (Gilbert & Daffern, 2017; Gilbert et al., 2013). Additionally, the current study's findings suggest that one's engagement with aggressive thinking may also provide an avenue for imagination whereby a theme of rehearing aggressive thoughts to provide a solution to a problem was identified. This has been elucidated by Gilbert and Daffern (2017) particularly when differentiating between conceptually related constructs such as aggressive scripts and aggressive fantasies. It is likely that in the current study the exploration of aggressive thinking associated with positive affect has revealed connections with aggressive fantasies, which are less focused on planning and preparation and are likely to serve an emotional regulation function (Gilbert & Daffern, 2017). The positive emotional experiences associated with aggressive thinking identified in the current study contrasts to understanding of AITs in OCD where it is emphasised within OCD literature that the experience of intrusive thoughts are highly ego-dystonic and associated with negative affect, including distress and fear (Veale et al., 2009). The current study may highlight the importance of considering an individual's subjective experience of their aggressive thought, including their emotional reaction to the thought and how this may impact on the risk of violence.

Strengths, Limitations and Future Research Directions

The current study presents with several strengths, including being one of the first studies to investigate the subjective experiences associated with aggressive thinking. The present study is one of the first to examine aggressive thinking and its relationship with OCD-related concepts, with the aim to identify relevant features associated with the rehearsal of aggressive thinking. However, the current study's findings should be considered in light of certain limitations. It is acknowledged that a small sample of participants were used, and given that the focus of qualitative data analyses is on theme identification and development, the authors were not concerned with theme saturation. Current issues with measurement instruments of aggressive thinking phenomena (i.e., AITs and aggressive scripts) made it difficult to differentiate what constructs respondents were discussing, especially if both phenomena were present. This limitation highlights the current lack of clear conceptualisations of these constructs, and how these phenomena differ. Further, the quantitative data from this study suggest that the aggressive thinking experiences were common in a group with moderate to severe clinical symptoms, and thus further research may consider investigating these experiences in other populations to develop a richer understanding of the experiences and impacts of aggressive thinking.

Conclusion

The present study has demonstrated that a range of features (e.g., intrusiveness, frequency, disruption of functioning), and both positive and negative emotional experiences may be associated with aggressive thinking in a forensic sample. The present study's findings have highlighted that aggressive thinking can be largely idiosyncratic, precipitated by external events or used as a means of emotional regulation. The present study identified that elements of ego-dystonicity and ego-syntonicity may be related to aggressive thinking, but are dependent on thought content. Further, the likelihood of aggressive behaviour can be

attributed to the extent one holds violence supportive beliefs and has a history of aggressive behaviour. The role ego-dystonicity and ego-syntonicity play in influencing aggressive behaviour is still largely unknown, however, the present study has demonstrated that ego-dystonic aggressive thoughts are likely associated with unpleasant and negative emotions, which may influence violence risk levels. The current study's findings have implications for the assessment and treatment of aggressive thoughts, emphasising the need for clinical assessments to explore a range of features associated with one's thoughts. The current study has demonstrated that aggressive thoughts can be experienced both positively and negatively, and without further exploration of thought content, thought precipitants, emotional experience, and thought engagement and intention, little will be discerned regarding the likelihood of these thoughts becoming problematic or influencing aggressive behaviour.

PART III – GEN	NERAL DISC	USSION AND	CONCLUSION	Γ

CHAPTER 9 – INTEGRATED DISCUSSION

Experiencing thoughts about harming another person are said to be a universal phenomena (Rowa & Purdon, 2003), however, they are also a common symptom of various mental health disorders, including OCD (i.e., in the form of AITs), and have been consistently reported in forensic populations as cognitions that increase propensity for aggressive behaviour (i.e., in the form of aggressive scripts; Hosie et al., 2021; Moulding, Aardema, et al., 2014a). Although the outcomes associated with AITs and aggressive scripts are thought to be distinct, no empirical investigations have been conducted concurrently to differentiate these phenomena, including identifying features that may be unique to each construct. This thesis aimed to address these gaps in the literature by conducting a critical review exploring what features, already established in OCD literature, may overlap with aggressive script rehearsal, as well as conducting two empirical studies focusing on the experiences of AITs and aggressive scripts, and the role maladaptive beliefs play in their occurrence. This chapter provides an integrated discussion that draws the results of the critical review and the two empirical studies of the thesis. The purpose of this chapter is to condense the main findings of the research and discuss them in light of the overall thesis aims and within the context of potential implications for clinical practice. Given that the results of each empirical study have been discussed extensively in previous chapters, this discussion will only focus on the main findings.

The present research advances understanding of the differentiating features of AITs and aggressive script rehearsal. The critical review identified that some of the features used in the characterisation of intrusive thoughts may be common to the experience of aggressive script rehearsal. Specifically, similar to AITs, aggressive scripts may be experienced as frequent and recurrent, and thought control strategies may be employed to manage aggressive scripts when then come to mind. However, the review also indicated that there is less

conclusive evidence for similarities in the emotional response to these phenomena and perceptions of their intrusiveness (e.g., whether they are intrusive or unwanted). Empirical investigation of AITs and aggressive script rehearsal in empirical paper one confirmed relevant theoretical underpinnings of the phenomena, including that AITs are not related to aggressive behaviour or violence supportive beliefs where aggressive script rehearsal is. Additionally, AITs were found to relate significantly with ego-dystonicity; however the role ego-dystonicity plays within aggressive script rehearsal remained unclear, as evidenced by reports of participants in study two. Here, the subjective experiences associated with aggressive thoughts were explored in a forensic sample. Several themes relevant to aggressive thought rehearsal were identified including how aggressive thoughts are associated with both positive and negative experiences, thereby countering suggestions that aggressive thinking is consistently and universally accepted and experienced positively within forensic populations.

This chapter also discusses methodological issues identified through the empirical investigations, as well as the limitations of the current research. This chapter concludes with an exploration of the implications of the research, including considerations for risk assessment and treatment of these phenomena. Future research directions are also discussed.

9.1 The Features of AITs and Aggressive Scripts

This research aimed to explore the phenomenology of AITs and aggressive scripts, and to identify differentiating features between these constructs to improve our understanding of these thought phenomena. Differentiation might be particularly relevant to violence risk assessment. The research identified similarities between AITs and aggressive scripts including the content of the thought, whether they are experienced as intrusive, spontaneous, are associated with feelings of unpleasantness and distress, and if thought control strategies are employed to manage these thoughts. Results suggests that differentiation between AITs

and aggressive scripts may occur with regards to how the thoughts are appraised, including if they are experienced as ego-dystonic or ego-syntonic, and relate to one's history and attitudes, including whether they hold violence supportive beliefs and if they have a history of violence behaviour. Results also highlighted how these features may influence the experience of an aggressive thought and how one engages with these thoughts subsequently. Specifically, Table six provides a summary of the similarities and differences between AITs and aggressive scripts to aid in the conceptual clarification.

Table 6. List of features and their presence in the experience of AITs or Aggressive Scripts

Features	AITs	Aggressive Scripts
Aggressive content theme, including harming loved ones	Y	Y
History of violence and aggression	N	Y
Violence supportive attitudes	N	Y
Frequent and recurrent	Y	Y
Intrusive	Y	Y
Spontaneous	Y	Y
Deliberately rehearsed	N	Y
Deliberately generated	N	Y
Disrupts functioning	Y	Y
Negative emotional experiences (e.g., distress and discomfort)	Y	Y
Positive emotional experiences (e.g., excitement, pleasure)	N	Y
Ego-dystonic (i.e., inconsistent with sense of self)	Y	?
Ego-syntonic (i.e., consistent with sense of self)	N	?
Obsessive beliefs	Y	Y
Thought control and neutralising behaviours	Y	Y

Note. AITs = Aggressive Intrusive Thoughts; Y = feature is present; N = feature is not present; ? = features' relevance to construct requires more exploration

9.1.1 Similarities in Phenomenology of AITs and Aggressive Scripts

The content of AITs and aggressive scripts both concern thoughts about harming another person, which for both can be related to loved ones or unrelated others. Consistent with the cognitive behavioural model of OCD, findings demonstrated that the consequences associated with the experience of AITs are dependent on how the thoughts are appraised by the individual, and this appraisal process is influenced by maladapive beliefs. Further, the results from this thesis demonstrate that individuals with a history of violent behaviour can experience aggressive thoughts related to people they care for, and that the negative emotional reactions to these thoughts (e.g., distress, discomfort, unwanted) can be experienced similarly to that which has been reported in individuals with AITs in OCD. These findings are consistent with prior research that has examined the emotional sequalae associated with aggressive script rehearsal, where there is some evidence to suggest that aggressive thoughts experienced by individuals with a history of violent behaviour can be experienced as ego-dystonic, be associated with negative emotions (e.g., sadness, distress, fear, disgust; Hosie et al., 2021; Patel, 2015), and can impact general well-being (Poon & Wong, 2021). However, there is also evidence that some aggressive scripts can be regarded positively, depending on the thought content, as evidence by the results of study two and consistent also with Hosie and colleagues (2021).

Given there has been few empirical explorations of aggressive scripts, our previous understanding of the experiences associated with aggressive scripts was limited. Results from this thesis demonstrate that aggressive scripts can be associated with some of the array of features pertinent to AITs in OCD, including that they are experienced as frequent and recurrent, and are associated with thought control strategies. The critical review identified that it was unclear whether aggressive script rehearsal was related to features of unwantedness and intrusiveness. However, further research from this thesis identified that

aggressive script rehearsal may be related to the features of intrusiveness, spontaneity, unwantedness, and disruption of functioning. The results suggest that aggressive thoughts, operationalised as aggressive script rehearsal, can indeed be experienced by individuals with a history of violence as intrusive, disruptive, and distressing, but these features are dependent on the aggressive thought content type (e.g., related to loved ones or people they care for). This is consistent with phenomenological explanations of AITs in OCD (Moulding, Aardema, et al., 2014a), where the appraisal of the thought experience is an important factor in determining the clinical significance of the symptom. Based on descriptions provided by respondents in empirical study two, it appears that when an aggressive thought is experienced as unpleasant and distressing, it is not likely associated with intent towards or elaborations of aggressive behaviour, suggesting a potential interaction with ego-dystonicity. These results may also suggest that individuals with a history of violence may experience AITs and aggressive scripts concurrently and that simply because somebody with a history of violence experiences an aggressive though does not mean that thought is welcomed or experienced pleasurably, however limits to current measurements and operational defintions of these constructs influence adequate differentiation. Current models of aggression that include consideration of script rehearsal such as the GAM do not give much consideration of how one's interpretation of an aggressive script, such as whether the thought is inconsistent with one's sense of self (i.e., ego-dystonic) or not, may influence the experience of the thought and behavioural outcomes. The findings from this thesis elaborate aggressive scripts and have important theoretical implications, suggesting that the specific content of the aggressive thought, and differential features such as pleasant or unpleasant emotions, ego-dystonicity, attitudes supporting violence, and a history of violence may prove important.

Another feature of similarity between AITs and aggressive script rehearsal concerns the use of thought control strategies to manage aggressive thought experiences. Consistent with prior research (Jacoby et al., 2015), these results indicate that certain thought control strategies, such as punishment and worry, were associated with greater OC symptoms. This is in accordance with cognitive behavioural models of OCD that question the efficacy of compulsive, neutralising, and avoidance strategies in OCD (Clark, 2004; Salkovskis, 1989). While these strategies and mechanisms of control may alleviate the distress caused by the intrusive thought in the short term, their use may be detrimental in maintaining intrusive thought experiences over time (Ahern et al., 2015). With regards to aggressive scripts, this thesis identified that the use of thought control strategies, either in the form of direct distraction (e.g., listening to music, going for a drive) or experiential avoidance (e.g., removing oneself from or avoiding situations that elicit aggressive thoughts) were associated with reductions in aggressive thinking. Consistent with prior research on aggressive thoughts (Nagtegaal et al., 2006), the results suggest that the use of distraction or experiential avoidance may reduce one's engagement with their aggressive thinking, and limit the likelihood of aggressive behaviour. Further, although thought control strategies are used in both the experience of AITs and aggressive script rehearsal (Jacoby et al., 2015; Nagtegaal et al., 2006), it is not yet clearly understood whether avoidance strategies for aggressive scripts have the same negative long-term effects (i.e., increase thoughts over time) as they do with AITs in OCD. Further research is required to acertain the efficacy of thought control and avoidance strategies for aggressive scripts, and the influence these strategies may have in preventing aggressive behaviour.

9.1.2 Differences in Phenomenology of AITs and Aggressive Scripts

It has been reported that AITs in OCD differ from other thought phenomena because of the way they are experienced as unwanted, and distressing to the individual and are inconsistent with their sense of self and conduct (Clark, 2005; Moulding, Aardema, et al., 2014a; Purdon et al., 2007; Rowa & Purdon, 2003). Based on the findings from this thesis,

some of these features, including feelings of distress and unwantedness, may also be relevant to aggressive script rehearsal in individuals with a history of violence and thus cannot be relied upon as factors for differentiation the phenomena. The critical review into the features of AITs and aggressive scripts revealed that aggressive scripts may differ from AITs in the way that they are deliberately rehearsed, and how one's life history of aggression plays a significant role in influencing the rehearsal of the scripts.

It has long been held that the AITs in OCD are not associated with acts of aggression, and that the appearance of AITs in OCD should not be considered risk indicators for violence due to their ego-dystonic nature (Veale et al., 2009). As expected, AITs were not associated with violence supportive attitudes and a history of aggressive behaviour. These findings are consistent with cognitive models of OCD (Rachman, 1997; Salkovskis, 1985) that emphasise the ego-dystonic and abhorrent nature of intrusive thoughts in OCD. These cognitive models stipulate that the experience of intrusive thoughts that are inconsistent with the self are appraised as abhorrent to the self, influencing feelings of distress and attempts to control or prevent the perceived consequences associated with the thought (Rachman, 1997, 1998). Further, increased frequency of AITs are not associated with violence risk, whereas in forensic contexts, the frequency of aggressive scripts is commonly used as a risk indicator (Grisso et al., 2000).

Consistent with prior research (Hosie, Dunne, et al., 2022; Kelty et al., 2011), aggressive script rehearsal was associated with anger rumination, violence supportive beliefs, and a history of aggressive behaviour. This is consistent with understandings of script rehearsal, where according to Script Theory (Huesmann, 1998) aggressive scripts may be strengthened through the influence of violence supportive beliefs and behaviours that condone violence. Both empirical studies of this thesis identified that violence supportive attitudes, such as believing that violence should be tolerated and used in situations of

provocation, were associated with aggressive script rehearsal. Further, findings from the qualitative study suggest that engaging in rumination on past provocations and experiences of anger, likely perpetuates the rehearsal of aggressive thoughts. These findings align with the Multiple Systems Model (Denson, 2013) of anger rumination whereby perseverative thinking of anger inducing regarding events or past provocations may influence feelings of anger and thoughts of revenge (Sukhodolsky et al., 2001). Overall, these findings support the importance of considering one's attitudes towards violence and history of aggression in the clinical assessment of aggressive thoughts, as they have been reliably identified as relevant features of aggressive script rehearsal (Hosie, Simpson, et al., 2022). Further, understanding how the experience of aggressive thoughts align with one's sense of conduct and previous behaviours may likely prove beneficial in determining one's risk for aggressive behaviour in future.

The emotional reactions associated with AITs in OCD have been consistently identified as negative in nature, including feelings of distress, shame, disgust, and fear (Moulding, Aardema, et al., 2014a; Rowa & Purdon, 2003). Limited hypotheses could be made regarding the subjective experiences associated with aggressive script rehearsal as empirical investigations have been scarce. Understanding the emotional experiences associated with aggressive script rehearsal may help inform assessments of these phenomena and identify features related to aggressive behaviour. In line with prior research (Hosie et al., 2021; Hosie, Dunne, et al., 2022; Patel, 2015), aggressive thoughts reported by individuals with a violent history can be associated with negative experiences, and the emotional reactions to the thoughts can be diverse (Hosie, Dunne, et al., 2022). While the results indicated that individuals with a history of aggressive behaviour or anger problems may report that their aggressive thoughts are negative experiences and are associated with negative affect (e.g., distress, unpleasantness), the same individuals also reported pleasant

emotional experiences with other aggressive thought types. These findings suggest that one's emotional experience associated with aggressive thoughts is dependent on the thought content, is likely idiosyncratic, and may be influenced by the degree to which individuals regard their aggressive thought as ego-dystonic or ego-syntonic. These results also suggest that the relationship between aggressive script rehearsal and emotional experiences is complex, and may not follow a linear trend.

Cognitive models of OCD posits that intrusions are consistently reported as egodystonic and experienced as inconsistent with an individual's sense of self and previous behaviour (Purdon et al., 2007; Rachman, 1997; Salkovskis, 1985). Consistent with these cognitive models of OCD, empirical study one highlighted the possibility that egodystonicity may be used as a differentiating factor between AITs and aggressive script rehearsal. However, with further exploration of these constructs, empirical study two identified that while some aggressive thoughts were experienced as ego-dystonic to some participants, elements of ego-syntonicity (i.e., consistent with one's sense of self, beliefs, and previous behaviours) were also described by some participants in relation to their aggressive thought experiences. It was evident by participants descriptions in empirical study two that the experience of ego-syntonic aggressive thoughts were deliberately engaged with, welcomed, and sought after. This is consistent with conceputalisations of aggressive thinking within Script Theory and the GAM whereby the frequent rehearsal of aggressive scripts is likely to be accommodated and normalised by the individual with little resistance (Huesmann, 1988). These findings suggest that it may be likely that ego-syntonic aggressive thoughts that are deliberately engaged with and align with one's sense of conduct and previous behaviours, may be an important indicator for future risk behaviours. There is also some indication from these findings that aggressive scripts may be distinguished from AITs on the basis of ego-syntonicity and positive emotional responses, as these features are not

common within intrusive thought phenomenology. Considering this, it is important to highlight that a person may experience both aggressive scripts and AITs, however, it is the characterisation of these thoughts in relation to the individual's sense of self that may prove as an important indicator of differentiation between these two constructs. How one's sense of self is implicated through the experience of aggressive thoughts may influence the way it is interpreted and elaborated on. Further, it remains unclear whether the experience of egodystonicity follows a transient process in individuals with a history of violence. As posited by Purdon et al. (2007), when repugnant thoughts in OCD are repeatedly experienced, egodystonic thoughts may become more ego-syntonic as they are accommodated into a person's sense of self. It is important to note that the measurement of ego-dystonic aggressive thoughts in individuals with a history of violence should consider if one believes their thought is inconsistent with their sense of self, but also if their aggressive thought is inconsistent with one's prior experiences, behaviours, and conduct. Further, it is worth noting that the operationalisation of ego-dystonicity and ego-synotnicity should consider these constructs on a spectrum, where some thoughts may be percieved as more ego-dystonic/ego-synotinic than others. Neverthless, the relationship between ego-syntonicity and aggressive thoughts in those with a history of aggressive behaviour warrants further empirical investigation.

Further, the occurrence of AITs has been suggested to be internally generated without an identifiable source (see: autogenous obsessions; Lee & Kwon, 2003). Contrastingly, aggressive scripts have been described as often being deliberately generated to have a planning function (Gilbert & Daffern, 2017) or to assist in the regulation of emotion (Hosie, Dunne, et al., 2022). Some participants described that their aggressive thoughts were deliberately generated and elaborated, and that they were used as a form of escapism to imagine solutions for their problems. Aggressive thoughts were described as easily triggered, in particular by mood, and perpetuated by anger. Consistent with Script Theory (Huesmann,

1988, 1998) and the Multiple Systems Model (Denson, 2013), the rehearsal of aggressive scripts was associated with anger rumination, and respondents with a violent history described that their aggressive thoughts were likely precipitated and motivated by feelings of anger. These results support current formulations of aggressive script rehearsal (Gilbert & Daffern, 2017), and highlight the importance of considering one's mood and rumination behaviours when exploring aggressive thought features.

9.2 Limitations

9.2.1 Sample

While this thesis sampled participants from two diverse population groups, a non-clinical sample and a sample of individuals with a history of violence, generalisability of results may be limited. Empirical study one utilised a reliable methodology, including relevant measures of phenomenology, however, the study was conducted in a non-clinical sample so replication with a clinical sample of participants (people with OCD and people with a history of violence) is necessary to confirm the findings as they pertain to OCD symptoms and aggressive script rehearsal.

Empirical study two is limited by the small male forensic sample, and while results may be limited in generalisability, the qualitative aspect of the study highlights the importance of using certain questions to enquire about one's experience with aggressive thinking (e.g., how do you feel when you experience the aggressive thoughts?). It is therefore recommended that this form of questioning and qualitative investigation be replicated with a larger sample of individuals who have extensive histories of violent behaviour. Finally, empirical study two was unable to compare the experience of aggressive thinking in a forensic sample with individuals diagnosed with OCD who experience AITs. Although there is extant literature in OCD exploring the subjective experiences associated with intrusive thought experiences, it would be useful for a study to replicate the methodology of empirical

study two with an OCD sample to allow for direct comparisons. This comparison would improve the understanding of the features and properties of AITs and how they protect against aggressive behaviour, as well as how they differ from aggressive scripts. Replication of this empirical study and including a sample of participants diagnosed with OCD and who experience AITs is recommended.

9.2.2 Measurement

The measures used in this thesis were selected as they are the only and best known instruments and have been shown to reliably associate with their corresponding construct (e.g., obsessive beliefs relates to AITs; history of aggressive behaviour relates to aggressive script rehearsal). As highlighted in the critical review, the interview quality including the type of questioning, and the respondents understanding of what constitutes the thought being investigated (e.g., being prompted of what an intrusive thought is) are important components that need to be considered when measuring these phenomena. Both empirical studies in this thesis identified similarities in the measurement of AITs and aggressive scripts, whereby the instruments used to measure these constructs question participants on experiencing a thought about harming another person in a similar way. Further, the interviews in empirical study two further highlighted the lack of operational distinction between aggressive scripts and AITs, presenting difficulties in being able to differentiate whether participants were reporting an aggressive script, or an AITs.. While the QUIT and SIV both contain further questioning to gather information about the features related to the construct, the initial question regarding whether the participant experiences a thought about harming another person can be reported as present in both instances, even though AITs and aggressive scripts are believed to be distinct constructs. Further, the QUIT is able to contextualise what constitutes an AIT by providing respondents with a preamble of the characteristics and features of intrusive thoughts more generally. Current versions of the SIV have not yet included a descriptor of

aggressive scripts and this may have implications on respondents identifying the correct cognitive experience. Given that empirical study two and prior research (Hosie et al., 2021; Patel, 2015) have found that aggressive scripts may be associated with negative emotions such as distress, sadness, and shame, whether aggressive scripts and AITs can be differentiated by emotional reactions requires further exploration. Further, differentiation of AITs and aggressive script rehearsal becomes difficult as the current measures do no enquire whether a history of aggressive behaviour or violence supportive beliefs are present. These factors have been reliably found to associate with aggressive script rehearsal (Gilbert et al., 2013; Hosie, Dunne, et al., 2022) and appear to discriminate scripts from AITs, as evidenced by empirical study one's results. By including these factors within a comprehensive measure of aggressive thinking it would provide helpful information to begin distinguishing between AITs and aggressive scripts in an assessment context.

9.3 Implications of Research

The following section explores the implications for future research, including the measurement of AITs and aggressive scripts. It also addresses the implications of the research on the assessment and treatment of aggressive thoughts, including recommendation for the differentiation of AITs from aggressive script rehearsal.

9.3.1 Future Research Directions

The current research provides a basis for understanding the similarities and differences between AITs and aggressive script rehearsal. This research has created a foundation for further empirical examination of these phenomena through the consideration of the GAM, Script Theory, and the cognitive-behavioural model of OCD. Future research is therefore recommended to validate the relationships found and to improve current definition and measurements of AITs and aggressive scripts. The replication of empirical study two

with a sample of individuals with an extensive criminal history, and a comparison sample of individuals diagnosed with OCD who report AITs, will extend the implications of the results and provide clearer understandings of the definitions of these constructs. Further, a measurement tool that includes questions exploring one's history of aggressive behaviour, violence supportive beliefs, and how one interprets their thought and its contents (e.g., egodystonic versus ego-syntonic) would be beneficial for the assessment of aggressive thinking, and to differentiate between AITs and aggressive scripts. The development of guidelines to help to differentiate these similar phenomena and provide diagnostic clarification may also prove beneficial for the assessment and treatment of these aggressive thoughts. Table 6 above may provide a foundation for measurement development. Further research on the experience of ego-dystonic and ego-syntonic thoughts in relation to aggressive thinking is also required. By understanding which aspects of one's self may relate to aggressive thoughts, and how this may impact upon behavioural outcomes, it may refine current understandings of the relationship between aggressive thinking and ego-dystonicity. It is also recommended that future research consider examining the efficacy of certain question types in mearusing

9.3.2 Assessment Implications

The results from this thesis have highlighted several implications for the assessment of aggressive thinking. Results demonstrate that AITs and aggressive scripts may be differentiated by factors relating to a history of aggressive behaviour, violence supportive beliefs, and how one experiences their thoughts (e.g., ego-dystonic versus ego-syntonic). For example, AITs were found to not associate with history of violence or criminal attitudes, and were related to ego-dystonicity. Further, in individuals with a history of aggressive behaviour or problems with anger, it was identified that pleasurable scripts were likely to be experienced as consistent with one's sense of self and conduct (i.e., ego-syntonic), whereas aggressive thoughts experienced as distressing or unwanted were interpreted as being

inconsistent with one's intentions (i.e., ego-dystonic). The following section provides an overview of recommendations for the assessment of aggressive thoughts, including the exploration of specific features important in differentiating between AITs and aggressive scripts. Table 7 provides a list of example assessment questions that may aid in the exploration of relevant features pertinent to AITs and aggressive scripts.

Table 7. Example questions for the assessment of differential features of AITs and aggressive scripts

Feature	Assessment Question/Prompts
Content	What does the aggressive thought involve?
	What aggressive behaviour do you imagine doing?
	• Are the person(s) in the thought known to you, or are they strangers, or an imagined person?
	• What is your relationship to this person(s)?
Process	Do you find yourself purposefully thinking about these thoughts, or do they appear out of the blue?
	• What causes these thoughts to appear in your mind?
	• Do these thoughts occur after certain events or situations, or are they random?
	• Are you able to control the thoughts when they appear? Or do they feel uncontrollable?
	• Are these thoughts frequent? How often do they occur?
Appraisal	What do you think these thoughts mean?
	• What do you think these thoughts say about you?
Emotional sequalae	How do you feel when these thoughts appear in your mind?
	• Do you find these thoughts distressing, or are they pleasurable to you?
	• What part(s) of these thoughts are distressing?
	- Is it the content (i.e., the behaviour, the victims)
	- Is it the process (i.e., occurs randomly; frequently, or is something you cannot stop thinking about)
	- Is it the consequences associated with the thought, or the consequences of acting on the thought (i.e.,
	thought is distracting; acting on the thought could involve criminal consequences)
	• What part(s) of these thoughts are pleasurable?
Ego-dystonicity and ego-	Could you see yourself behaving in the way the thought describes? Why/why not?
syntonicity •	• How does this thought align with your desires or intentions? Describe
	• How does the behaviour in the thought align with who you perceive you are?
	• Consider who you are; is this thought consistent or inconsistent with this sense of self? Describe
Previous history and attitudes	Have you behaved [insert aggressive behaviour described in thought] in the past?
	• What do you think about aggressive behaviour more generally? Is it acceptable or not acceptable to you?
	What are your attitudes towards violence? Do you consider it acceptable? Why/why not?

Firstly, when assessing aggressive thoughts, it is recommended that extensive information is gathered regarding the content of the thought, including potential victims, relationship to victim, and method and severity of aggressive behaviour being thought about.

Current risk assessment instruments do not consider certain features that this thesis has identified as important in differentiating similar phenomena concerning aggressive thoughts. Secondly, exploring the content of the thought with regards to one's appraisal of the thought and the meaning derived by the thoughts occurence is recommended. Thirdly, inquiring about the emotional experience associated with the thought, and how these thoughts fit with one's sense of self and conduct is recommended. Further, in determining the risk this thought poses, gathering information regarding one's history of aggressive behaviour, and their overall attitudes towards violence is essential. The thesis identified that aggressive scripts may be reported as distressing, and ego-dystonic to some degree by individuals with a history of violence. It is therefore essential for assessment of these thoughts to consider what aspects of the thought is distressing to the individual (e.g., the content, the frequent nature of it, or the consequences associated with acting on the thought), and the degree to which the thought is experienced as consistent with one's intentions, desires, or previous behaviour. Further, the rehearsal of aggressive scripts have been found to increase one's propensity towards aggressive behaviour (Daff et al., 2015), and thus form an important risk indicator for violence.

9.3.1 Clinical and Treatment Implications

Although several considerations for the treatment of AITs and aggressive scripts have surfaced from the findings of this thesis, it is important to highlight that the scope of this thesis was to understand the differentiating features between AITs and aggressive scripts. The results of this thesis may be useful in the early detection and understanding of aggressive thoughts predominately in an assessment setting, rather than treatment context. This includes gathering information about the characteristics of one's aggressive thoughts including features related to intrusiveness, frequency, emotional sequelae, ego-dystonicity and ego-syntonicity, attitudes towards violence, and prior violent behaviours. Nevertheless, treatment avenues

should consider the population group these thoughts are identified in, and appropriate models should be applied.

For example, in individuals with an offending history, research suggests that effective intervention depends on the level of consideration given to an individual's needs and risk levels (Andrews et al., 2011). Treatment should focus on understanding the features associated with aggressive scripts in individuals with a violent history, and exploring avenues for reducing the frequency of script rehearsal over time. This may include exploring thought control strategies or the use of behavioural monitoring, such as journaling, to explore potential antecedants of the script rehearsal. Given the variability in the emotional experience associated with aggressive script rehearsal, and the commonality of aggressive scripts in individuals with a violent history, treatment may focus on normalising the experience of aggressive thoughts and understanding one's subjective experience. Treatment for the experience of AITs in OCD should follow cognitive behavioural models of OCD that posit the importance of restructuring maladaptive beliefs that perpetuate obsessive and compulsive behaviours, as well as normalising the experience of unwanted intrusive thoughts (Clark, 2004).

This thesis has highlighted the potential factors that may be considered for differention in AITs and aggressive scripts, including the degree to which the thoughts are experienced as ego-dystonic, and one's history of violence. Thus, treatment should also consider the exploration of how one's aggressive thoughts, both in individuals with OCD and individuals with a violent history, align with their values, intentions, and sense of conduct. In situations where aggressive scripts are regarded as ego-syntonic by the individual, it may be beneficial to explore the prescence of certain of values that could be used to increase the experience of ego-dystonicity towards the aggressive script. Further, in the context of AITs

within OCD, the importance of considering maladaptive appraisals of the intrusive thought experience is emphasised by the results in this thesis.

9.4 Conclusion and Final Comments

Experiencing thoughts about harming another person is considered a normal phenomenon. However, in certain instances such as when experienced by individuals diagnosed with OCD or in individuals who have a history of aggressive behaviour, these thoughts may become problematic. The cognitive appraisal model of OCD emphasises the role that maladaptive beliefs and self-appraisals have in the experience of AITs, perpetuating the use of compulsive behaviours over time. Further, the GAM recognises that precursors of aggressive behaviour may be explained by the frequent rehearsal of aggressive scripts, a history of aggressive behaviour, and beliefs that normalise violence.

This thesis has explored the phenomena of AITs and aggressive scripts; two similar constructs in terms of content, but distinct with regards to potential behavioural outcomes. Overall, this thesis has identified some of the similarities and differences between AITs and aggressive scripts, and explored the subjective experiences of aggressive thoughts in individuals with a history of violence. This research has improved the understanding of what factors may differentiate AITs from aggressive scripts, including a history of violent behaviour, attitudes that support violence, and the recognition that one's aggressive thought is inconsistent with one's sense of self and conduct. Further, this research has highlighted the impact of belief systems and experiences, including how attitudes towards violence and obsessive beliefs may impact on how AITs or aggressive scripts are appraised and recognised. This research has the potential to improve understandings of aggressive thinking, including identifying factors that may protect against certain behaviours. Further, this thesis has identified issues that exist in current measures of AITs and aggressive scripts, which may encourage improvements to these measures in future research. Finally, the results of this

thesis may inform assessment and treatment approaches for AITs and aggressive scripts, aiding in the early detection of these thoughts to improve outcomes for those who report them.

References

- Aardema, F., Moulding, R., Melli, G., Radomsky, A. S., Doron, G., Audet, J. S., & Purcell-Lalonde, M. (2017). The role of feared possible selves in obsessive-compulsive and related disorders: A comparative analysis of a core cognitive self-construct in clinical samples. *Clinical Psychology & Psychotherapy*, 25(1), 19-29. https://doi.org/10.1002/cpp.2121
- Aardema, F., Moulding, R., Radomsky, A. S., Doron, G., Allamby, J., & Souki, E. (2013). Fear of self and obsessionality: Development and validation of the Fear of Self Questionnaire. *Journal of Obsessive-Compulsive and Related Disorders*, 2(3), 306-315. https://doi.org/10.1016/j.jocrd.2013.05.005
- Aardema, F., & O'Connor, K. (2007). The menace within: Obsessions and the self. *Journal of Cognitive Psychotherapy*, 21(3), 182-197. https://doi.org/10.1891/088983907781494573
- Aardema, F., Radomsky, A. S., Moulding, R., Wong, S. F., Bourguignon, L., & Giraldo-O'Meara, M. (2021). Development and validation of the multidimensional version of the fear of self questionnaire: Corrupted, culpable and malformed feared possible selves in obsessive-compulsive and body-dysmorphic symptoms. *Clinical Psychology & Psychotherapy*, 1-21. https://doi.org/10.1002/cpp.2565
- Aardema, F., Wong, S. F., Audet, J. S., Melli, G., & Baraby, L. P. (2019). Reduced fear-of-self is associated with improvement in concerns related to repugnant obsessions in obsessive-compulsive disorder. *British Journal of Clinical Psychology*, *58*, 327-341. https://doi.org/10.1111/bjc.12214
- Abramovitch, A., Abramowitz, J. S., Riemann, B. C., & McKay, D. (2020). Severity benchmarks and contemporary clinical norms for the Obsessive-Compulsive Inventory-Revised (OCI-R). *Journal of Obsessive-Compulsive and Related Disorders*, 27. https://doi.org/10.1016/j.jocrd.2020.100557

- Abramowitz, J. S., Fabricant, L. E., Taylor, S., Deacon, B. J., McKay, D., & Storch, E. A. (2014). The relevance of analogue studies for understanding obsessions and compulsions. *Clinical Psychology Review*, *34*(3), 206-217. https://doi.org/10.1016/j.cpr.2014.01.004
- Abramowitz, J. S., Lackey, G. R., & Wheaton, M. G. (2009). Obsessive-compulsive symptoms: the contribution of obsessional beliefs and experiential avoidance. *J Anxiety Disord*, 23(2), 160-166. https://doi.org/10.1016/j.janxdis.2008.06.003
- Abramowitz, J. S., Schwartz, S. A., & Moore, K. M. (2003). Obsessional thoughts in postpartum females and their partners: content, severity, and relationship with depression. *Journal of Clinical Psychology in Medical Settings*, 10(3), 157-164. https://doi.org/https://doi.org/10.1023/A:1025454627242
- Ahern, C., Kyrios, M., & Meyer, D. (2015). Exposure to unwanted intrusions, neutralizing and their effects on self-worth and obsessive-compulsive phenomena. *Journal of Behavior Therapy and Experimental Psychiatry*, 49(Pt B), 216-222. https://doi.org/10.1016/j.jbtep.2015.07.008
- Amir, N., Cashman, L., & Foa, E. B. (1997). Strategies of thought control in obsessive-compulsive disorder. *Behaviour Research and Therapy*, *35*(8), 775-777. https://doi.org/10.1016/S0005-7967(97)00030-2
- Anderson, C. A., & Bushman, B. J. (2002). Human aggression. *Annual Review of Psychology*, *53*(1), 27-51. https://doi.org/10.1146/annurev.psych.53.100901.135231
- Anderson, C. A., Gentile, D. A., & Buckley, K. E. (2007). Violent video game effects on children and adolescents: Theory, research, and public policy. University Press.
- Andrews, D. A., Bonta, J., & Wormith, J. S. (2011). The Risk-Need-Responsivity (RNR) Model. *Criminal Justice and Behavior*, *38*(7), 735-755. https://doi.org/10.1177/0093854811406356

- Anestis, M. D., Anestis, J. C., Selby, E. A., & Joiner, T. E. (2009). Anger rumination across forms of aggression. *Personality and Individual Differences*, 46(2), 192-196. https://doi.org/10.1016/j.paid.2008.09.026
- Archer, J., & Haigh, A. (1997). Beliefs about aggression among male and female prisoners.

 *Aggressive Behavior: Official Journal of the International Society for Research on Aggression, 23(6), 405-415. https://doi.org/10.1002/(SICI)1098-2337(1997)23:6<a href="https://doi.org/10.1002/(SICI)1098-2337(1997)23:6<a href="https://doi.org/10.1002/(SICI)1098-2337(1997)23:6<a href="https://doi.org/10.1002/(SICI)1098-2337(1998)23:6<a href="https://doi.org/10.1002
- Audet, J.-S., Aardema, F., & Moulding, R. (2016). Contextual determinants of intrusions and obsessions: The role of ego-dystonicity and the reality of obsessional thoughts.

 Journal of Obsessive-Compulsive and Related Disorders, 9, 96-106.

 https://doi.org/10.1016/j.jocrd.2016.04.003
- Auvinen-Lintunen, L., Häkkänen-Nyholm, H., Ilonen, T., & Tikkanen, R. (2015). Sex Differences in Homicidal Fantasies among Finnish University Students. *Psychology*, 6(1), 39-47. https://doi.org/10.4236/psych.2015.61004
- Bäckström, M., & Björklund, F. (2008). The Measures of Criminal Attitudes and Associates (MCAA). *Criminal Justice and Behavior*, *35*(11), 1398-1410. https://doi.org/10.1177/0093854808322239
- Baer, L. (1994). Factor analysis of symptom subtypes of obsessive compulsive disorder and their relation to personality and tic disorders. *The Journal of Clinical Psychiatry*, 55(3), 18-23.
- Belloch, A., Morillo, C., & Garcia-Soriano, G. (2007). Strategies to Control Unwanted Intrusive Thoughts: Which are Relevant and Specific in Obsessive-Compulsive Disorder? *Cognitive Therapy and Research*, *33*(1), 75-89. https://doi.org/10.1007/s10608-007-9141-2

- Belloch, A., Morillo, C., Lucero, M., Cabedo, E., & Carrió, C. (2004). Intrusive thoughts in non-clinical subjects: the role of frequency and unpleasantness on appraisal ratings and control strategies. *Clinical Psychology & Psychotherapy*, 11(2), 100-110. https://doi.org/10.1002/cpp.397
- Belloch, A., Roncero, M., & Perpiñá, C. (2012). Ego-Syntonicity and Ego-Dystonicity Associated with Upsetting Intrusive Cognitions. *Journal of Psychopathology and Behavioral Assessment*, 34(1), 94-106. https://doi.org/10.1007/s10862-011-9255-4
- Beres, D. (1960). Perception, imagination, and reality. *International Journal of Psycho-Analysis*, 41(327-334).
- Berkowitz, L. (1993). *Aggression: Its causes, consequences, and control*. Mcgraw-Hill Book Company.
- Berman, N. C., Wheaton, M. G., & Abramowitz, J. S. (2012). The "Arnold Schwarzenegger Effect": is strength of the "victim" related to misinterpretations of harm intrusions? Behaviour Research and Therapy, 50(12), 761-766.

 https://doi.org/10.1016/j.brat.2012.09.002
- Berry, L., & Laskey, B. (2012). A review of obsessive intrusive thoughts in the general population. *Journal of Obsessive-Compulsive and Related Disorders*, 1(2), 125-132. https://doi.org/10.1016/j.jocrd.2012.02.002
- Bersani, B. E., & Doherty, E. E. (2018). Desistance from offending in the twenty-first century. *Annual Review of Criminology, 1*, 311-334. https://doi.org/10.1146/annurev-criminol-032317-092112
- Bhar, S. S., & Kyrios, M. (2007). An investigation of self-ambivalence in obsessive-compulsive disorder. *Behaviour Research and Therapy*, *45*(8), 1845-1857. https://doi.org/10.1016/j.brat.2007.02.005

- Bloch, M. H., Landeros-Weisenberger, A., Rosario, M. C., Pittenger, C., & Leckman, J. F. (2008). Meta-analysis of the symptom structure of obsessive-compulsive disorder. *American Journal of Psychiatry*, 165(12), 1532-1542.

 https://doi.org/10.1176/appi.ajp.2008.08020320
- Booth, B. D., Friedman, S. H., Curry, S., Ward, H., & Stewart, S. E. (2014). Obsessions of child murder: underrecognized manifestations of obsessive-compulsive disorder. *Journal of the American Academy of Psychiatry and the Law, 42* (1), 6-74.
- Bouvard, M., Fournet, N., Denis, A., Sixdenier, A., & Clark, D. (2017). Intrusive thoughts in patients with obsessive compulsive disorder and non-clinical participants: a comparison using the International Intrusive Thought Interview Schedule. *Cognitive Behavior Therapy*, 46(4), 287-299. https://doi.org/10.1080/16506073.2016.1262894
- Brakoulias, V., Starcevic, V., Belloch, A., Brown, C., Ferrao, Y. A., Fontenelle, L. F.,
 Lochner, C., Marazziti, D., Matsunaga, H., Miguel, E. C., Reddy, Y. C. J., do Rosario,
 M. C., Shavitt, R. G., Shyam Sundar, A., Stein, D. J., Torres, A. R., & Viswasam, K.
 (2017, Jul). Comorbidity, age of onset and suicidality in obsessive-compulsive
 disorder (OCD): An international collaboration. *Compr Psychiatry*, 76, 79-86.
 https://doi.org/10.1016/j.comppsych.2017.04.002
- Brakoulias, V., Starcevic, V., Berle, D., Milicevic, D., Hannan, A., & Martin, A. (2014). The relationships between obsessive-compulsive symptom dimensions and cognitions in obsessive-compulsive disorder. *Psychiatric Quarterly*, 85(2), 133-142. https://doi.org/10.1007/s11126-013-9278-y
- Brakoulias, V., Starcevic, V., Berle, D., Milicevic, D., Moses, K., Hannan, A., Sammut, P., & Martin, A. (2013). The characteristics of unacceptable/taboo thoughts in obsessive-compulsive disorder. *Comprehensive Psychiatry*, *54*(7), 750-757. https://doi.org/10.1016/j.comppsych.2013.02.005
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77-101. https://doi.org/10.1191/1478088706qp0630a

- Brown, G. L., Goodwin, F. K., Ballenger, J. C., Goyer, P. F., & Major, L. F. (1979).

 Aggression in humans correlates with cerebrospinal fluid amine metabolites.

 Psychiatry Research, 1(2), 131-139. https://doi.org/10.1016/0165-1781(79)90053-2
- Burgess, A. W., Hartman, C. R., Ressler, R. K., Douglas, J. E., & McCormack, A. (1986).

 Sexual homicide: A motivational model. *Journal of interpersonal violence*, *I*(3), 251-272. https://www.ojp.gov/ncjrs/virtual-library/abstracts/sexual-homicide-motivational-model
- Bushman, B. J. (2002). Does venting anger feed or extinguish the flame? Catharsis, rumination, distraction, anger, and aggressive responding. *Personality and Social Psychology Bulletin*, 28(6), 724-731. https://doi.org/10.1177/0146167202289002
- Bushman, B. J., & Anderson, C. A. (2002). Violent Video Games and Hostile Expectations:

 A Test of the General Aggression Model. *Personality and Social Psychology Bulletin*,

 28(12), 1679-1686. https://doi.org/10.1177/014616702237649
- Carabellese, F., Maniglio, R., Greco, O., & Catanesi, R. (2011). The role of fantasy in a serial sexual offender: a brief review of the literature and a case report. *Journal of Forensic Sciences*, 56(1), 256-260. https://doi.org/10.1111/j.1556-4029.2010.01536.x
- Ching, T. H., Williams, M., & Siev, J. (2017). Violent obsessions are associated with suicidality in an OCD analog sample of college students. *Cognitive Behavior Therapy*, 46(2), 129-140. https://doi.org/10.1080/16506073.2016.1228084
- Clark, D. A. (2004). Cognitive-behavioral therapy for OCD. Guilford Press.
- Clark, D. A. (2005). *Intrusive thoughts in clinical disorders: Theory, research, and treatment* Guilford Press.
- Clark, D. A., Abramowitz, J., Alcolado, G. M., Alonso, P., Belloch, A., Bouvard, M., Coles, M. E., Doron, G., Fernández-Álvarez, H., Garcia-Soriano, G., Ghisi, M., Gomez, B.,

- Inozu, M., Moulding, R., Radomsky, A. S., Shams, G., Sica, C., Simos, G., & Wong, W. (2014). Part 3. A question of perspective: The association between intrusive thoughts and obsessionality in 11 countries. *Journal of Obsessive-Compulsive and Related Disorders*, *3*(3), 292-299. https://doi.org/10.1016/j.jocrd.2013.12.006
- Clark, D. A., & Purdon, C. (1995). The assessment of unwanted intrusive thoughts: a review and critique of the literature. *Behaviour Research and Therapy*, *33*(8), 967-976. https://doi.org/https://doi.org/10.1016/0005-7967(95)00030-2
- Clark, D. A., Purdon, C., & Byers, E. S. (2000). Appraisal and control of sexual and non-sexual intrusive thoughts in university students. *Behaviour Research and Therapy*, 38(5), 439-455. https://doi.org/doi.org/10.1016/S0005-7967(99)00047-9
- Coccaro, E. F., Berman, M. E., & Kavoussi, R. J. (1997). Assessment of life history of aggression: development and psychometric characteristics. *Psychiatry Research*, 73(3), 147-157. https://doi.org/10.1016/S0165-1781(97)00119-4
- Crabb, P. B. (2000). The material culture of homicidal fantasies. *Official Journal of the International Society for Research on Aggression*, 26(3), 225-234. https://doi.org/10.1002/(SICI)1098-2337(2000)26:3<225::AID-AB2>3.0.CO;2-R
- Daff, E. S., Gilbert, F., & Daffern, M. (2015). The Relationship Between Anger and Aggressive Script Rehearsal in an Offender Population. *Psychiatry, Psychology and Law, 22*(5), 731-739. https://doi.org/10.1080/13218719.2014.986837
- DeLapp, R. C., Chasson, G. S., Swerbilow, J., Gibby, B., Tellawi, G., & Williams, M. T. (2018). The normative nature of aggressive intrusive thinking among an underserved incarcerated population compared with a student sample. *International Journal of Offender Therapy and Comparative Criminology*, 62(13), 4142-4157. https://doi.org/10.1177/0306624X18758534

- Denson, T. F. (2013). The multiple systems model of angry rumination. *Personality and Social Psychology Review*, 17(2), 103-123. https://doi.org/10.1177/1088868312467086
- Denson, T. F., Moulds, M. L., & Grisham, J. R. (2012). The effects of analytical rumination, reappraisal, and distraction on anger experience. *Behav Ther*, 43(2), 355-364. https://doi.org/10.1016/j.beth.2011.08.001
- Doron, G., & Kyrios, M. (2005). Obsessive compulsive disorder: a review of possible specific internal representations within a broader cognitive theory. *Clinical Psychology Review*, 25(4), 415-432. https://doi.org/10.1016/j.cpr.2005.02.002
- Doron, G., Moulding, R., Kyrios, M., & Nedeljkovic, M. (2008). Sensitivity of self-beliefs in obsessive compulsive disorder. *Depression and Anxiety*, 25(10), 874-884. https://doi.org/10.1002/da.20369
- Dunne, A. L., Gilbert, F., Lee, S., & Daffern, M. (2018). The role of aggression-related early maladaptive schemas and schema modes in aggression in a prisoner sample. *Aggress Behav*, 44(3), 246-256. https://doi.org/10.1002/ab.21747
- Fairbrother, N., Collardeau, F., Woody, S. R., Wolfe, D. A., & Fawcett, J. M. (2022).

 Postpartum thoughts of infant-related harm and obsessive-compulsive disorder: relation to maternal physical aggression toward the infant. *The Journal of Clinical Psychiatry*, 83(2), 39944. https://doi.org/10.4088/JCP.21m14006
- Fairbrother, N., & Woody, S. R. (2008). New mothers' thoughts of harm related to the newborn. *Archive Womens Mental Health*, 11(3), 221-229. https://doi.org/10.1007/s00737-008-0016-7
- Fernandez, S. J., Daffern, M., Moulding, R., & Nedeljkovic, M. (2022). Exploring predictors of aggressive intrusive thoughts and aggressive scripts: Similarities and differences in phenomenology. *Aggress Behavior*, 1-3. https://doi.org/10.1002/ab.22061

- Ferrier, S., & Brewin, C. R. (2005). Feared identity and obsessive-compulsive disorder. *Behaviour Research Therapy*, 43(10), 1363-1374.

 https://doi.org/10.1016/j.brat.2004.10.005
- Foa, E. B., Huppert, J. D., Leiberg, S., Langner, R., Kichic, R., Hajcak, G., & Salkovskis, P. M. (2002). The Obsessive-Complusive Inventory: Development and validation of a short version. *Psychological Assessment*, 14(4), 485-495.
 https://doi.org/10.1037/1040-3590.14.4.485
- Freeston, M. H., Ladouceur, R., Provencher, M., & Blais, F. (1995). Strategies used with intrusive thoughts: Context, appraisal, mood, and efficacy. *Journal of Anxiety Disorders*, 9(3), 201-215. https://doi.org/10.1016/0887-6185(95)00002-6
- Freeston, M. H., Ladouceur, R., Thibodeau, N., & Gagnon, F. (1991). Cognitive intrusions in a non-clinical population. I. Response style, subjective experience, and appraisal. *Behaviour Research and Therapy*, 29(6), 585-597. https://doi.org/10.1016/0005-7967(91)90008-Q
- Friedman, S. H., Sorrentino, R. M., Stankowski, J. E., Holden, C. E., & Resnick, P. J. (2008). Psychiatrists' knowledge about maternal filicidal thoughts. *Compr Psychiatry*, 49(1), 106-110. https://doi.org/10.1016/j.comppsych.2007.07.001
- Garcia-Soriano, G., Belloch, A., Morillo, C., & Clark, D. A. (2011). Symptom dimensions in obsessive-compulsive disorder: from normal cognitive intrusions to clinical obsessions. *Journal of Anxiety Disorders*, 25(4), 474-482. https://doi.org/10.1016/j.janxdis.2010.11.012
- Gendreau, P., Little, T., & Goggin, C. (1996). A meta-analysis of the predictors of adult offender recidivism: What works! *Criminology*, *34*(4), 575-608. https://doi.org/10.1111/j.1745-9125.1996.tb01220.x

- Gilbert, F., & Daffern, M. (2017). Aggressive scripts, violent fantasy and violent behavior: A conceptual clarification and review. *Aggression and Violent Behavior*, *36*, 98-107. https://doi.org/10.1016/j.avb.2017.05.001
- Gilbert, F., Daffern, M., & Anderson, C. A. (2017). The General Aggression Model and Its Application to Violent Offender Assessment and Treatment. In *The Wiley Handbook of Violence and Aggression* (pp. 1-13). https://doi.org/10.1002/9781119057574.whbva037
- Gilbert, F., Daffern, M., Talevski, D., & Ogloff, J. R. (2013). The role of aggression-related cognition in the aggressive behavior of offenders: A general aggression model perspective. *Criminal Justice and Behavior*, 40(2), 119-138. https://doi.org/doi.org/10.1177/0093854812467943
- Grisso, T., Davis, J., Vesselinov, R., Appelbaum, P. S., & Monahan, J. (2000). Violent thoughts and violent behavior following hospitalization for mental disorder. *Journal of Consulting and Clinical Psychology*, 68(3), 388-398. https://doi.org/10.1037/0022-006x.68.3.388
- Gross, J. J. (1998). The emerging field of emotion regulation: An integrative review. *Review of general psychology*, 2(3), 271-299.
- Guidano, V., & Liotti, G. (1983). Cognitive processes and emotional disorders. Guildford Press.
- Halderman, B. L., Zelhart, P. F., & Jackson, T. T. (1985). A study of fantasy: Determinants of fantasy function and content. . *Journal of Clinical Psychology*, 41(3), 325-330. https://doi.org/10.1002/1097-4679(198505)41:3<325::AID-JCLP2270410304>3.0.CO;2-K
- Higgins, E. T. (1987). Self-discrepancy: A theory relating self and affect. *Psychological Review*, 94(3), 319-340. https://doi.org/10.1037/0033-295X.94.3.319

- Hosie, J., Dunne, A., Simpson, K., & Daffern, M. (2021). Aggressive Script Rehearsal in Adult Offenders: Characteristics and Association with Self-Reported Aggression. *Journal of interpersonal violence*. https://doi.org/10.1177/08862605211062992
- Hosie, J., Dunne, A. L., Meyer, D., & Daffern, M. (2022). Aggressive script rehearsal in adult offenders: Relationships with emotion regulation difficulties and aggressive behavior. *Aggress Behav*, 48(1), 5-16. https://doi.org/10.1002/ab.22000
- Hosie, J., Gilbert, F., Simpson, K., & Daffern, M. (2014). An examination of the relationship between personality and aggression using the general aggression and five factor models. *Aggress Behav*, 40(2), 189-196. https://doi.org/10.1002/ab.21510
- Hosie, J., Simpson, K., Dunne, A., & Daffern, M. (2022). A study of the relationships between rumination, anger rumination, aggressive script rehearsal, and aggressive behavior in a sample of incarcerated adult males. *Journal of Clinical Psychology*. https://doi.org/10.1002/jclp.23341
- Huesmann, L. R. (1988). An information processing model for the development of aggression. *Aggressive Behavior*, 14(1), 13-24. https://doi.org/10.1002/1098-2337(1988)14:1https://doi.org/10.1002/1098-2337(1988)14:1<a href="https://doi.org/10.1002/1098-2337(1988)14:1<a href="https://doi.org/10.1002/10.1002/10.1002/10.1002/10.1002/10.1002/10.1002/10.10
- Huesmann, L. R. (1998). The role of social information processing and cognitive schema in the acquisition and maintenance of habitual aggressive behavior. In *Human Aggression* (pp. 73-109). Academic Press. https://doi.org/10.1016/B978-012278805-5/50005-5
- Huesmann, L. R., & Eron, L. D. (1984). Cognitive processes and the persistence of aggressive behavior. *Aggressive Behavior*, 10(3), 243-251. https://doi.org/10.1002/1098-2337(1984)10:3<243::AID-AB2480100308>3.0.CO;2-6
- Huesmann, L. R., & Eron, L. D. (1989). Individual differences and the trait of aggression. *European Journal of Personality*, 3(2), 95-106.

 https://doi.org/10.1002/per.2410030204

- Huesmann, L. R., & Eron, L. D. (2013). *Television and the aggressive child: A cross-national comparison*. Taylor and Francis.
- Huesmann, L. R., & Guerra, N. G. (1997). Children's normative beliefs about aggression and aggressive behavior. *Journal of Personality and Social Psychology*, 72(2), 408-419. https://doi.org/doi.org/10.1037/0022-3514.72.2.408
- Hvenegaard, M., Watkins, E. R., Poulsen, S., Rosenberg, N. K., Gondan, M., Grafton, B., Austin, S. F., Howard, H., & Moeller, S. B. (2015). Rumination-focused Cognitive Behaviour Therapy vs. Cognitive Behaviour Therapy for depression: Study protocol for a randomised controlled superiority trial. *Trials*, *16*(1), 344-350. https://doi.org/http://10.1186/s13063-015-0875-y
- Izadi, R., Asgari, K., Neshatdust, H., & Abedi, M. (2012). Assessment of obsessive beliefs in individuals with obsessive- compulsive disorder in comparison to healthy sample.
 International Journal of Psychology and Counselling, 4(7), 81-85.
 https://doi.org/10.5897/ijpc12.010
- Jacoby, R. J., Leonard, R. C., Riemann, B. C., & Abramowitz, J. S. (2015). Self-Punishment as a Maladaptive Thought Control Strategy Mediates the Relationship Between Beliefs About Thoughts and Repugnant Obsessions. *Cognitive Therapy and Research*, 40(2), 179-187. https://doi.org/10.1007/s10608-015-9741-1
- Jaeger, T., Moulding, R., Yang, Y. H., David, J., Knight, T., & Norberg, M. M. (2021). A systematic review of obsessive-compulsive disorder and self: Self-esteem, feared self, self-ambivalence, egodystonicity, early maladaptive schemas, and self concealment.

 Journal of Obsessive-Compulsive and Related Disorders, 31.

 https://doi.org/10.1016/j.jocrd.2021.100665
- Julien, D., O'Connor, K. P., & Aardema, F. (2007). Intrusive thoughts, obsessions, and appraisals in obsessive-compulsive disorder: a critical review. *Clin Psychol Rev*, 27(3), 366-383. https://doi.org/10.1016/j.cpr.2006.12.004

- Kelty, S. F., Hall, G., & Watt, B. D. (2011). You Have to Hit Some People! Measurement and Criminogenic Nature of Violent Sentiments in Australia. *Psychiatry, Psychology and Law*, 18(1), 15-32. https://doi.org/10.1080/13218710903566961
- Kenrick, D. T., & Sheets, V. (1993). Homicidal fantasies. *Ethology and Sociobiology, 14*(4), 231-246. https://doi.org/10.1016/0162-3095(93)90019-E
- Klinger, E. (1990). *Daydreaming: Using waking fantasy and imagery for self-knowledge and creativity*. J P Tarcher.
- Lee, H. J., & Kwon, S. M. (2003). Two different types of obsession: autogenous obsessions and reactive obsessions. *Behaviour Research and Therapy*, 41(1), 11-29. https://doi.org/10.1016/S0005-7967(01)00101-2
- Leonard, R. C., & Riemann, B. C. (2012). The co-occurrence of obsessions and compulsions in OCD. *Journal of Obsessive-Compulsive and Related Disorders*, 1(3), 211-215. https://doi.org/10.1016/j.jocrd.2012.06.002
- Link, B., Monahan, J., Stueve, A., & Cullen, F. (1999). Real in their consequences: a socialogical approach to understanding the association between psychotic symptoms and violenc. *Americal Sociological Review*, *64*, 316-332. https://doi.org/10.2307/2657535
- Llorens-Aguilar, S., Arnaez, S., Aardema, F., & Garcia-Soriano, G. (2021). The relationship between obsessions and the self: Feared and actual self-descriptions in a clinical obsessive-compulsive disorder sample. *Clin Psychol Psychother*.

 https://doi.org/10.1002/cpp.2656
- Lovibond, P. F., & Lovibond, S. H. (1995). The structure of negative emotional states:

 Comparison of the Depression Anxiety Stress Scales (DASS) with the Beck

 Depression and Anxiety Inventories. *Behaviour Research and Therapy*, 33(3), 335-343. https://doi.org/https://doi.org/10.1016/0005-7967(94)00075-U

- Luciano, J. V., Belloch, A., Algarabel, S., Tomás, J. M., Morillo, C., & Lucero, M. (2006). Confirmatory Factor Analysis of the White Bear Suppression Inventory and the Thought Control Questionnaire. *European Journal of Psychological Assessment*, 22(4), 250-258. https://doi.org/10.1027/1015-5759.22.4.250
- Lück, H. E., & Timaeus, E. (1969). Scales for the measurement of manifest anxiety (MAS) and social desirability (SDS-E and SDS-CM). *Diagnostica*, *15*, 134-141.
- Melli, G., Aardema, F., & Moulding, R. (2016). Fear of Self and Unacceptable Thoughts in Obsessive-Compulsive Disorder. *Clinical Psychology Psychotherapy*, 23(3), 226-235. https://doi.org/10.1002/cpp.1950
- Mills, J. F., Kroner, D. G., & Forth, A. E. (2002). Measures of Criminal Attitudes and Associates (MCAA) development, factor structure, reliability, and validity.

 Assessment, 9(3), 240-253. https://doi.org/10.1177/1073191102009003003
- Moeller, S. B., Gondan, M., & Novaco, R. W. (2017). Violent images, anger and physical aggression among male forensic inpatients. *Personality and Individual Differences*, 105, 268-274. https://doi.org/10.1016/j.paid.2016.10.009
- Morillo, C., Belloch, A., & Garcia-Soriano, G. (2007). Clinical obsessions in obsessive-compulsive patients and obsession-relevant intrusive thoughts in non-clinical, depressed and anxious subjects: where are the differences? *Behav Res Ther*, 45(6), 1319-1333. https://doi.org/10.1016/j.brat.2006.11.005
- Morrison, F. P., Fullam, R., Thomson, K., Meyer, D., & Daffern, M. (2022). International Journal of Offender Therapy and Comparative Criminology, 0306624X221086581. (2022). Results From a Non-Randomized Pilot Study Evaluating the Impact of a Novel Group Treatment Program Targeting Aggressive Script Rehearsal and Emotion Regulation in a Sample of Incarcerated Males. *International Journal of Offender Therapy and Comparative Criminology*, 1-19. https://doi.org/10.1177/0306624X221086581

- Moscovitch, D. A., McCabe, R. E., Antony, M. M., Rocca, L., & Swinson, R. P. (2008). Anger experience and expression across the anxiety disorders. *Depression and Anxiety*, 25(2), 107-113. https://doi.org/10.1002/da.20280
- Moulding, R., Aardema, F., & O'Connor, K. (2014a). Repugnant obsessions: A review of the phenomenology, theoretical models, and treatment of sexual and aggressive obsessional themes in OCD. *Journal of Obsessive-Compulsive and Related Disorders*, 3(2), 161-168. https://doi.org/10.1016/j.jocrd.2013.11.006
- Moulding, R., Aardema, F., & O'Connor, K. P. (2014b). Repugnant obsessions: A review of the phenomenology, theoretical models, and treatment of sexual and aggressive obsessional themes in OCD. *Journal of Obsessive-Compulsive and Related Disorders*, 3(2), 161-168. https://doi.org/10.1016/j.jocrd.2013.11.006
- Moulding, R., Anglim, J., Nedeljkovic, M., Doron, G., Kyrios, M., & Ayalon, A. (2011). The Obsessive Beliefs Questionnaire (OBQ): examination in nonclinical samples and development of a short version. *Assessment*, 18(3), 357-374. https://doi.org/10.1177/1073191110376490
- Moulding, R., Coles, M. E., Abramowitz, J. S., Alcolado, G. M., Alonso, P., Belloch, A.,
 Bouvard, M., Clark, D. A., Doron, G., Fernández-Álvarez, H., García-Soriano, G.,
 Ghisi, M., Gómez, B., Inozu, M., Radomsky, A. S., Shams, G., Sica, C., Simos, G., & Wong, W. (2014). Part 2. They scare because we care: The relationship between obsessive intrusive thoughts and appraisals and control strategies across 15 cities.
 Journal of Obsessive-Compulsive and Related Disorders, 3(3), 280-291.
 https://doi.org/10.1016/j.jocrd.2014.02.006
- Myers, S. G., Fisher, P. L., & Wells, A. (2008). Belief domains of the Obsessive Beliefs

 Questionnaire-44 (OBQ-44) and their specific relationship with obsessive-compulsive symptoms. *J Anxiety Disord*, 22(3), 475-484.

 https://doi.org/10.1016/j.janxdis.2007.03.012

- Nagtegaal, M. H., Rassin, E., & Muris, P. (2006). Aggressive fantasies, thought control strategies, and their connection to aggressive behaviour. *Personality and Individual Differences*, 41(8), 1397-1407. https://doi.org/10.1016/j.paid.2006.05.009
- Nikodijevic, A., Moulding, R., Anglim, J., Aardema, F., & Nedeljkovic, M. (2015). Fear of self, doubt and obsessive compulsive symptoms. *J Behav Ther Exp Psychiatry*, 49(Pt B), 164-172. https://doi.org/10.1016/j.jbtep.2015.02.005
- O'Connor, K. (2002). Intrusions and inferences in obsessive compulsive disorder. *Clinical Psychology & Psychotherapy*, 9(1), 38-46. https://doi.org/10.1002/cpp.303
- O'Neill, M. L., Nenzel, M. E., & Caldwell, W. (2009). Intrusive thoughts and psychopathy in a student and incarcerated sample. *Journal of Behavior Therapy and Experimental Psychiatry*, 40(1), 147-157. https://doi.org/10.1016/j.jbtep.2008.07.004
- O'Connor, K., O'Connor, D., J., & Aardema, F. (2009). Intrusions related to obsessive-compulsive disorder: a question of content or context? *Journal of Clinical Psychology*, 65(7), 709-722. https://doi.org/10.1002/jclp.20578
- Obsessive Compulsive Cognitions Working Group. (2003). Psychometric validation of the Obsessive Beliefs Questionnaire and the Interpretation of Intrusions Inventory: Part I. *Behaviour Research and Therapy*, 41(8), 863-878. https://doi.org/10.1016/s0005-7967(02)00099-2
- OCCWG. (1997). Cognitive assessment of obsessive-compulsive disorder. *Behaviour Research and Therapy*, 35(7), 667-681. https://doi.org/10.1016/S0005-7967(97)00017-X
- OCCWG. (2005). Psychometric validation of the obsessive belief questionnaire and interpretation of intrusions inventory--Part 2: Factor analyses and testing of a brief version. *Behavior Research Therapy*, 43(11), 1527-1542. https://doi.org/10.1016/j.brat.2004.07.010

- Oyserman, D., & Markus, H. R. (1990). Possible selves and delinquency. *Journal of Personality and Social Psychology*, 59, 112-125. https://doi.org/10.1037/0022-3514.59.1.112
- Papalia, N., Spivak, B., Daffern, M., & Ogloff, J. R. P. (2019). A meta-analytic review of the efficacy of psychological treatments for violent offenders in correctional and forensic mental health settings. *Clinical Psychology: Science and Practice, 26*(2). https://doi.org/10.1111/cpsp.12282
- Parkinson, L., & Rachman, S. (1981). Part II. The nature of intrusive thoughts. *Advances in Behaviour Research and Therapy*, *3*(3), 101-110. https://doi.org/https://doi.org/10.1016/0146-6402(81)90008-4
- Parsons, E. M., Straub, K. T., Smith, A. R., & Clerkin, E. M. (2017). Body Dysmorphic, Obsessive-Compulsive, and Social Anxiety Disorder Beliefs as Predictors of In Vivo Stressor Responding. *The Journal of Nervous and Mental Disease*, 205(6), 471-479. https://doi.org/10.1097/NMD.000000000000000656
- Pascual-Vera, B., Akin, B., Belloch, A., Bottesi, G., Clark, D. A., Doron, G., Fernandez-Alvarez, H., Ghisi, M., Gomez, B., Inozu, M., Jimenez-Ros, A., Moulding, R., Ruiz, M. A., Shams, G., & Sica, C. (2019). The cross-cultural and transdiagnostic nature of unwanted mental intrusions. *International Journal of Clinical Health Psychology*, 19(2), 85-96. https://doi.org/10.1016/j.ijchp.2019.02.005
- Patel, G. (2015). Violent thoughts and fantasies in a high secure mentally disordered offender group: an exploratory study University of Nottingham].

 http://eprints.nottingham.ac.uk/29445/1/PhD_FINAL2015.pdf
- Paternoster, R., & Bushway, S. (2009). Desistance and the feared self: Toward an identity theory of criminal desistance. *Journal of Criminal Law and Criminology*, 99(4), 1103-1156. https://www.jstor.org/stable/20685067

- Paulhus, D. L. (2017). Socially Desirable Responding on Self-Reports. In *Encyclopedia of Personality and Individual Differences* (pp. 1-5). https://doi.org/10.1007/978-3-319-28099-8 1349-1
- Pedersen, W. C., Denson, T. F., Goss, R. J., Vasquez, E. A., Kelley, N. J., & Miller, N. (2011). The impact of rumination on aggressive thoughts, feelings, arousal, and behaviour. *British Journal of Social Psychology*, *50*(Pt 2), 281-301. https://doi.org/10.1348/014466610X515696
- Peled, M., & Moretti, M. M. (2009). Ruminating on Rumination: are Rumination on Anger and Sadness Differentially Related to Aggression and Depressed Mood? *Journal of Psychopathology and Behavioral Assessment*, 32(1), 108-117. https://doi.org/10.1007/s10862-009-9136-2
- Pinto, A., Greenberg, B. D., Grados, M. A., Bienvenu, O. J., 3rd, Samuels, J. F., Murphy, D. L., Hasler, G., Stout, R. L., Rauch, S. L., Shugart, Y. Y., Pauls, D. L., Knowles, J. A., Fyer, A. J., McCracken, J. T., Piacentini, J., Wang, Y., Willour, V. L., Cullen, B., Liang, K. Y., Hoehn-Saric, R., Riddle, M. A., Rasmussen, S. A., & Nestadt, G. (2008a, Jul 15). Further development of YBOCS dimensions in the OCD Collaborative Genetics study: symptoms vs. categories. *Psychiatry Res*, 160(1), 83-93. https://doi.org/10.1016/j.psychres.2007.07.010
- Pinto, A., Greenberg, B. D., Grados, M. A., Bienvenu, O. J., Samuels, J. F., Murphy, D. L., Hasler, G., Stout, R. L., Rauch, S. L., Shugart, Y. Y., Pauls, D. L., Knowles, J. A., Fyer, A. J., McCracken, J. T., Piacentini, J., Wang, Y., Willour, V. L., Cullen, B., Liang, K. Y., Hoehn-Saric, R., Riddle, M. A., Rasmussen, S. A., & Nestadt, G. (2008b). Further development of YBOCS dimensions in the OCD Collaborative Genetics study: symptoms vs. categories. *Psychiatry Research*, 160(1), 83-93. https://doi.org/10.1016/j.psychres.2007.07.010
- Podubinski, T., Lee, S., Hollander, Y., & Daffern, M. (2017). Patient characteristics associated with aggression in mental health units. *Psychiatry Research*, *250*, 141-145. https://doi.org/10.1016/j.psychres.2017.01.078

- Poon, K. T., & Wong, W. Y. (2021). Stuck on the train of ruminative thoughts: The effect of aggressive fantasy on subjective well-being. *Journal of interpersonal violence*, 36(11-12), 1-21. https://doi.org/10.1177/0886260518812796
- Purdon, C., & Clark, D. A. (1993). Obsessive intrusive thoughts in nonclinical subjects. Part I. Content and relation with depressive, anxious and obsessional symptoms.

 *Behaviour Research and Therapy, 31(8), 713-720.

 https://doi.org/https://doi.org/10.1016/0005-7967(93)90001-B
- Purdon, C., & Clark, D. A. (1994a). Obsessive intrusive thoughts in nonclinical subjects. Part II. Cognitive appraisal, emotional response and thought control strategies. *Behaviour Research and Therapy*, 32(4), 403-410. https://doi.org/doi.org/10.1016/0005-7967(94)90003-5
- Purdon, C., & Clark, D. A. (1994b). Perceived control and appraisal of obsessional intrusive thoughts: A replication and extension. *Behavioural and Cognitive Psychotherapy*, 22(4), 269-285. https://doi.org/doi.org/10.1017/S1352465800013163
- Purdon, C., & Clark, D. A. (1999). Metacognition and obsessions. *Clinical Psychology & Psychotherapy: An International Journal of Theory & Practice, 6*(2), 102-110. https://doi.org/10.1002/(SICI)1099-0879(199905)6:2<102::AID-CPP191>3.0.CO;2-5
- Purdon, C., & Clark, D. A. (2001). Suppression of obsession-like thoughts in nonclinical individuals: Impact on thought frequency, appraisal and mood state. *Behaviour Research and Therapy*, 39(10), 1163-1181. https://doi.org/10.1016/S0005-7967(00)00092-9
- Purdon, C., Cripps, E., Faull, M., Joseph, S., & Rowa, K. (2007). Development of a measure of egodystonicity. *Journal of Cognitive Psychotherapy*, 21(3), 198-216. https://doi.org/10.1891/088983907781494537
- Rachman, S. (1971). Obsessional ruminations. Behaviour Research and Therapy, 9, 229-235.

- Rachman, S. (1981). Part I. Unwanted intrusive cognitions. *Advances in Behaviour Research* and Therapy, 3(3), 89-99. https://doi.org/doi.org/10.1016/0146-6402(81)90007-2
- Rachman, S. (1997). A cognitive theory of obsessions. *Behaviour Research and Therapy*, 35(9), 793-802. https://doi.org/10.1016/S0005-7967(97)00040-5
- Rachman, S. (1998). A cognitive theory of obsessions: Elaborations. *Behaviour Research and Therapy*, 36(7), 385-401. https://doi.org/10.1016/S0005-7967(97)10041-9
- Rachman, S., & de Silva, P. (1978). Abnormal and normal obsessions. . *Behaviour Research and Therapy*, *16*(4), 233-248. https://doi.org/https://doi.org/10.1016/0005-7967(78)90022-0
- Rachman, S. J., & Hodgson, R. J. (1980). *Obsessions and Compulsions*. Englewood Cliffs: Prentice-Hall.
- Radomsky, A. S., Alcolado, G. M., Abramowitz, J. S., Alonso, P., Belloch, A., Bouvard, M., Clark, D. A., Coles, M. E., Doron, G., Fernández-Álvarez, H., Garcia-Soriano, G., Ghisi, M., Gomez, B., Inozu, M., Moulding, R., Shams, G., Sica, C., Simos, G., & Wong, W. (2014). Part 1—You can run but you can't hide: Intrusive thoughts on six continents. *Journal of Obsessive-Compulsive and Related Disorders*, *3*(3), 269-279. https://doi.org/10.1016/j.jocrd.2013.09.002
- Radomsky, A. S., Ashbaugh, A. R., & Gelfand, L. A. (2007). Relationships between anger, symptoms, and cognitive factors in OCD checkers. *Behavior Research and Therapy*, 45(11), 2712-2725. https://doi.org/10.1016/j.brat.2007.07.009
- Reynolds, M., & Salkovskis, P. M. (1992). Comparison of positive and negative intrusive thoughts and experimental investigation of the differential effects of mood. *Behaviour Research and Therapy*, 30(3), 273-281. https://doi.org/10.1016/0005-7967(92)90073-P

- Riskind, J. H., Ayers, C. R., & Wright, E. (2007). Simulated interpersonal provocation and fears of a loss of impulse control as determinants of aggressive mental intrusions. *Journal of Cognitive Psychotherapy*, 21(4), 285-294.
- Rokach, A. (1990). Content analysis of sexual fantasies of males and females. *The Journal of Psychology*, 124(4), 427-436. https://doi.org/10.1080/00223980.1990.10543238
- Rosenfeld, E., Huesmann, L. R., Eron, L. D., & Torney-Purta, J. V. (1982). Measuring patterns of fantasy behavior in children. *Journal of Personality and Social Psychology*, 42(2), 347-366.
- Rowa, K., & Purdon, C. (2003). Why Are Certain Intrusive Thoughts More Upsetting Than Others? *Behavioural and Cognitive Psychotherapy*, 31(1), 1-11. https://doi.org/10.1017/s1352465803001024
- Rowa, K., Purdon, C., Summerfeldt, L. J., & Antony, M. M. (2005). Why are some obsessions more upsetting than others? *Behavor Research and Therapy*, 43(11), 1453-1465. https://doi.org/10.1016/j.brat.2004.11.003
- Salkovskis, P. M. (1985). Obsessional-compulsive problems: A cognitive-behavioural analysis. *Behaviour Research and Therapy*, *23*(5), 571-583. https://doi.org/https://doi.org/10.1016/0005-7967(85)90105-6
- Salkovskis, P. M. (1989). Cognitive-behavioural factors and the persistence of intrusive thoughts in obsessional problems. *Behaviour Research and Therapy*, *27*(6), 677-682. https://doi.org/https://doi.org/10.1016/0005-7967(89)90152-6
- Shafran, R., & Rachman, S. (2004). Thought-action fusion: a review. *J Behav Ther Exp Psychiatry*, 35(2), 87-107. https://doi.org/10.1016/j.jbtep.2004.04.002
- Shafran, R., Teachman, B. A., Kerry, S., & Rachman, S. (1999). A cognitive distortion associated with eating disorders: Thought-shape fusion. *British Journal of Clinical Psychology*, 38(2), 167-179. https://doi.org/10.1348/014466599162728

- Sheldon, K., & Patel, G. (2009). Fantasies of a dangerous and severe personality disordered clinical forensic population: An exploratory study Stockholm.
- Shingler, J. (2009). Managing intrusive risky thoughts: What works? *Journal of Sexual Aggression*, 15(1), 39-53. https://doi.org/10.1080/13552600802542011
- Sinclair, S. J., Siefert, C. J., Slavin-Mulford, J. M., Stein, M. B., Renna, M., & Blais, M. A. (2012). Psychometric evaluation and normative data for the depression, anxiety, and stress scales-21 (DASS-21) in a nonclinical sample of U.S. adults. *Evaluation and the Health Professions*, 35(3), 259-279. https://doi.org/10.1177/0163278711424282
- Stöber, J. (2001). The Social Desirability Scale-17 (SDS-17): Convergent validity, discriminant validity, and relationship with age. *European Journal of Psychological Assessment*, 17(3), 222-232.
- Stokes, A., Poindexter, M., Bell, K., & Mellman, T. A. (2022). Strategies for Controlling
 Unwanted Intrusive Thoughts and Insomnia Severity in Urban-Residing Young Adult
 African Americans. *Behav Sleep Med*, 1-8.

 https://doi.org/10.1080/15402002.2022.2057986
- Sukhodolsky, D. G., Golub, A., & Cromwell, E. N. (2001). Development and validation of the anger rumination scale. *Personality and Individual Differences*, *31*(5), 689-700. https://doi.org/10.1016/S0191-8869(00)00171-9
- Sukhodolsky, D. G., & Ruchkin, V. V. (2004). Association of normative beliefs and anger with aggression and antisocial behavior in Russian male juvenile offenders and high school students. *Journal of Abnormal Child Psychology*, 32(2), 225-236.
- Veale, D., Freeston, M., Krebs, G., Heyman, I., & Salkovskis, P. (2009). Risk assessment and management in obsessive—compulsive disorder. *Advances in Psychiatric Treatment*, 15(5), 332-343. https://doi.org/10.1192/apt.bp.107.004705

- Vella-Zarb, R. A., Cohen, J. N., McCabe, R. E., & Rowa, K. . (2017). Differentiating sexual thoughts in obsessive-compulsive disorder from paraphilias and nonparaphilic sexual disorders. *Cognitive and Behavioral Practice*, 24(3), 342-352. https://doi.org/10.1016/j.cbpra.2016.06.007
- Wegner, D. M., Schneider, D. J., Carter, S. R., & White, T. L. (1987). Paradoxical effects of thought suppression. *Journal of Personality and Social Psychology*, *53*(1), 5-13. https://doi.org/10.1037/0022-3514.53.1.5
- Wells, A., & Davies, M. I. (1994). The Thought Control Questionnaire: A measure of individual differences in the control of unwanted thoughts. *Behaviour Research and Therapy*, 32(8), 871-878.
- Wheaton, M. G., Abramowitz, J. S., Berman, N. C., Riemann, B. C., & Hale, L. R. (2010). The relationship between obsessive beliefs and symptom dimensions in obsessive-compulsive disorder. *Behav Res Ther*, 48(10), 949-954. https://doi.org/10.1016/j.brat.2010.05.027
- Whiteside, S. P., & Abramowitz, J. S. (2005). The expression of anger and its relationship to symptoms and cognitions in obsessive-compulsive disorder. *Depression and Anxiety*, 21(3), 106-111. https://doi.org/10.1002/da.20066
- Whittal, M. L., Thordarson, D. S., & McLean, P. D. (2005). Treatment of obsessive-compulsive disorder: cognitive behavior therapy vs. exposure and response prevention. *Behav Res Ther*, 43(12), 1559-1576. https://doi.org/10.1016/j.brat.2004.11.012
- Williams, M. T., Farris, S. G., Turkheimer, E., Pinto, A., Ozanick, K., Franklin, M. E., Liebowitz, M., Simpson, H. B., & Foa, E. B. (2011). Myth of the pure obsessional type in obsessive--compulsive disorder. *Depress Anxiety*, 28(6), 495-500. https://doi.org/10.1002/da.20820

- Wong, S. F., Aardema, F., Mendel, B., Trespalacios, F., & Radomsky, A. S. (2020).

 Manipulating feared self-perceptions in an analogue sample using virtual reality and its impact on harm-related intrusions and urges to neutralise. *Journal of Obsessive-Compulsive and Related Disorders*, 27. https://doi.org/10.1016/j.jocrd.2020.100585
- Yardley, L., & Bishop, F. (2007). Mixing qualitative and quantitative methods: A pragmatic approach. In *The Sage handbook of qualitative research in psychology* (Rogers ed., pp. 352-454). Sage.

APPENDICES

Appendix A

Empirical Study One: Swinburne Research Ethics Committee Approval Certificate

Swinburne University of Technology Human Research Ethics Committee Approval certificate



The ethics application for your project Exploring Phenomena of Intrusive Thoughts and Aggressive Thoughts: Associated Beliefs and Featureshas been approved.

Chief Investigator: A/Prof. Maja Nedeljkovic

Ref: 20190386-1937

Approved Duration: 08/11/2019 to 08/11/2022

I refer to the ethical review of the above project protocol by Swinburne's Human Research Ethics Committee (SUHREC) or its sub-committees.

I am pleased to advise that, as submitted to date, the project may proceed in line with standard on-going ethics clearance conditions outlined below.

- The approved duration is as shown above unless an extension request is subsequently approved.
- All human research activity undertaken under Swinburne auspices must conform to Swinburne and external regulatory standards, including the National Statement on Ethical Conduct in Human Research (2018) and with respect to secure data use, retention and disposal.
- The named Swinburne Chief Investigator/Supervisor remains responsible for any personnel appointed to or associated with the project being made aware of ethics
 clearance conditions, including research and consent procedures or instruments approved. Any change in Chief Investigator/Supervisor, and addition or removal of other
 personnel/students from the project, requires timely notification and SUHREC endorsement.
- The above project has been approved as submitted for ethical review by or on behalf of SUHREC. Amendments to approved procedures or instruments ordinarily require
 prior ethical appraisal/clearance from SUHREC for approval. SUHREC must be notified immediately or as soon as possible thereafter of (a) any serious or unexpected
 adverse effects on participants and any redress measures; (b) proposed changes in protocols; and (c) unforeseen events which might affect continued ethical
 acceptability of the project.
- · At a minimum, an annual report on the progress of the project is required as well as at the conclusion (or abandonment) of the project.
- A duly authorised external or internal audit of the project may be undertaken at any time.
- · Please forward this approval certificate to relevant members of the project team.

The following investigators have been approved to work on the project:

Chief Investigator
Maja Nedeljkovic
Associate Investigators
Michael Daffern, Richard Moulding
Student Investigators
Stephanie Fernandez
Please contact the Swinburne Research Ethics Office if you have any queries.
Regards,
Dr Astrid Nordmann

on behalf of

Research Ethics Office

Swinburne University of Technology

P: +61 3 9214 3845 | E: resethics@swin.edu.au

Thursday, March 2, 2023 at 8:12:38 AM Australian Eastern Daylight Time

Subject: Ethics clearance: Exploring Phenomena of Intrusive Thoughts and Aggressive Thoughts:

Associated Beliefs and Features

Date: Friday, 8 November 2019 at 2:25:00 pm Australian Eastern Daylight Time

From: donotreply@infonetica.net
To: mnedelkjovic@swin.edu.au

CC: RES Ethics, STEPHANIE FERNANDEZ, Michael Daffern, Richard Moulding

Attachments: Letter.pdf

Dear All,

Ref: 20190386-1937 : Exploring Phenomena of Intrusive Thoughts and Aggressive Thoughts: Associated Beliefs and Features

Your ethics application has been approved. Please see the attachment for details of the approval.

Please contact the Swinburne Research Ethics Office if you have any queries.

Regards,

Dr Astrid Nordmann

Research Ethics Office

Swinburne University of Technology

P: +61 3 9214 3845 | E: resethics@swin.edu.au

Appendix B

Empirical Study Two: Swinburne Research Ethics Committee Approval Certificate

Swinburne University of Technology Human Research Ethics Committee Approval certificate



30/06/202

The ethics application for your project Exploring Phenomena of Aggressive Intrusive Thoughts and Aggressive Scripts - Associated Beliefs and Features has been approved.

Chief investigator: Maja Nedeljkovic

Ref. 20215556-7961

Approved Duration: 30/06/2021 to 30/06/2023

I refer to the ethical review of the above project protocol by Swinburne's Human Research Ethics Committee (SUHREC) or its sub-committees.

I am pleased to advise that, as submitted to date, the project may proceed in line with standard on-going ethics clearance conditions outlined below.

- The approved duration is as shown above unless an extension request is subsequently approved.
- All human research activity undertaken under Swinburne auspices must conform to Swinburne and external regulatory standards, including the National Statement on Ethical Conduct in Human Research (2018) and with respect to secure data use, retention and disposal.
- The named Swinburne Chief Investigator/Supervisor remains responsible for any personnel appointed to or associated with the project being made aware of ethics
 clearance conditions, including research and consent procedures or instruments approved. Any change in Chief Investigator/Supervisor, and addition or removal of other
 personnel/students from the project, requires timely notification and SUHREC endorsement.
- The above project has been approved as submitted for ethical review by or on behalf of SUHREC. Amendments to approved procedures or instruments ordinarily require
 prior ethical appraisal/clearance from SUHREC for approval. SUHREC must be notified immediately or as soon as possible thereafter of (a) any serious or unexpected
 adverse effects on participants and any redress measures; (b) proposed changes in protocols; and (c) unforeseen events which might affect continued ethical
 acceptability of the project.
- · At a minimum, an annual report on the progress of the project is required as well as at the conclusion (or abandonment) of the project.
- A duly authorised external or internal audit of the project may be undertaken at any time.
- · Please forward this approval certificate to relevant members of the project team.

This research project was approved during COVID-19 restrictions. The conduct of the research during this period should reflect any changes in relation to university and government COVID-19 mandates in the relevant jurisdictions. To ensure you have accommodated these mandates please refer to the Swinburne Ethics COVID-19 website here.

Emics COVID-19 website nere.
The following investigators have been approved to work on the project:
Chief Investigator
Mala Mahalikevile

Associate Investigators

Richard Moulding, Michael Daffern

Student Investigators

Stephanie Fernandez

Please contact the Swinburne Research Ethics Office If you have any queries.

Regards,

Ms I eah Barham

on behalf of

Research Ethics Office

Swinburne University of Technology

P: +61 3 9214 8145 | E: resethics@swin.edu.au

Subject: Ethics clearance: Exploring Phenomena of Aggressive Intrusive Thoughts and Aggressive

Scripts - Associated Beliefs and Features

Date: Wednesday, 30 June 2021 at 5:46:04 pm Australian Eastern Standard Time

From: donotreply@infonetica.net

To: Maja Nedeljkovic

CC: RES Ethics, STEPHANIE FERNANDEZ, richard.moulding@cairnmillar.edu.au, Michael

Daffern

Attachments: Letter.pdf

Dear All,

Ref: 20215556-7961 : Exploring Phenomena of Aggressive Intrusive Thoughts and Aggressive Scripts - Associated Beliefs and Features

Your ethics application has been approved. Please see the attachment for details of the approval.

Please contact the Swinburne Research Ethics Office if you have any queries.

Regards,

Ms Leah Barham

Research Ethics Office

Swinburne University of Technology

P: +61 3 9214 8145 | E: resethics@swin.edu.au

Appendix C

Empirical Study Two: Forensicare Operational Research Committee



14 July 2021

A/Prof. Maja Nedeljkovic, Swinburne University ABN 32 807 323 885 Yarra Bend Road Fairfield Victoria 3078

T+61 3 9495 9100 F+61 3 9495 9190

E info@forensicare.vic.gov.au www.forensicare.vic.gov.au

Dear A/Prof. Maja Nedeljkovic,

Re: Exploring Phenomena of Aggressive Intrusive Thoughts and Aggressive Scripts: The Influence of Associated Beliefs and Features

The Forensicare Operational Research Committee has completed a full review of your research application. The Committee has given operational approval for your research to be conducted at Forensicare. This approval is subject to the following:

- Provision of evidence of approval by the Swinburne University Human Research Ethics Committee. Please forward a copy of the approval letter from the Research Ethics Committee as soon as it is received. You may not commence the research until evidence of ethical approval has been provided.
- Approval is given for the period between the anticipated commencement and completion dates as set out in the documentation. If the study has not been completed by the nominated completion date, an application for extension will be required.
- To enable the Committee to meet its obligations in relation to monitoring Forensicare's research program, you are required to provide brief annual progress reports upon request each year the study is in progress to cover the previous financial year (regardless of project start date). A Final Report and lay summary of findings must be provided within 60 days of the completion of the project.
- Forensicare is committed to the dissemination and translation of research outcomes. As such, researchers must make reasonable efforts to engage in appropriate dissemination activities upon request. This includes the preparation of a brief report, or presentation of the study findings to Forensicare staff or consumers.

Failure to comply will lead to the withdrawal of approval and suspension of the research project. In addition, any adverse events need to be notified promptly to the Operational Research Committee as well as the approving Human Research Ethics Committee. Please ensure that the Operational Research Committee is notified of any matter that arises that may affect the conduct of the approved program.

Appendix D

Study one: Advertising material for students



PARTICIPANTS NEEDED



Project Title: Exploring Phenomena of Intrusive Aggressive Thoughts and Aggressive Thoughts: Associated Beliefs and Features

Researchers at Swinburne University of Technology are investigating the experience of intrusive aggressive thoughts, and their associated beliefs. Most individuals have experienced an unpleasant or disturbing thought during their life and sometimes these thoughts occur spontaneously and without provocation. This common experience is said to occur in approximately 93% of individuals at some point in their life, and is referred to as an intrusive thought. By examining these <u>areas</u> it will increase our understanding of intrusive thoughts and how belief systems may be implicated in the process of aggressive thinking.

What does my participation require?

You will be asked to respond to several questionnaires asking about aggressive intrusive thoughts, beliefs about yourself, others and the world, and your attitudes towards aggression and violence. The questionnaire will take approximately 45 minutes to complete.

You are able to take part in the research project if you:

Are 18+ years of age

If you are a Swinburne Psychology student, you are able to receive 1 course credit for your participation.

If you are interested in participating please follow the link:

Stephanie Fernandez

Email: sfernanez@swin.edu.au

Appendix E

Study one: Advertising material for social media



PARTICIPANTS NEEDED



Project Title: Exploring Phenomena of Intrusive
Aggressive Thoughts and Aggressive Thoughts:
Associated Beliefs and Features

Researchers at Swinburne University of Technology are investigating the experience of intrusive aggressive thoughts, and their associated beliefs. Most individuals have experienced an unpleasant or disturbing thought during their life and sometimes these thoughts occur spontaneously and without provocation. This common experience is said to occur in approximately 93% of individuals at some point in their life, and is referred to as an intrusive thought. By examining these areas it will increase our understanding of intrusive thoughts and how belief systems may be implicated in the process of aggressive thinking.

What does my participation require?

You will be asked to respond to several questionnaires asking about aggressive intrusive thoughts, beliefs about yourself, others and the world, and your attitudes towards aggression and violence. The questionnaire will take approximately 45 minutes to complete.

You are able to take part in the research project if you:

- Are 18+ years of age
- Currently live in Australia

Your participation can put you in the draw to win 1 of 4 \$100 voucher (Coles)

If you are interested in participating please follow the link:

https://swinuw.au1.qualtrics.com/jfe/form/SV_3q3gWX6ym6YzZHv

Stephanie Fernandez

Appendix F Consent Information Statement



Project Title: Exploring Phenomena of Intrusive Aggressive

Thoughts and Aggressive Thoughts: Associated Beliefs and

Investigator(s): Associate Professor Maja Nedeljkovic, (Chief Investigator, supervisor)

Professor Michael Daffern, (Secondary Supervisor)

Dr Richard Moulding, (Co-supervisor)

Miss Stephanie Fernandez, (Student Investigator)

Introduction to Project and Invitation to Participate

This Consent Information Statement provides a detailed background about the research project. The purpose of this statement is to explain to you as openly and clearly as possible the procedures involved in this project before you decide whether or not you would like to participate in it.

Once you have understood what the survey will involve and if you agree to take part in it, you will be asked to complete a survey. Proceeding with the study indicates that you understand the information about the project and you give consent to be involved in the project. You are free to withdraw from the project at any time.

What is the Project About and Why is it being Undertaken?

Most individuals have experienced an unpleasant or disturbing thought during their life and sometimes these thoughts occurs spontaneously and without provocation. This common experience is said to occur in approximately 93% of individuals at some point in their life, and is referred to as an intrusive thought. Of particular interest to the current study are intrusive aggressive thoughts and how they are experienced by the individual. The present study aims to explore how aggressive intrusive thoughts, and aggressive thoughts more generally, are experienced by the general population. Additionally, we aim to examine how certain beliefs individuals have about themselves, others, and aggression may relate to the experience of intrusive aggressive thoughts. By examining these areas it will increase our understanding of intrusive thoughts and how belief systems may be implicated in the process of aggressive thinking.

Project and Researcher Interests

The project is a partial requirement of the Doctor of Psychology (Clinical and Forensic) program for Stephanie Fernandez.

What Participation will Involve

In this study you will asked to complete a battery of self-report questionnaires. The questionnaires will relate to topics such as aggressive intrusive thoughts and your beliefs, feelings, and reactions towards these. Following this, you will be asked questions relating to your beliefs about aggression and violence, and also your experience with aggressive-type behaviours. The questionnaire will take approximately 45 minutes to complete. Please note that some of the questions asked during this project may be of a sensitive nature (e.g., description of aggressive intrusive thoughts; aggressive-type attitudes and behaviours). A pop-up notification will be provided to you before sensitive questions are

presented. If you are uncomfortable with these types of questions you may choose to skip them, you can withdraw from the study at any time. Your responses to the questions will be recorded anonymously, and thus will be untraceable to you. While this study will be asking you about symptoms of OCD, Anxiety and Depression, we are seeking participants without a diagnosis of a mental illness. If you have a mental illness and wish to participate, you are free to do so, but participation may be found confronting. It is also important to note that we cannot provide diagnosis on the basis of these questions, and if you have concerns about experiencing any of these symptoms you should contact your relevant health professional (e.g., general practitioner, psychologist). Also, some of the questions may be particularly confronting for individuals who have experienced violence or trauma. If you believe that your participation may cause significant distress you may choose not to participate.

Participant rights and interests

Although you may not experience any direct benefits from having taken part in the study, the results that you contribute towards will help us in understanding aggressive intrusive thoughts and their subjective experience.

<u>vou can withdraw at any time.</u> If you do not wish to take part in this study, you have no obligations to do so. You can choose to not answer questions if you are uncomfortable answering. If you decide to take part in the project and later change your mind, you are free to withdraw from the project at any time. Your decision of whether to take part or not, and the withdraw, will not affect your relationship with Swinburne University of Technology.

Additionally, if participation in the project is causing any distress or discomfort please note that you can stop participation at any time without providing reason or explanation.

Support services and facilities

If you are experiencing a crisis, cannot contact a counsellor and need help urgently phone Lifeline on 131 114 or the Suicide Help Line on 1300 651 251. For international support please visit: www.befrienders.org

Appointments for counselling are also available at the Swinburne Psychology Clinic for a low cost at the Hawthorn campus George Swinburne (GS) Building, 34 Wakefield Street, Level 4. Phone: (03) 9214 8653

If you are a Swinburne Student, counselling and psychological services are available free of charge via Student Services, Health and Wellbeing Clinic. Phone: (03) 9214 8483

Research output

The results derived from this project will be analysed and summarised in the Doctoral thesis of Stephanie Fernandez. It is hoped that the present study will be published in a journal and presented at national/international conferences. The anonymous group data may be used in future research on aggressive intrusive thoughts conducted by the researchers. At no point will individual responses be identifiable.

Data Management

Data from the study will be stored in an electronic format and stored on a password protected, secure computer only accessible by the researchers named on this application

If you would like further information about the project:

Please do not hesitate to contact:

Associate Professor Maja Nedeljkovic, Clinical Psychologist and Lecturer Department of Psychological Sciences, Swinburne University of Technology mnedeljkovic@swin.edu.au, Tel No(s): (03) 9214 4428

Concerns/complaints about the project – who to contact:

This project has been approved by or on behalf of Swinburne's Human Research

Ethics Committee (SUHREC) in line with the *National Statement on Ethical Conduct*in Human Research. If you have any concerns or complaints about the conduct of this
project, you can contact:

Research Ethics Officer, Swinburne Research (H68),
Swinburne University of Technology, P O Box 218, HAWTHORN VIC 3122.

Tel (03) 9214 3845 or +61 3 9214 3845 or resethics@swin.edu.au

Appendix G

Debriefing Statement



DEBRIEFING STATEMENT

Project Title: Exploring Intrusive Aggressive Thoughts, and their

Associated Beliefs and Features

Investigator(s):

Associate Professor Maja Nedeljkovic, (Chief Investigator,

supervisor)

Professor Michael Daffern, (Secondary Supervisor)

Dr Richard Moulding, (Co-supervisor)

Miss Stephanie Fernandez, (Student Investigator)

Thank you for your participation. This sheet contains more detailed information about the purpose of the study and what we hope to achieve.

The present study aims to explore how aggressive intrusive thoughts, and aggressive thoughts more generally, are experienced by the general population. The independent variables are aggressive thoughts and beliefs, and the dependent variable are the symptoms of Obsessive Compulsive Disorder. We are specifically interested in examining how certain beliefs individual have about themselves, others, and aggression may relate to the experience of intrusive aggressive thoughts.

The information gathered from this study will help us gain better understanding of the factors that may influence aggressive intrusive thoughts, and how belief systems may be implicated

in the process of aggressive thinking If you would like further information about the project:

Please do not hesitate to contact:

Associate Professor Maja Nedeljkovic, Clinical Psychologist and Lecturer Department of Psychological Sciences, Swinburne University of Technology mnedeljkovic@swin.edu.au, Tel No(s): (03) 9214 4428

Support services and facilities

If you are experiencing a crisis, cannot contact a counsellor and need help urgently phone Lifeline on 131 114 or the Suicide Help Line on 1300 651 251. For international support please visit: www.befrienders.org

Appointments for counselling are also available at the Swinburne Psychology Clinic for a low cost at the Hawthorn campus George Swinburne (GS) Building, 34 Wakefield Street, Level 4. Phone: (03) 9214 8653

If you are a Swinburne Student, counselling and psychological services are available free of charge via Student Services, Health and Wellbeing Clinic. Phone: (03) 9214 8483

Concerns/complaints about the project – who to contact:

This project has been approved by or on behalf of Swinburne's Human Research

Ethics Committee (SUHREC) in line with the *National Statement on Ethical Conduct*in Human Research. If you have any concerns or complaints about the conduct of this
project, you can contact:

Research Ethics Officer, Swinburne Research (H68),

Swinburne University of Technology, P O Box 218, HAWTHORN VIC 3122.

Tel (03) 9214 3845 or +61 3 9214 3845 or resethics@swin.edu.au

Appendix H

Study two: advertising material for OCD sample



PARTICIPANTS WITH A DIAGNOSIS OF OCD NEEDED



Project Title: Exploring Phenomena of Intrusive Aggressive Thoughts and Aggressive Thoughts: Associated Beliefs and Features

Researchers at Swinburne University of Technology are investigating the experience of intrusive aggressive thoughts. Most individuals have experienced an unpleasant or disturbing thought during their life and sometimes these thoughts occur spontaneously and without provocation. The current study is interested in examining individuals' subjective experience associated with aggressive intrusive thoughts and aggressive thinking, exploring the role associated beliefs have on experience of these thoughts. By examining these experiences we hope to increase our understanding of intrusive thoughts and how beliefs about these experiences may be implicated in the process of aggressive thinking.

What does my participation require?

You will be asked to complete several questionnaires asking about your experience with aggressive thoughts. You will also be asked about the beliefs you hold about yourself, others and the world, and your attitudes towards aggression and violence.

- An interview will be conducted to further explore your experience of aggressive intrusive thoughts, and aggressive thoughts more generally
- The interview will take approximately 2 hours to complete

Am I eligible to participate?

To be eligible for research participation, you will: (a) be aged 18+, (b) have a diagnosis of OCD, (c) do not have a past or present experience of psychosis, (d) identify as either male, female, or gender non-specific, (e) have the capacity to provide informed consent, (f) have English skills that will allow for conversation during interviews.

For participating you will be offered a \$30 Coles Voucher

If you are interested in participating, please provide your details using the following link:

tinyurl.com/kw6krfmw

Stephanie Fernandez will be in contact with you to discuss the study

Email: sfernandez@swin.edu.au

Appendix I

Empirical Study Two: advertising material for CFMHS sample



PARTICIPANTS NEEDED FOR A STUDY



Project Title: Exploring Phenomena of Intrusive Aggressive Thoughts and Aggressive Thoughts: Associated Beliefs and Features

Researchers at Swinburne University of Technology are investigating the experience of intrusive aggressive thoughts. Most individuals have experienced an unpleasant or disturbing thought during their life and sometimes these thoughts occur spontaneously and without provocation. The current study is interested in examining individuals' subjective experience associated with aggressive intrusive thoughts and aggressive thinking, exploring the role associated beliefs have on experience of these thoughts. By examining these thoughts we hope to increase our understanding of intrusive thoughts and how beliefs about these experiences may be implicated in the process of aggressive thinking.

What does my participation require?

You will be asked to complete several questionnaires asking about your experience with aggressive thoughts. You will also be asked about the beliefs you hold about yourself, others and the world, and your attitudes towards aggression and violence.

- An interview will be conducted to further explore your experience of aggressive intrusive thoughts, and aggressive thoughts more generally
- · The interview will take approximately 2 hours to complete

Am I eligible to participate?

To be eligible for research participation, you will: (a) be aged 18+, (b) do not have a past or present experience of psychosis, (c) identify as either male, female, or gender non-specific, (d) have the capacity to provide informed consent, (e) have English skills that will allow for conversation during interviews.

For participating you will be offered a \$30 Coles Voucher

If you are interested in participating, please provide your details using the following link:

tinyurl.com/kw6krfmw

Stephanie Fernandez will be in contact with you to discuss the study

Email: sfernandez@swin.edu.au

Appendix J

Empirical Study Two: Explanatory Statement







Project Title: Exploring Phenomena of Intrusive Aggressive Thoughts and Aggressive Thoughts: Associated Beliefs and Features

Investigator(s): Associate Professor Maja Nedeljkovic, (Chief Investigator, supervisor)

Professor Michael Daffern, (Secondary Supervisor)

Dr Richard Moulding, (Co-supervisor)

Miss Stephanie Fernandez, (Student Investigator)

Introduction to Project and Invitation to Participate

This Information Statement provides a detailed background about the research project. The purpose of this statement is to explain to you as openly and clearly as possible the procedures involved in this project before you decide whether or not you would like to participate in it. Once you have understood what the survey will involve and if you agree to take part in it, you will be asked to complete an interview with the researcher. Proceeding with the study indicates that you understand the information about the project and you give consent to be involved in the project. You are free to withdraw from the project at any time. You will be offered a \$30 Coles voucher for your participation in the study.

Consent process

As part of the project recruitment process, your verbal consent to be contacted by the student researcher may have been obtained by your treating clinician. Your treating clinician may have passed on your contact details with your consent, to the student researcher.

It is important to understand that your consent to be contacted by the student researcher is not your consent to participate in the project, and you are under no obligation to participate should you decide that the project is of no interest to you.

Am I eligible to participate?

To be eligible for research participation, you will: (a) be aged 18+, (b) do not have a past or present experience of psychosis, (c) identify as either male, female, or gender non-specific, (d) have the capacity to provide informed consent, (e) have English skills that will allow for conversation during interviews.

What is the Project About and Why is it being Undertaken?

Most individuals have experienced an unpleasant or disturbing thought during their life and sometimes these thoughts occurs spontaneously and without provocation. This common experience is said to occur in approximately 93% of individuals at some point in their life, and is referred to as an intrusive thought. Of particular interest to the current study are intrusive aggressive thoughts and how they are experienced by the individual. The present study aims to explore how aggressive intrusive thoughts, and aggressive thoughts more generally, are experienced by individuals who have a history of aggression or violence difficulties. Additionally,







we aim to examine how certain beliefs individuals have about themselves, others, and aggression may relate to the experience of intrusive aggressive thoughts. By examining these thoughts we hope to increase our understanding of intrusive thoughts and how belief systems may be implicated in the process of aggressive thinking.

Project and Researcher Interests

The project is a partial requirement of the Doctor of Psychology (Clinical and Forensic) program for Stephanie

What Participation will involve

In this study you will asked to complete a battery of self-report questionnaires and participate in an interview further exploring your experiences. The questionnaires will relate to topics such as aggressive intrusive thoughts and your beliefs, feelings, and reactions towards these. You will be asked questions relating to your beliefs about aggression and violence, and also your experience with aggressive-type behaviours. The questionnaire and interview will take approximately 2 hours to complete. To facilitate data collection for the student researcher, the interviews will be audio recorded. Please note that some of the questions asked during this project may regarded by you as sensitive (e.g., description of aggressive intrusive thoughts; aggressive-type attitudes and behaviours). If you are uncomfortable with these types of questions you may choose to skip them by letting the researcher know.

It is important to understand that your participation in this project is voluntary and you can withdraw at any time prior to and during the interview and data collection phase of the study. If you do not wish to take part in this study, you have no obligations to do so. You can choose to not answer any question or withdraw any data up until such data is published. Your decision of whether to take part or not, or to withdraw, will not affect your relationship with Swinburne University of Technology, Forensicare, or your treating clinician.

Additionally, if participation in the project is causing any distress or discomfort please note that you can stop participation at any time without providing reason or explanation.

Risks & Benefits to taking part in the study

Although you may not experience any direct benefits from having taken part in the study, the results that you contribute towards will help us in understanding aggressive intrusive thoughts and their subjective experience. There are only very small risks to taking part in the study. The questions asked of you in the research interview may be regarded as sensitive and have potential to cause you some discomfort. If you are uncomfortable with these types of questions you may choose to skip them or stop the interview by letting the researcher know. We can also assist in referring you to professional services for any problems you may experience due to taking part in the study. Given the sensitivity of the information you are providing, there is a small risk to your privacy in providing us with personal information. In order to minimise this risk, we will not record your name with any of the information you provide. We will also check all the information in any reports produced to make sure that you cannot be identified by the information provided.

Support services and facilities







While this study will be asking you about symptoms of OCD, Anxiety and Depression we cannot provide you with a diagnosis on the basis of the questions in this interview. If you have concerns about experiencing any of these symptoms you should contact your relevant health professional (e.g., general practitioner, psychologist). Also, some of the questions may be challenging for people who have experienced violence or trauma. If you believe that your participation may cause significant distress you may choose not to participate. If you take part in in this study we will provide you with a list of services to contact if you are feeling any distress at the end of the interview or in the weeks following. You can also tell the person conducting the interview know if you feel any discomfort and they will assist you to find a professional to speak with. If you are experiencing a crisis, cannot contact a counsellor and need help urgently phone Lifeline on 131 114 or the Suicide Help Line on 1300 651 251. For international support please visit: www.befrienders.org Appointments for counselling are also available at the Swinburne Psychology Clinic for a low cost at the Hawthorn campus George Swinburne (GS) Building, 34 Wakefield Street, Level 4. Phone: (03) 9214 8653 If you are a Swinburne Student, counselling and psychological services are available free of charge via Student Services, Health and Wellbeing Clinic. Phone: (03) 9214 8483

Research output

The results derived from this project will be analysed and summarised in the Doctoral thesis of Stephanie Fernandez. It is hoped that the present study will be published in a journal and presented at national/international conferences. You may also wish to receive a copy of summary report of the findings of the project. In order to receive this, you will be required to provide an email or postal address to have the summary report distributed to you.

Data Management

Data from the study will be stored in both paper and electronic formats. Paper responses will be filed in secure cabinets, and electronic formats will be stored on a password protected, secure computer only accessible by the researchers named on this application. Your data collected from the interviews with the student researcher will be in a de-identified format. The student researcher will assign a unique code to your data, and this unique code will be used in further data analyses. Should you wish to withdraw from the study at any point, your data will be destroyed and not used in the publication of the results. Your privacy will be protected by ensuring all your de-identified data is stored on a password protected folder and computer. The anonymous group data may be used in future research on aggressive intrusive thoughts conducted by the researchers. However, data from the study will be kept for a maximum of 5 year after any publication or published outcome.

Appendix K

Empirical Study Two: Statement of Informed Consent to Participate







Project Title: Exploring Phenomena of Intrusive Aggressive Thoughts and Aggressive Thoughts: Associated Beliefs and Features

Investigator(s): Associate Professor Maja Nedeljkovic, Professor Michael Daffern, Dr Richard Moulding, Miss Stephanie Fernandez

If you wish to take part in this project, please email the Student Researcher (Stephanie) using the details below.

Email: sfernandez@swin.edu.au

Consent form

I consent to participate in the project named above. I have been provided a copy of the project consent information statement to which this consent form relates and any questions I have asked have been answered to my satisfaction.

In relation to this project, please circle your response to the following:

•	I agree to be interviewed by the researcher	Yes	No
•	I agree to allow the interview to be recorded by electronic device	Yes	No
•	l agree to complete questionnaires asking me about clinical symptoms and	Yes	No

I acknowledge that:

- (a) my participation is voluntary and that I am free to withdraw from the project at any time without explanation;
- (b) the Swinburne project is for the purpose of research and not for profit;
- (c) any identifiable information about me which is gathered in the course of and as the result of my participating in this project will be (i) collected and retained for the purpose of this project and (ii) accessed and analysed by the researcher(s) for the purpose of conducting this project;
- (d) I understand the length of time researcher/s will have access to this information;
- (e) my anonymity is preserved and I will not be identified in publications or otherwise without my express written consent.

By signing this document I agree to participate in this project.

Name of Participant:	
•	
Signature & Date:	

Appendix L

Social Desirability Scale (SDS; Stöber, 2001)

Below you will find a list of statements. Please read each statement carefully and decide if that statement describes you or not. If it describes you, check the word "true"; if not, check the word "false."

- 1. I sometimes litter
- 2. I always admit my mistakes openly and face the potential negative consequences
- 3. In traffic I am always polite and considerate of others
- 4. I have tried illegal drugs (for example, marijuana, cocaine, etc.)
- 5. I always accept others' opinions, even when they don't agree with my own
- 6. I take out my bad moods on others now and then.
- 7. There has been an occasion when I took advantage of someone else.
- 8. In conversations I always listen attentively and let others finish their sentences.
- 9. I never hesitate to help someone in case of emergency
- 10. When I have made a promise, I keep it no ifs, ands or buts.
- 11. I occasionally speak badly of others behind their back
- 12. I would never live off other people
- 13. I always stay friendly and courteous with other people, even when I am stressed out.
- 14. During arguments I always stay objective and matter-of-fact.
- 15. There has been at least one occasion when I failed to return an item that I borrowed.
- 16. I always eat a healthy diet.
- 17. Sometimes I only help because I expect something in return.

Appendix M

Schedule of Imagined Violence (SIV; Grisso et al., 2000)

Have you ever had daydreams or thoughts about physically hurting or injuring some other person? YES/NO

1. How often do you have thoughts about hurting or injuring other people?	
— several times a day	
— once a day	
— several times a week	
— once a week	
— several times a month (less than once a week)	
— several times a year	
— never	
1a) In what form or forms (i.e., as a thought, image, or impulse, daydream) do you	
usually have these thoughts? An image is like a photograph that appear in our minds, an	nd
an impulse is like a sudden urge to do or say something	
— Thought	
— Image	
— Impulse	
— Daydream	
2. When was the last time you had such a thought?	
— Today	
— In the past 2 days	
— In the past 3-7 days	
— During the past 2 months	
— During the past month	
— More than 2 months ago	
3. When did you start having these thoughts?	
— Since a specific event (specify event)	
— As long as can remember	
— During the past month	
— Since several months ago	
— More than a year ago	
— During the past 3-6 months	
— Since several years ago	

4. when you have these thoughts, in	what way do you think about behaving aggressively?
Please indicate below which option(s) relate to you.
Verbally abusing/swearing	gStabbing
Hitting/punching	Damaging property
Slapping	Throwing an object
Kicking	Threatening
Shooting	Torturing (e.g., burning)
Sexual aggression	Other (list)
5. When you have these thoughts, ar	e they usually about the same person, or might they be
about many different people?	
Same	Different
6. Are they usually about the same p Same person Description	erson, or might they be about many different people?
	F F F F
·	ving these thoughts, have the injuries that you think about
	or have they been about the same?
Less serious Same	More serious
8. In the past 2 months, have you even	er had these thoughts while actually being with or
watching the person whom you imag	ine hurting?
Yes	No
9. How often do you have thoughts v	where you are planning to behave aggressively?
Never (discontinue) Several times a year	Once a week Several times a week Once a day

Several ti	mes a month (less than or	nce a week) Several	times a d	ay
10. When do you mo	st think about behaving ag	ggressively? (X as many	as apply)	
After behaving a While behaving a		Before behavir	ng aggress	ively
11. Think about a ti	me when you were think	ting about hurting or inj	uring som	eone. What
happened to make	you think this way?			
12. What feelings do	you have when you think	about behaving aggressi	vely?	
Disgust Surprise Anger Trust	Annoyance Fear Calm Shame	Boredom Trust Love Anticipation	Jo Fe Co	onfusion
Amusement Security	Hope	Hate	De	espair
Please circle never, se	ometimes or always for th	e following questions		
11. I like thinking abo Always	out behaving aggressively		Never	Sometimes
12. I wish I didn't thi Always	nk about behaving aggres	sively	Never	Sometimes
13. When I think abov	ut behaving aggressively l	I feel in control	Never	Sometimes
14. I feel sad when th Always	inking about behaving ag	gressively	Never	Sometimes
15. Thinking about be Always	ehaving aggressively clam	ns me down	Never	Sometimes
16. I feel excited whe Always	n thinking about behaving	g aggressively	Never	Sometimes

17. I try to stop thinking of behaving aggressively Always	Never	Sometimes
18. Thinking about behaving aggressively makes me feel better about myself Always	Never	Sometimes
19. Most people have thoughts about behaving aggressively Always	Never	Sometimes
20. I feel bad about having thoughts of behaving aggressively Always	Never	Sometimes
21. The more I think about behaving aggressively the angrier I get Always	Never	Sometimes
22. I think about aggressive ways that I can teach someone a lesson Always	Never	Sometimes
23. I feel anxious when I think about behaving aggressively Always	Never	Sometimes
24. I worry I will hurt people when I think aggressively Always	Never	Sometimes

Appendix N

Questionnaire of Unpleasant Intrusive Thoughts (QUIT; Pascual-Vera et al., 2019)

MENTAL INTRUSIONS WITH UNPLEASANT CONTENT

Indicate the **frequency** and **discomfort** for each of the intrusions on

the list. Use these response scales.

Response scale for FREQUENCY

0	1	2	3	4	5	6
NEVER:	RARELY:	OCCASIONALLY:	SOMETIMES:	OFTEN:	QUITE	ALWAYS:
"I have	"I have had	"I have had this	"I have this	"I have this	OFTEN:	"I have this
had this	mental	a few times a year"	once or twice a	intrusion once	"I have	frequently
mental	intrusion		month"	or	this	throughout the
intrusion"	once or twice in			twice a week"	intrusion	day"
	my life"				every	
					day"	

Response scale for DISCOMFORT

0	1	3	4				
Does not bother	Is somewhat	Disturb	Disturbs	Is extremely			

Nº	CONTENT OF THE SUDDEN, UNINVITED FREQUENCY DISCO							CO	OMF				
1	While using a sharp object, like a knife, scissors, or a tool, I have had mental intrusions about injuring or harming a person close to me with it, including family members or friends.	0	1	2	3	4	5	6	0	1	2	3	4
2	When in a high place, like a cliff, a bridge or a tall building, I have had mental intrusions of iumping off.	0	1	2	3	4	5	6	0	1	2	3	4
3					3	4	5	6	0	1	2	3	4
4	For no special reason, I have had mental intrusions about existential doubts that don't make sense about myself, my feelings, life, the world	0	1	2	3	4	5	6	0	1	2	3	4
5	Even though I know it's unlikely, I have had mental intrusions about having dirtied or contaminated myself through contact, although very slight, with body fluids such as sweat, saliva, urine, excrement or feces.	0	1	2	3	4	5	6	0	1	2	3	4
6	When in a public place, I have had mental intrusions about being contaminated by, or contracting a disease from, touching something that strangers have touched, such as door knobs, the toilet seat, money, or public phones.	0	1	2	3	4	5	6	0	1	2	3	4
7	Even though I know it is probably not true, I have had mental intrusions that I may have left something on at home. For example, having an intrusion about leaving the kitchen stove, gas, heater, lights, iron, hair straightener on, or a cigarette burning	0	1	2	3	4	5	6	0	1	2	3	4

8	Even though I know it is probably not true, I have had mental intrusions that I may have			2	3	4	5	6	0	1	2	3	4
	forgotten somethingimportant.												
9	9 I have had a mental intrusions that certain objects such as furniture or clothes are not correctly organized, or do not follow a certain order. For example, being asymmetrical, not matching in colour, etc		1	2	3	4	5	6	0	1	2	3	4
10	Even though in an office, house or another location that seems organized, I have had mental intrusions that certain things have to be in "their right" place.	0	1	2	3	4	5	6	0	1	2	3	4
11	Even though I know it's unlikely, I have had mental intrusions that certain actions or situations can cause future misfortune or bad luck in general. For example: walking under a ladder, seeing a black cat, someone looking askance at me, etc.	0	1	2	3	4	5	6	0	1	2	3	4
12	Even though I know it's unlikely, I have had mental intrusions that certain thoughts or images can cause future misfortune or bad luck in general. For example: thinking about something bad, remembering a song, counting numbers, etc.	0	1	2	3	4	5	6	0	1	2	3	4

Now please focus on the mental intrusions from the **list above THAT YOU HAVE EXPERIENCED IN THE PAST 3 MONTHS**. Decide which of these intrusions is the **most DISTURBING**, the most UNPLEASANT, or the one that bothers you most when you have it with a FREQUENCY OF AT LEAST 1.

In the case you have not have had any intrusion with a frequency of at least 1 in the past 3 months, please leave this page blank.

The intrusion that is the **most DISTURBING** to me, from those that **I have experienced** in the past three months, is number_from the list above with a frequency of:

1	2	3	4	5	6
RARELY:	OCCASIONALLY:	SOMETIMES:	OFTEN:	QUITE OFTEN:	ALWAYS:
"I have had this mental	"I have had this intrusion a	"I have this intrusion	"I have this intrusion	"I have this	"I have this intrusion
intrusion once or twice in	few times a year"	once or twice a	once or twice a week"	intrusion every day"	frequently throughout
my life"		month"			the day"

In what form or forms do	you usually have it? ((You can mark various responses):
--------------------------	------------------------	-----------------------------------

Thought or doubt □ image□ impulse□ physical sensation□

Please, write the most disturbing intrusion that you chose from the list above by using your own words:

When was the last time you had this intrusion?:

3 months
Past
Past 7
Past 3
Yesterday

ago
month
days
days

Do you remember what you were doing when you had the intrusion, or if something had happened before you have it? If <u>YES</u>, please write it:

The following questions are related to what YOU EXPERIENCE OR DO when you have YOUR MOST DISTURBING INTRUSION. Indicate the degree to which you agree, or

whether each of these statements applies to you. Use this scale to respond.

0	1	2	3	4

						$\overline{}$
1	It scares me; I get nervous.		1	2	3	4
2	2 It makes me sad.		1	2	3	4
3	3 It interrupts what I'm doing; it distracts me.		1	2	3	4
4	4 I find it unacceptable; it goes against what I want, or against my values		1	2	3	4
5	5 I believe that if I have it, it must be important.		1	2	3	4
6	6 I believe it is important to control it and stop thinking about it.		1	2	3	4
7	7 I believe that if I think it, it's more likely to happen, or to be true.		1	2	3	4
8	8 I don't do anything; I just let it be there.		1	2	3	4
9	9 I try not to think about the intrusion; I try to mentally suppress it		1	2	3	4
10	10 1 gov more of my bon for more mig that more more more market my		1	2	3	4
11	11 I do something that I know counters the unpleasantness provoked by the intrusion. For example, organize, check, touch things or touch		1	2	3	4
	myself, wash, clean, pray, count, repeat an action, say a particular word or phrase, repeat a prayer, think opposite thought, etc.					l
12	I look for information about the intrusion, or ask other people about the intrusion to calm myself down or make	0	1	2	3	4
	sure that the intrusion is not important, or that nothing bad is going to happen because I have it.					
	1 / 8 8 8 11					
13	I distract myself by doing something pleasant or relaxing (i.e. turning up the radio, the TV, playing computer,	0	1	2	3	4
	reading, etc.)					
14 I try to avoid anything that will trigger the intrusion		0	1	2	3	4

Today

Appendix O

Anger Rumination Scale (Sukhodolsky, Golub & Cromwell, 2001)

Instructions: Please rate the following statements in terms of how well they correspond to your beliefs about yourself using the following scale:

I = "almost never" to 4 = "almost always"

- 1. I have long living fantasies of revenge after the conflict is over
- 2. when someone make me angry I can't stop thinking about how to get back at this person
- 3. I have thoughts of a violent nature
- 4. I have difficulty forgiving people who have hurt me

Appendix P

Life History of Aggression (LHA; Coccaro, Berman & Kavoussi, 1997)

Please rate the following statements, with regards to the occurrences of these behaviours since you were aged 13 using the following scale:

 $0 = no \ occurrences; \ 1 = one \ event; \ 2 = two \ or \ three \ events; \ 3 = four \ to \ nine \ events; \ 4 = 10$ or more events; $5 = more \ events \ than \ can \ be \ counted$

- 1. Verbal aggression (e.g., shouting or swearing at somebody or making threats to harm another person)
- 2. Indirect Aggression (aggression directed toward inanimate objects/property, e.g., destroying property or punching a hole in the wall)
- 3. Non-specific fighting (physical aggression like hitting, slapping, kicking etc. whether or not the fight was started by you)
- 4. Physical assault against people (physical aggression that you planned or thought about beforehand)
- 5. Temper tantrums

Appendix Q

Measures of Criminal Attitudes and Associations (MCAA; Mills, Kroner & Forth, 2002).

Please rate your agreement or disagreement with the following attitudinal statements using

this scale: 1 = Disagree, 2 = Agree

- 1. Someone who makes you very angry deserves to be hit
- 2. there is nothing wrong with beating up someone who asks for it
- 3. It's understandable to hit someone who insults you
- 4. It's all right to fight someone if they stole from you
- 5. Someone who make you really angry shouldn't complain if they get hit
- 6. People who get beat up usually had it coming
- 7.It is reasonable to fight someone who cheated you
- 8. It is important that you pay attention, please answer 4= agree
- 9. It's not wrong to fight to save face
- 10. Sometimes you have to fight to keep your self-respect
- 11. There is nothing wrong with beating up a child molester
- 12. It's not wrong to hit someone who puts you down
- 13. Child molesters get what they have coming

Appendix R

Ego-Dystonicity Questionnaire - Reduced Version (EDQ-R; Belloch et al., 2012)

Instructions: Focusing on the most upsetting Aggressive Intrusive Thought, please answer the following on a scale of: $I = strongly \ agree \ to \ 7 = strongly \ disagree$

- 1.Thought is immoral
- 2. Do not want it to come true
- 3. Thought conflicts with personality
- 4. How can I have this thought?
- 5. Not the kind of thought I would expect
- 6. Immorality of having this thought
- 7. Do not want it to come true
- 8. Takes me by surprise
- 9. Get it out of my mind and keep it out
- 10. Would never do anything to make it true
- 11. Does not reflect what I want
- 12. Thought is repulsive
- 13. Need to ensure it will not come true
- 14. Do anything to get rid of it
- 15. Against what is right
- 16. More I have it, less I want it to come true
- 17. Need to prove I'm not the person it suggests
- 18. Absurd thought
- 19. Thought doesn't mean anything at all
- 20. I would be a better person if no thoughts
- 21. More I have it, more I worry it will come true

- 22. Thought is distressing
- 23. No good reason to have this thought
- 24. Makes no sense to have this thought
- 25. Thought is irrational
- 26. Surprise about having this thought
- 27. Nothing appealing about it coming true

Appendix S

Thought Control Questionnaire (TCQ; Luciano et al., 2006)

Most people experience unpleasant, and/or unwanted thoughts (in verbal and/or picture form), which can be difficult to control. We are interested in the techniques that you generally use to control such thoughts. Below are a number of things that people do to control these thoughts. There are no right or wrong answers. Do not spend too much time thinking about each one. Please read each statement carefully, and indicate how often you use each technique, on a scale of 1-4 (1 = never; 2 = sometimes; 3 = often; 4 = almost always).

"When I experience an unpleasant-unwanted thought":

- 1. I call to mind positive images instead
- 2. I punish myself for thinking the thought
- 3. I dwell on other worries
- 4. I get angry at myself for having the thought
- 5. I shout at myself for having the thought
- 6. I think pleasant thoughts instead
- 7. I find out how my friends deal with these thoughts
- 8. I worry about more minor things instead
- 9. I do something that I enjoy
- 10. I try to reinterpret the thought
- 11. It is important that you pay attention, please answer 3 = often to this question
- 12. I think about something else
- 13. I think more about the more minor problems I have
- 14. I try a different way of thinking about it
- 15. I ask my friends if they have similar thoughts
- 16. I talk to a friend about the thought

Appendix T

Obsessive Beliefs Questionnaire (OBQ-20; Moulding et al., 2011)

This inventory lists different attitudes or beliefs that people sometimes hold. Read each statement carefully and decide how much you agree or disagree with it. For each statement, choose the number matching the answer that best describes how you think. Because people are different, there are no right or wrong answers. To decide whether a given statement is typical of your way of looking at things, simple keep in mind what you are like most of the time. Use the following scale:

1=Disagree Very Much, 2=Disagree Moderately, 3=Disagree a little, 4=Neither agree nor disagree, 5=Agree a little, 6=Agree moderately, 7= Agree very much

- 1. If I'm not absolutely sure of something, I'm bound to make a mistake
- 2. To be a worthwhile person, I must be perfect at everything I do
- 3. Even if harm is very unlikely, I should try to prevent it at any cost
- 4. For me, having bad urges is as bad as actually carrying them out
- 5. If I don't act when I foresee danger, then I am to blame for consequences
- 6. In all kinds of daily situations, failing to prevent harm is just as bad as deliberately causing it
- 7. For me, not preventing harm is as bad as causing harm
- 8. I should be upset if I make a mistake
- 9. For me, things are not right if they are not perfect
- 10. Having nasty thoughts means I am a terrible person
- 11. If I do not take extra precautions, I am more likely than others to have or cause a serious disaster
- 12. I am more likely than other people to accidentally cause harm to myself or to others
- 13. Having bad thoughts means I am weird or abnormal

- 14. Even when I am careful, I often think bad things will happen
- 15. Having intrusive thoughts means I'm out of control
- 16. Harmful events will happen unless I am very careful
- 17. I must keep working until it's done exactly right
- 18. To me, failing to prevent disaster is as bad as causing it
- 19. Having a bad thought is morally no different than doing a bad deed
- 20. No matter what I do, it won't be good enough

Appendix U

Self-Ambivalence Measure (SAM; Bhar & Kyrios, 2007)

- 1. I feel torn between different parts of my personality
- 2. I tend to move from one extreme to the other in how I think about myself
- 3. I question the extent to which others want to be close to me
- 4. I have mixed feelings about my self-worth
- 5. I feel that I am full of contradictions
- 6. I think about my worth as a person
- 7. I am constantly aware of how others perceive me
- 8. I doubt whether others really like me
- 9. I fear I am capable of doing something terrible
- 10. I constantly worry about whether I will make anything of my life
- 11. I am secure in my sense of self-worth
- 12. I think about how I can improve myself
- 13. I am mindful about how I come across to others
- 14. I am constantly worried about whether I am a good or bad person
- 15. I question whether I am a moral person
- 16. I tend to think of myself in terms of categories such as "good" or "bad"
- 17. I question whether I am morally a good or bad person
- 18. I am constantly concerned about whether I am "decent" human being
- If I inadvertently allow harm to come to others, this proves I am untrustworthy

Appendix V

Fear of Self Questionnaire (FSQ; Aardema et al., 2013)

Instructions: Please rate your agreement or disagreement with the following statements using this scale:

1 = Strongly disagree, 2 = Disagree, 3 = Somewhat disagree, 4 = Somewhat agree, 5 = Agree, 6 = Strongly agree

- 1. I often question my own character
- 2. I can easily imagine myself as the kind of person that should definitely feel guilty
- 3. I often question my own sanity
- 4. I am sometimes afraid to look inside of myself because I am afraid of what I could find
- 5. I worry about being the sort of person who might do very immoral things
- 6. I'm afraid of the kind of person I might become if I'm not very careful
- 7. I often feel that I do not honestly show the negative reality inside myself
- 8. I must be very careful in order to avoid doing something awful

Appendix W

Obsessive Compulsive Inventory – Revised (OCI-R; Foa et al., 2002).

The following statements refer to experiences that many people have in their everyday lives.

Please indicate the number that best describes **HOW MUCH** that experience has

DISTRESSED or **BOTHERED** you during the **PAST MONTH**.

The numbers refer to the following verbal labels:

 $0 = Not \ at \ all; \ 1 = a \ little; \ 2 = moderately; \ 3 = a \ lot; \ 4 = extremely$

- 1. I have saved up so many things
- 2. I check things more often than necessary
- 3. I get upset if objects are not arranged properly
- 4. I feel compelled to count while I am doing things
- 5. I find it difficult to touch an object when I know it has been touched by strangers or certain people
- 6. I find it difficult to control my own thoughts
- 7. I collect things I don't need
- 8. I repeatedly check doors, windows, drawers etc.
- 9. I get upset if others change the way I have arranged things
- 10. I feel I have to repeat certain numbers
- 11. I sometimes have to wash or clean myself simply because I feel contaminated
- 12. I am upset by unpleasant thoughts that come into my mind against my will
- 13. I avoid throwing things away because I am afraid I might need them later
- 14. I repeatedly check gas and water taps and light switches after turning them off
- 15. I need things to be arranged in a particular order
- 16. I feel that there are good and bad numbers

- 17. I wash my hands more often and longer than necessary
- 18. I frequently get nasty thoughts and have difficulty in getting rid of them

Appendix X

Depression Anxiety Stress Scale (Lovibond & Lovibond, 1995)

Instructions: Please read each statement and circle a number 0, 1, 2 or 3 which indicates how much the statement applied to you **over the past week.** There are no right or wrong answers. Do not spend too much time on any statement.

0 = Did not apply to me at all; 1 = Applied to me to some degree, or some of the time; 2 = Applied to me to a considerable degree or a good part of time; 3 = Applied to me very much or most of the time

- 1. I found it hard to wind down
- 2. I was aware of dryness of my mouth
- 3. I couldn't seem to experience any positive feeling at all
- 4. I experienced breathing difficulty (e.g. excessively rapid breathing, breathlessness in the absence of physical exertion)
- 5. I found it difficult to work up the initiative to do things
- 6. tended to over-react to situations
- 7. I experienced trembling (e.g. in the hands)
- 8. I felt that I was using a lot of nervous energy
- 9. I was worried about situations in which I might panic and make a fool of myself
- 10. I felt that I had nothing to look forward to
- 11. I found myself getting agitated
- 12. I found it difficult to relax
- 13. I felt down-hearted and blue
- 14. I was intolerant of anything that kept me from getting on with what I was doing
- 15. I felt I was close to panic

- 16. I was unable to become enthusiastic about anything
- 17. I felt I wasn't worth much as a person
- 18. I felt that I was rather touchy
- 19. I was aware of the action of my heart in the absence of physical exertion (e.g. sense of heart rate increase, heart missing a beat)
- 20. I felt scared without any good reason
- 21. I felt that life was meaningless

Appendix Y

Semi-Structured Interview

Introduction

Thank the participant for agreeing to participate in the interview, and provide them with a reminder that the interview purpose is to better understand their experience of aggressive intrusive thoughts/aggressive scripts. Specifically address that the interview will ask them about how they have experienced these thoughts in the past, and what they felt whilst experiencing them

- 1. Have you experienced thoughts about physically harming or injuring another person?
 - a) Tell me about how this thought arrived in your mind?
 - b) When did you first notice this thought?
 - c) Did this thought appear out of the blue, or have you been thinking about this for several days?
 - d) How does this thought interrupt what you are doing?
 - e) What is it about this thought that make it feel like it intrudes your mind?
- 2. Think about the content of thought what stands out for you?
 - a) Do you experience this thought as unwanted?; what is about this thought that makes it unwanted for you?
 - b) How do you feel when you experience this thought?
 - c) How would you describe your reaction to this thought?
 - d) What is about this thought that is distressing?
- 3. People have different emotional reactions to these types of thoughts; what would you say your emotional reaction is when this thought appears in your mind?

^{*}Provide participant with list of different emotions*

- a) How do you feel when you experience this thought?
- 4. Think about what you do when you experience these types of thoughts:
 - b) Do you find the thought difficult to control?
 - c) What do you find useful in managing these thoughts?
 - d) Do you have any strategies to cope with the thought; if YES, explore further
- 5. Sometimes our thoughts reflect things that we want to happen, or things we don't want to happen; can you tell me about what the thought reflects to you?
 - a) What do you believe this thought means about you?
 - b) What do you think these thoughts mean?

INTRUSIVENESS

- > Tell me more about how this thought arrived in your mind?
- ➤ What is it about this this thought that makes it feel like it intrudes your mind?
- To what extent does this thought intrude your mind?
- ➤ How does this thought interrupt what you are doing?
- ➤ How does this thought disrupt your day to day life?

UNWANTED

- ➤ What is it about this thought that makes it unwanted for you?
- ➤ In what ways is this thought unwanted?

SPONTANEOUS

- ➤ When did you first notice this thought?
- ➤ When does this thought appear in your mind?
- ➤ Did this thought appear out of the blue, or was it pre-conceived?

DISTRESS

- ➤ How do you feel when you experience this thought?
- ➤ How would you describe your reaction to this thought?
- ➤ What is about this thought that is distressing/disturbing?

EMOTIONAL REACTION

- ➤ How do you feel when you experience this thought?
- ➤ What feelings do you have when you have these thoughts about hurting or injuring other people?
- a) Interested
 c) Excited
 e) Strong
 g) Scared
 i) Enthusiastic
 k) Irritable
 m) Ashamed
 o) Nervous

q) Attentive

s) Active

b) Distressed
d) Upset
f) Guilty
h) Hostile
j) Proud
l) Alert
n) Inspired
p) Determined
r) Jittery
t) Afraid

CONTROL AND NEUTRALISING

- > Do you find the thought difficult to control?
- What do you do with the thought when it appears in your mind?
- ➤ What do you find useful in managing this thought?
- > Do you have any strategies to cope with the thought?;if YES further explore.

BELIEFS: EGO-DYSTONICITY; FEARED SELF

- Sometimes our thoughts reflect things that we want to happen, or things we don't want to happen; can you tell me about what the thought reflects to you?
- What do you believe this thought means about you?
- ➤ What do you think these thoughts mean?

Appendix Z

Supplementary Material S1

Anger Rumination Scale: Thoughts of Revenge Subscale (ARS; Sukhodolsky et al., 2001). The present study utilised the Thoughts of Revenge Subscale which assess one's thoughts about anger, and attitudes towards retaliation after provoking situations. Items such as "when someone makes me angry I can't stop thinking about how to get back at this person" are rated on a 4-point Likert scale ranging from 1 (almost never) to 4 (almost always). The ARS has demonstrated good internal consistency across the full scale ($\alpha = 0.93$) and subscales ($\alpha = 0.72 - 0.85$).

Depression Anxiety Stress Scales Short Form (DASS-21; Lovibond & Lovibond, 1995). The DASS assess one's emotional states across three subscales: depression, anxiety, and stress symptoms. It contains 21 items such as "I found it difficult to relax", and participants are asked to rate items on a 4-point Likert scale ranging from 0 (did not apply to me at all) to 3 (applied to me very much or most of the time), with reference to the past week. The DASS-21 has demonstrated good internal consistency (α s = 0.91; 0.80; 0.84, for Depression, Anxiety, and Stress respectively), and good convergent validity with the Mental Component Summary score (r = -0.58 to -0.69; Sinclair et al., 2012).

Ego-Dystonicity Questionnaire- Reduced Version (EDQ-R; Belloch et al., 2012). The EDQ is a self-report containing 27 items that assess beliefs concerning the content of one's thoughts and how they may be experienced as inconsistent with ones self-beliefs, values, and moral attitude. The present study modified the EDQ to ask participants to focus on their most upsetting 'aggressive thought' whilst providing their ratings. Participants rate items such as that the "Thought conflicts with personality" on a 7-point Likert scale ranging from 1 (strongly agree) to 7 (strongly disagree). The EDQ-R has demonstrated good internal

consistency (α s = 0.86) and has been found to measure three separate factors (undesirability: α s = 0.92; irrationality: α s = 0.80; immorality: α s = 0.70; Belloch et al., 2012).

Fear of Self Questionnaire (FSQ; Aardema et al., 2013). The FSQ is an eight item self-report measure which assess beliefs concerning hidden or covert aspects of one's personality. Participants rate items such as "I often question my own sanity" on a 6-point Likert scale ranging from 1 (strongly disagree) to 6 (strongly disagree). The FSQ has demonstrated excellent internal consistency ($\alpha = 0.93$), good convergent validity with other measure of self-related beliefs and constructs (i.e., Self-Ambivalence Measure r = 0.68; Inferential confusion Questionnaire r = 0.72; Aardema et al., 2013).

Life History of Aggression (LHA; Coccaro et al., 1997). The LHA, as revised by Coccaro et al. (1997), is a self-report measure that assesses the number of occurrences of aggressive behaviours since the age of 13. The Aggression subscale, which assesses the experience of overt aggressive behaviour, was only utilised in the current study. Participants rate items such as "Physical assault against people" on a 5-point Likert scale ranging from 0 (no occurrences) to 5 (more events than can be counted). In the current study, the Aggression subscale demonstrated good internal consistency ($\alpha = 0.79$). The LHA full scale and the Aggression subscale have demonstrated good internal consistency ($\alpha = 0.88$; $\alpha = 0.87$, respectively; Coccaro et al., 1997).

Measures of Criminal Attitudes and Associations (MCAA; Mills et al., 2002). The complete MCAA is a two-part self-report questionnaire, comprising Violence, Entitlement, Antisocial Intent, and Associates subscale. The present study only utilised 13 items from the Violence subscale which included items such as "Someone who makes you very angry deserves to be hit" and are rated on a dichotomous scale of agree/disagree. The MCAA full scale ($\alpha = 0.90$) and the subscale of Violence ($\alpha = 0.80$) have demonstrated excellent internal consistency (Mills et al., 2002).

Obsessive Beliefs Questionnaire (OBQ-20; Moulding et al., 2011). The OBQ-20 is a short form of the Obsessive Beliefs Questionnaire (OCCWG, 2005). The OBQ contains 20-items and assesses four obsessive belief domains which have been identified through factor analyses: (1) Threat, (2) Responsibility, (3) Importance of Thoughts, and (4) Perfectionism. Participants rate items such as "For me, not preventing harm is as bad as causing harm" on a 7-point Likert scale ranging from 1 (Disagree very much) to 7 (Agree very much). The OBQ-20 has demonstrated good internal consistency (α s = 0.77 – 0.83; Moulding et al., 2011).

Obsessive Compulsive Inventory – Revised (OCI-R; Foa et al., 2002). The OCI-R is an 18-item self-report measure that assesses obsessive compulsive symptoms and associated distress associated. Participants rate items such as "I check things more often than necessary' on a 4-point Likert scale ranging from 0 (not at all) to 4 (extremely). The OCI-R has demonstrated good psychometric properties, and excellent test-retest reliability for a 2 week period (r = 0.74 to 0.91).

Schedule of Imagined Violence (SIV; Grisso et al., 2000). The original SIV contains eight items which assess participants' experience of a violent thought, as well as subsequent aggressive actions. Only the frequency item of the SIV ("How often do you have thoughts about hurting or injuring other people?") was used in the current study. Participants rated their responses to this item on a 7-point Likert scale ranging from 0 (never) to 7 (several times a day). This item has been used in prior research to measure the frequency of one's aggressive script rehearsal (Daff et al., 2015; Hosie, Simpson, et al., 2022; Podubinski et al., 2017).

Social Desirability Scale (SDS; Stöber, 2001). The SDS is a 17 item self-report measure, devised from the Marlowe-Crowne Scale (Lück & Timaeus, 1969), that assess the extent to which participants engage in social desirable responding. Participants respond to

items such as "I sometimes litter" on a dichotomous scale of true or false. The SDS has demonstrated adequate internal consistency ($\alpha = 0.72$), and good test-retest reliability over a 4-week period (r = 0.82; Stöber, 2001).

Self-Ambivalence Measure (SAM; Bhar & Kyrios, 2007). The SAM is a 19-item measure of self-ambivalence—which encompasses beliefs regarding uncertainty towards the self, and dichotomous perceptions about one's self-concept. Participants respond to items such as "I feel that I am full of contradictions" on a 5-point Likert scale ranging from 0 (not at all) to 4 (agree totally). The SAM has demonstrated acceptable reliability and good internal consistency (α s = 0.88 – 0.85; Bhar & Kyrios, 2007).

Thought Control Questionnaire (TCQ; Luciano et al., 2006). The TCQ contains 16 items assessing the use of different thought control strategies. Participants rate items such as "I get angry at myself for having the thought" on a 4-point Likert scale ranging from 1 (never) to 4 (almost always). The 16 item version of the TCQ has demonstrated good internal consistency ($\alpha = 0.75$), and acceptable discriminant validity amongst the five factors (Luciano et al., 2006).

Questionnaire of Unpleasant Intrusive Thoughts (QUIT; Pascual-Vera et al., 2019). The QUIT assesses the experience of a range of intrusive thought content themes. Only the unpleasant content domain was used in the current study, and analyses involving the QUIT only used the frequency item of unwanted aggressive intrusive thoughts. Participants rated their responses on a 7-point Likert scale ranging from 0 (*never*) to 6 (*always*). The psychometric properties of the full QUIT have been assessed cross-culturally (Pascual-Vera et al., 2019), and has demonstrated good to excellent internal consistency across different countries.